CANADA

SUPERIOR COURT Commercial Division

PROVINCE OF QUÉBEC DISTRICT OF MONTRÉAL (Sitting as a court designated pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. c. C-36)

N°: 500-11-049256-155

IN THE MATTER OF THE PLAN OF COMPROMISE AND ARRANGEMENT OF:

MAGASIN LAURA (P.V.) INC. / LAURA'S SHOPPE (P.V.) INC.

Debtor / Petitioner

-and-

BOUTIQUE LAURA CANADA LTÉE / LAURA'S SHOPPE CANADA LTD.

-and-

3482731 CANADA INC.

-and-

9318-5494 QUEBEC INC.

-and-

KALMAN FISHER

Stayed Parties

-and-

KPMG INC.

Monitor

MOTION FOR SANCTION OF THE PETITIONER'S RE-AMENDED PLAN OF COMPROMISE AND ARRANGEMENT AND FOR OTHER RELIEF

(Sections 6, 9, 11 and 11.02 of the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36)

TO THE HONOURABLE JUSTICE MARIE-ANNE PAQUETTE OR TO ONE OF THE OTHER HONOURABLE JUDGES OF THE SUPERIOR COURT, SITTING IN THE COMMERCIAL DIVISION, IN AND FOR THE DISTRICT OF MONTREAL, THE PETITIONER RESPECTFULLY SUBMITS AS FOLLOWS:

Unless otherwise indicated, capitalized terms found herein shall have the same meaning ascribed thereto in the Petitioner's Re-Amended Plan of Compromise and Arrangement dated November 18, 2015.

I. INTRODUCTION

- 1. On August 12, 2015, this Honourable Court (the "Court") issued an order extending the protection of the Companies' Creditors Arrangement Act (the "CCAA") to Debtor/Petitioner Magasin Laura (P.V.) Inc./Laura's Shoppe (P.V.) Inc. (the "Petitioner") and, to a lesser extent, to each of Boutique Laura Canada Ltée/Laura's Shoppe Canada Ltd., 3482731 Canada Inc., 9318-5494 Quebec Inc. and Kalman Fisher (the "Stayed Parties") pursuant to the provisions of the CCAA, in general, and Section 11.02 thereof, in particular (such order, together with all subsequent amendments thereto, the "Initial Order").
- 2. Pursuant to the Initial Order, KPMG Inc. (the "Monitor") was appointed monitor of Petitioner and a stay of proceedings was ordered in respect of Petitioner and the Stayed Parties up to and including September 11, 2015.
- **3.** On September 11, 2015, this Court issued an order extending the foregoing stay of proceedings until November 30, 2015 (the "**Stay Period**").

II. INTERIM FINANCING AND REFINANCING

- **4.** The Initial Order approved interim financing to Petitioner by The Cadillac Fairview Corporation Limited ("**CF**") and created first ranking charges in CF's favour over Petitioner's property ranking ahead of charges in favour of, *inter alia*, Salus Capital Partners, LLC ("**Salus**").
- 5. On October 30, 2015, this Court issued an Order authorizing the pay-out and replacement of the credit facilities provided to Petitioner by each of CF and Salus by Third Eye Capital Corporation, acting as agent and hypothecary representative for and on behalf of certain affiliates and funds it manages ("Third Eye"), approving such refinancing of Petitioner by Third Eye and creating a new first ranking charge in Third Eye's favour over all of Petitioner's property.

III. INITIAL PLAN AND RE-AMENDED PLAN

6. On October 23, 2015, Petitioner filed with this Court Petitioner's "Plan of Compromise and Arrangement" (the "Initial Plan") and this Court issued an "Order for the Filing of the Plan of Compromise and Arrangement and the Calling of a Creditors' Meeting" (the "Creditors' Meeting Order") authorizing the calling of a meeting of Affected Creditors to consider and vote on the Initial Plan.

- 7. On November 17, 2015, Petitioner filed with this Court Petitioner's "Amended Plan of Compromise and Arrangement" dated November 13, 2015 (the "Amended Plan") and distributed such Amended Plan to the Service List herein.
- 8. On November 18, 2015, as a result of discussions with certain of its major landlords, Petitioner amended the Amended Plan in order to be more consistent with Section 32(7) CCAA. Accordingly, Petitioner distributed its "Re-Amended Plan of Compromise and Arrangement" dated November 18, 2015 (the "Re-Amended Plan") to the Service List herein and subsequently filed the Re-Amended Plan with this Court on November 19, 2015.
- 9. As appears from a copy of the email of November 18, 2015 from the Monitor's legal counsel communicated herewith as Exhibit P-1 (the "Email"), the Monitor is of the view that such changes made to the Amended Plan are reasonable under the circumstances.
- 10. A copy of the Initial Plan is communicated herewith as Exhibit P-2, a copy of the Re-Amended Plan is communicated herewith as Exhibit P-3, a compared version between the Initial Plan and the Re-Amended Plan is communicated herewith as Exhibit P-4 and a compared version between the Amended Plan and the Re-Amended Plan is communicated herewith as Exhibit P-5.

IV. CLAIMS PROCEDURE

- 11. On September 11, 2015, this Court issued a "Claims Procedure Order" (the "Claims Procedure Order") establishing a procedure for the filing with the Monitor of Proofs of Claim and establishing a bar date for the filing of such Proofs of Claim (the "Claims Bar Date").
- 12. In accordance with the Claims Procedure Order, the Monitor forwarded the requisite notices, materials and Proof of Claim forms to the Affected Creditors and the Monitor, prior to the Claims Bar Date, received Proofs of Claim from such Affected Creditors.
- 13. Thereafter, the Monitor, together with the Company, examined the filed Proofs of Claim and determined that certain Proofs of Claim filed by landlords of the Petitioner should be either partially or wholly disallowed in accordance with the Amended Plan. In the case of the Petitioner's suppliers, the Monitor has not yet been able to complete its review of the Proofs of Claim filed. Accordingly, following the procedure set forth in the Claims Procedure Order, certain Proofs of Claim were partially or wholly disputed by the Monitor (the "Disputed Claims").
- 14. As a result of the Re-Amended Plan, on November 18, 2015, the Monitor withdrew the Notices of Disallowance or Revision which had been sent to certain landlords in respect of the Disputed Claims and the Monitor intends to reassess such Claims in conformity with the Re-Amended Plan, the whole subject to the supervision of the Court.

V. CREDITORS' MEETING

- 15. In accordance with the Creditors' Meeting Order, the Monitor published and gave the requisite notices in order to properly convene and hold a meeting of the Affected Creditors on November 19, 2015.
- **16.** In accordance with the foregoing notices, a meeting of the Affected Creditors was convened and held on November 19, 2015 (the "Creditors' Meeting").

17. At the Creditors' Meeting:

- (a) the Re-Amended Plan was tabled and distributed to all Affected Creditors present at such Creditors' Meeting;
- (b) the provisions of the Re-Amended Plan were summarized and explained by Petitioner's representatives and legal counsel, the Monitor and the Monitor's legal counsel to those Affected Creditors present at such Creditors' Meeting; and
- (c) the Monitor conducted a vote on the Re-Amended Plan by those Affected Creditors present (in person or by proxy) at such Creditors' Meeting.
- **18.** As will be more fully detailed by the Monitor in its report to be filed with the Court, the results of such vote were as follows:
 - (a) 96% of the Affected Creditors present at such Creditors' Meeting (in person or by proxy) voted to approve the Re-Amended Plan; and
 - (b) 99% in value of the Proofs of Claim of all Affected Creditors present at such Creditors' Meeting (in person or by proxy) voted to approve the Re-Amended Plan.
- **19.** Accordingly, the Re-Amended Plan has been overwhelmingly approved by far more than the Required Majorities at the Creditors' Meeting.

VI. SALIENT POINTS OF THE RE-AMENDED PLAN

- **20.** The salient points of the Re-Amended Plan are as follows:
 - (a) all Claims and all Restructuring Claims, to the extent that they are not specifically enumerated in the Re-Amended Plan as Unaffected Claims, are Affected Claims entitled to distributions of the Aggregate Distribution under the Re-Amended Plan;
 - (b) the Affected Claims include Disclaimed Landlord Claims, being determined in accordance with Section 32(7) CCAA;

- the Unaffected Claims (i.e. Claims not constituting Affected Claims) include any Excluded Claims, the Administration Claim, the DIP Claim, the KERP Claims, any Gift Card Claims, any Crown Priority Claims, any Section 6(5) Claims, any Section 19(2) Claims, the Secured Claims, the Replacement Financing Claim and the Fisher Group Claims;
- (d) the Affected Creditors having filed Proofs of Claim prior to the Claims Bar Date which are not ultimately Disallowed Claims, are entitled to participate in an Aggregate Distribution in the amount of \$4 Million, of which, \$1 Million will be paid by Petitioner by April 30, 2016, a further \$1 Million will be paid by Petitioner by July 31, 2016, a further \$1.8 Million will be paid by Petitioner by December 15, 2016 and a final \$200,000 will be paid by Petitioner by February 15, 2017, in full and final payment, settlement and release of all Affected Claims;
- (e) in accordance with the requirements of the CCAA, all Crown Priority Claims described in Section 6(3) CCAA, if any, will be paid in full by Petitioner within 6 months immediately following the Sanction Date;
- (f) in accordance with the requirements of the CCAA, all Section 6(5) Claims, if any, will be paid in such amounts as required under the CCAA immediately after the Sanction Date;
- (g) in accordance with the requirements of the CCAA, all Section 19(2) Claims, if any, will be paid by Petitioner as and when they become due;
- (h) conditional upon approval of the Re-Amended Plan by the Affected Creditors, this Court's issuance of the Sanction Order and implementation of the Re-Amended Plan, each member of the Fisher Group:
 - (i) has subordinated and postponed all of the Fisher Group Claims (i.e. outstanding loans owed by Petitioner to members of the Fisher Group) to and in favour of full payment of the Aggregate Distribution to the Affected Creditors; and
 - (ii) has waived and renounced to any right to be an Affected Creditor under the Re-Amended Plan;
- (i) any Affected Creditor who filed its Proof of Claim prior to the Claims Bar Date (or who has been permitted to file a late claim pursuant to any other order issued by this Court) shall be entitled to amend its Proof Claim in order for its Proof of Claim to conform to the provisions of the Re-Amended Plan;
- (j) implementation of the Re-Amended Plan is conditional upon:

- the Affected Creditors' approval of the Re-Amended Plan by the Required Majorities, which occurred at the Creditors' Meeting;
- (ii) this Court's sanction of the Re-Amended Plan as requested by the present Motion; and
- (iii) the occurrence of the Replacement Financing (i.e. the repayment and/or replacement of Third Eye's above-described financing by new credit facilities to be provided to Petitioner by Third Eye) or waiver of such condition by Petitioner and Third Eye,

all by no later than December 18, 2015 or such later date as may be ordered by this Court.

- 21. Petitioner and Third Eye are diligently working towards such Replacement Financing, which is expected to be completed shortly. Petitioner and Third Eye contemplate seeking Court approval for the Replacement Financing in early December.
- **22.** Petitioner, through the *CCAA* process, has significantly restructured its business by, *inter alia*:
 - (a) disclaiming leases for unprofitable retail stores and either ultimately disclaiming and closing such retail stores or entering into new lease arrangements to render such retail stores viable going forward;
 - (b) disclaiming leases and significantly reducing Petitioner's head office premises; and
 - (c) downsizing its personnel both at the head office and retail store level.
- 23. Given the challenges of the Canadian economy, in general, and the Canadian retail industry, in particular, Petitioner's restructuring has been successful and Petitioner's Re-Amended Plan is fair and reasonable in that:
 - (a) the Re-Amended Plan allows Petitioner to remain in business, operating a Canadian owned major women's apparel store chain with stores in almost every Canadian province;
 - (b) Petitioner will remain a vibrant rent-paying tenant in approximately 140 retail stores across Canada;
 - (c) Petitioner will continue to purchase almost \$100 million of merchandise annually, mostly from Canadian (and some US) wholesalers; and
 - (d) Petitioner will continue to employ almost 2000 employees throughout Canada, including some 240 employees at its head office in Laval, Quebec.

VII. AMENDMENT OF CLAIMS PROCEDURE ORDER

- 24. The Claims Procedure Order is currently silent on who should litigate any Disputed Claim which may not have been settled between the relevant Affected Creditor, the Monitor and the Petitioner.
- 25. Both Petitioner and the Monitor agree that it may be more practical and economical for such Disputed Claims to be litigated by either Petitioner or the Monitor.
- 26. For such reason, Petitioner, with the Monitor's concurrence and by this present Motion, seeks an amendment to the Claims Procedure Order so as to allow a Disputed Claim to be litigated either by the Monitor or Petitioner, the whole without prejudice to the right of the Monitor to bring a motion before this Honourable Court seeking directions as to the manner in which to quantify the Disclaimed Lease Claims and the Restructuring Claims filed by landlords.

VIII. EXTENSION OF THE STAY PERIOD

- 27. As set forth above, implementation of the Re-Amended Plan is to occur upon occurrence (or waiver, as the case may be) of all the Plan Implementation Conditions, but no later than December 18, 2015, or such other date as may be ordered by this Court (being the Plan Implementation Date under the Re-Amended Plan).
- 28. Petitioner accordingly seeks that the current November 30, 2015 Stay Period be extended until December 18, 2015, or such later date as may be ordered by this Court. In approving the Re-Amended Plan, the Affected Creditors agreed to such extension (see Section 6.8 of the Re-Amended Plan).
- 29. The above and other matters are detailed in the Fourth Report of the Monitor dated November 17, 2015 and supplemented by the Email. A further report from the Monitor confirming the above details of the Creditors' Meeting will be filed with the Court at or prior to hearing of the present Motion.
- **30.** The present Motion is well founded in fact and in law.

WHEREFORE Petitioner prays for Judgment of this Court:

- (A) GRANTING the present Motion;
- (B) ORDERING that the Claims Procedure Order be amended by adding a new paragraph 7.2.1 thereto reading as follows:
 - "7.2.1 any dispute ensuing from an appeal of a Notice of Revision or Disallowance may be litigated by either the Monitor or the Petitioner"

- (C) ORDERING that the Stay Period, as defined in the Initial Order, shall be and remain in full force and effect up until December 18, 2015, or such later date as ordered by this Court, the whole subject to all other terms of the Initial Order and any subsequent Orders or judgments of this Court;
- (D) ISSUING the Sanction Order in the form of the Draft Order produced herewith as Exhibit P-6;

THE WHOLE without costs, save and except in the event of contestation.

MONTREAL, November 19, 2015

(SGD.) Kugler Kandestin LLP

KUGLER KANDESTIN LLP Attorneys for Debtor/Petitioner

TRUE COPY

KUGLER KÄNDESTIN LLP

<u>AFFIDAVIT</u>

- I, the undersigned, **KALMAN FISHER**, businessman, doing business in the City of Montreal, Province of Quebec and residing at 4 Granville, in the City of Hampstead, Province of Quebec, solemnly affirm that:
 - 1. I am the President of the Petitioner; and
 - 2. All of the facts alleged by the Applicant in the present *Motion for Sanction of the Petitioner's Re-Amended Plan of Compromise and Arrangement and for Other Relief* which do not appear of record in this Court file are true and correct.

AND I HAVE SIGNED:

(SGD.) Kalman Fisher

KALMAN FISHER

SOLEMNLY AFFIRMED before me at the City of Montreal, Province of Quebec, this 19th day of **November**, 2015.

(SGD.) Darlene W. Pitt #198 577

Commissioner of Oaths for the Province of Quebec

TRUE COPY

KUGLER KANDESTIN LLP

NOTICE OF PRESENTATION

TO: SERVICE LIST and other parties.

TAKE NOTICE that the present *Motion for Sanction of the Petitioner's Re-Amended Plan of Compromise and Arrangement and for Other Relief* will be presented for adjudication before the Commercial Division of the Superior Court of Quebec, in and for the district of Montreal, on **November 27, 2015** at 9:30 a.m. in a room to be announced.

DO GOVERN YOURSELVES ACCORDINGLY

MONTREAL, November 19, 2015

(SGD.) Kugler Kandestin LLP

KUGLER KANDESTIN LLP

Attorneys for the Debtor/Petitioner

TRUE COPY

KUGTER KANDESTIN LI P

LIST OF EXHIBITS

EXHIBIT P-1: Email of November 18, 2015 from Monitor's counsel.

EXHIBIT P-2: Initial Plan dated October 16, 2015.

EXHIBIT P-3: Re-Amended Plan dated November 18, 2015.

EXHIBIT P-4: Compared version between the Initial Plan and the Re-Amended

Plan.

EXHIBIT P-5: Compared version between the Amended Plan and the Re-

Amended Plan.

EXHIBIT P-6: Draft Sanction Order.

MONTREAL, November 19, 2015

(SGD.) Kugler Kandestin LLP

KUGLER KANDESTIN LLP

Attorneys for the Debtor/Petitioner

TRUE COPY

VUCLED KANDESTIN LLD

EXHIBIT P-1

(E-Mail of November 18, 2015 from Monitor's counsel)

Jeremy Cuttler

Subject:

From: Mark Meland <mmeland@ffmp.ca>

Sent: November-18-15 6:54 PM

To: Marie-Anne Paquette (bureau-CS) (marie-anne.paquette@judex.qc.ca); Darlène Jérome

(darlene.jerome@judex.qc.ca)

Cc: Gerald F. Kandestin; David Stolow; Jean-francois Carpentier; Jeremy Cuttler;

rmassi@richter.ca; ebarbieri@richter.ca; obenchaya@richter.ca; francois.viau@gowlings.com; patrice.benoit@gowlings.com; pierre-

luc.beauchesne@gowlings.com; paule.tardif@gowlings.com; St-Onge, Denis;

ingrid.anton@gowlings.com; claudio.filippone@ca.pwc.com; david.malin@ca.pwc.com; valerie.berger@ca.pwc.com; dcoossa@kpmq.ca; mcodere@kpmq.ca; cadjami@kpmq.ca;

jsidawy@kpmg.ca; Avram Fishman; Jason Dolman; sabitan@osler.com; mdesrosiers@osler.com; zshi@osler.com; krosenstein@airdberlis.com; sgraff@airdberlis.com; sbabe@airdberlis.com; gerry.apostolatos@lkd.ca; pascal.archambault@lkd.ca; william.hart@lkd.ca; wstevenson@mcleankerr.com; lgalessiere@mcleankerr.com; nstein@steinandstein.com; jjbb@aikins.com;

dsaponara@kutmanlaw.com; reisa@bellnet.ca; Imorin@fasken.com; c.labelle@lplv.com;

kmailloux@dwpv.com; reneg@lazchar.com; gerry.apostolatos@lkd.ca;

pascal.archambault@lkd.ca; mekalash@phkavocats.com; randysgarcha@gmail.com;

mcitak@mcleankerr.com; n.guizani@lplv.com; sraicek@dgclex.com;

grosen@dgclex.com; louis.sebastien@justice.gc.ca; Heather.Wellman@gov.bc.ca;

Lauren.Longbottom@gov.bc.ca; pmcgovern@businesslawboutique.com;

kpimentel@dv-law.com; ggalati@dv-law.com; Sylvain.rigaud@nortonrosefulbright.com;

Serge.levy@nortonrosefulbright.com; Chrystal.ashby@nortonrosefulbright.com LAURA'S SHOPPE (P.V.) INC. (In the Matter of the Plan of Compromise and

Arrangement) S.C.M.: 500-11-049256-155

To the Honourable Madame Justice Paquette and colleagues:

Subsequent to the filing by Laura's Shoppe (P.V.) Inc. ("Laura") of an amended plan of compromise and arrangement on November 17, 2015 (the "November 17th Plan"), KPMG Inc., in its capacity as Monitor to Laura, issued Notices of Revision or Disallowance to certain of Laura's landlords.

On November 18, 2015, Laura served a re-amended plan of compromise and arrangement (the "Re-Amended Plan") that contains changes to the definitions of "Disclaimed Lease" and "Restructuring Claim". More specifically, paragraph 2.3 thereof now provides that Disclaimed Landlord Claims shall be determined in accordance with s. 32(7) of the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36 (the "CCAA"), which provision stipulates that, if an agreement is disclaimed or resiliated, a party to the agreement who suffers a loss in relation to the disclaimer or resiliation is considered to have a provable claim. Furthermore, the definition of Restructuring Claim now includes reference to claims arising out of Laura's renegotiation of leases after the Determination Date.

The Monitor has withdrawn the Notices of Disallowance that had been issued to certain landlords in respect of the November 17th Plan and shall re-evaluate the Claims referred to therein in accordance with the terms of the Re-Amended Plan. If deemed appropriate, the Monitor shall seek directions from the Court as to the quantification of Claims asserted by landlords.

It is the Monitor's view that because the amendments contained in the Re-Amended Plan are consistent with the CCAA, they are reasonable under the circumstances and do not prevent any Creditors from asserting their positions in respect

of the manner in which the Disclaimed Landlord Claims and the Restructuring Claims will be quantified, the whole subject to the supervision of the Court.

Yours very truly,

Mark E. Meland

FISHMAN FLANZ MELAND PAQUIN LLP 1250 René-Lévesque Blvd. West Suite 4100 Montreal QC H3B 4W8

Tel: 514-932-4100 Fax: 514-932-4170

email: mmeland@ffmp.ca



Ce courriel est confidentiel et peut être protégé par le secret professionnel. Toute utilisation, reproduction, diffusion, publication, modification ou retransmission, sous quelque forme, même partielle, de ce courriel et de son contenu, est strictement interdite. Si vous recevez ce courriel par erreur, veuillez s'il vous plaît nous en aviser et le détruire immédiatement. This email is confidential and may be privileged. It is strictly forbidden to utilize, reproduce, diffuse, publish, modify or retransmit, in any way, even partially, this email and its contents. If you receive this email by mistake, please advise us and destroy it immediately.

EXHIBIT P-2

(Initial Plan dated October 16, 2015)

CANADA

PROVINCE OF QUEBEC DISTRICT OF MONTREAL

NO.: 500-11-049256-155

SUPERIOR COURT (Commercial Division)

(Sitting as a court designated pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. c. C-36)

IN THE MATTER OF THE PLAN OF COMPROMISE AND ARRANGEMENT OF:

MAGASIN LAURA (P.V.) INC. / LAURA'S SHOPPE (P.V.) INC.

Debtor/Petitioner

-and-

BOUTIQUE LAURA CANADA LTÉE / LAURA'S SHOPPE CANADA LTD.

-and-

3482731 CANADA INC.

-and-

9318-5494 QUEBEC INC.

-and-

KALMAN FISHER

Stayed Parties

-and-

KPMG INC.

Monitor

PLAN OF COMPROMISE AND ARRANGEMENT
Under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36

SECTION 1 INTERPRETATION

1.1 <u>Definitions</u>

In this Plan, unless otherwise stated or the context otherwise requires:

- "Administration Charge" has the meaning ascribed to such term in the Initial Order,
- "Administration Claim" means a claim or any other indebtedness or obligation secured by the Administration Charge;
- "Affected Claim" means any Claim other than an Unaffected Claim. For greater certainty, Affected Claims include Disclaimed Landlord Claims, any Claim of Her Majesty the Queen in right of Canada or any province (other than Crown Priority Claims), Directors Claims and Restructuring Claims;
- "Affected Creditor" means a Creditor holding an Affected Claim, but only to the extent of its Affected Claim;
- "Affected Creditors Class" has the meaning set forth in Section 2.2 hereof;
- "Aggregate Distribution" has the meaning set forth in Section 2.4 hereof;
- "BIA" means the Bankruptcy and Insolvency Act, R.S.C., 1985, c. B-3, as amended;
- "Business Day" means a day, other than a Saturday, a Sunday, or a non-juridical day (as defined in article 6 of the Quebec Code of Civil Procedure, R.S.Q., c. C-25, as amended);
- "CCAA" means the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended;
- "CCAA Charge(s)" means the Administration Charge, the DIP Lender Charge and the KERP Charge;
- "CCAA Proceedings" means proceedings in respect of the Company before the Court commenced, taken up and continued under the CCAA;
- "Certificate of Implementation" has the meaning set forth in Section 8.3 hereof;
- "Certificate of Performance" has the meaning set forth in Section 6.1 hereof;
- "Certificate of Non-Implementation" has the meaning set forth in Section 8.5 hereof;

"Chair" means a representative of the Monitor, or any other individual designated by the Monitor to preside as chairperson at the Creditors' Meeting;

"Claim" means any right of any Person against the Company in connection with any indebtedness, liability or obligation of any kind of the Company owed to such Person and any interest, or penalties accrued thereon or costs payable in respect thereof, whether liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, unsecured, present, future, known or unknown, by guarantee, surely or otherwise, and whether or not such right is executory or anticipatory in nature, including the right or ability of any Person to advance a claim for contribution or indemnity or otherwise with respect to any matter, action, cause or chose in action, whether existing at present or commenced in the future, which indebtedness, liability or obligation is based in whole or in part on facts existing, or transactions which occurred, prior to the Determination Date, or which would have been claims provable in bankruptcy had the Company become bankrupt on the Determination Date, and, without limitation, including any Restructuring Claim, but excluding any Unaffected Claim and "Claims" means all of them;

"Claims Bar Date" has the meaning as set forth in the Claims Procedure Order;

"Claims Procedure Order" means the "Claims Procedure Order" issued by the Court on September 11, 2015, establishing, among other things, procedures for proving Claims, as amended or supplemented from time to time by further Order(s) of the Court;

"Company", "Debtor" or "Petitioner" means Magasin Laura (P.V.) Inc./Laura's Shoppe (P.V.) Inc.;

"Court" means the Commercial Division of the Quebec Superior Court for the District of Montreal;

"Creditor(s)" means any Person(s) having a Claim and may, where the context requires, include the assignee of a Claim or a trustee, interim receiver, receiver, receiver and manager, or other Person acting on behalf of such Person(s);

"Creditors' Meeting" means the meeting of Affected Creditors to be convened for the purposes of voting on the Plan, or any adjournment of such meeting;

"Crown Priority Claims" means any Claims of Her Majesty the Queen in right of Canada or in right of any province as described in Section 6(3) or Section 38(2) of the CCAA. For greater certainty, any Claim of Her Majesty the Queen in right of Canada or in right of any Province other than Crown Priority Claims shall be an Affected Claim hereunder;

"Determination Date" means July 31, 2015, being the date on which the Company commenced proceedings under the *BIA* by filing a notice of intention to make a proposal thereunder, which proceedings were subsequently taken up and continued by the *CCAA* Proceedings under the *CCAA*;

"DIP Claim" means any and all obligations of the Company to the DIP Lender pursuant to the DIP Facility Documents or the Initial Order as of the Plan Implementation Date;

"DIP Facility Documents" means the "Interim Financing Term Sheet" dated August 10, 2015 by and among the Company and the DIP Lender (as amended, modified, restated and/or supplemented from time to time) together with any related collateral, loan or security documents executed in connection therewith or which relate thereto;

"DIP Lender" means The Cadillac Fairview Corporation Limited;

"DIP Lender Charge" means the "Interim Lender Charge" as set forth in the Initial Order;

"Directors" means all of the Company's past and present directors as well as any Persons who were or are deemed to be directors of the Company pursuant to any applicable Laws;

"Directors' Claims" means all Claims, of any nature or source whatsoever, of any of the Creditors against any or all of the Directors which arose before the Determination Date and which relate to obligations of the Company where such Directors are by law liable in their capacity as directors for payment of such obligations. Director's Liability shall, however, exclude all Claims which (i) relate to contractual rights of one or more Creditors arising from contracts with one or more Directors, including any guarantees given by any Directors relating to the Company's obligations, (ii) seek injunctive relieve against any of the Directors in relation to the Company, or (iii) are based on allegations of misrepresentation made by any Directors to Creditors or of wrongful or oppressive conduct by Directors;

"Disallowed Claim" means any Claim, or that portion thereof which has been revised or disallowed by the Monitor pursuant to the Claims Procedure Order or any other Order in respect of which an appeal by the Creditor has been dismissed or all appeal periods, if any, have expired;

"Disclaimed Leases" means each and every one of the leases and/or other agreements for the occupancy of real or immovable property (as well as all amendments thereto and renewals thereof) which (i) were disclaimed or resiliated by the Company pursuant to Section 60 (1.1) of the BIA (as recognized by the Initial Order) or Section 32 of the CCAA, as exhaustively enumerated in the annex forming part of this Plan, and (ii) may hereafter be disclaimed or resiliated by the Company pursuant to Section 32 of the CCAA, to the extent (in either case) that any such disclaimers or resiliations have not

been or shall not hereafter be rescinded by the Company, and "Disclaimed Lease" means each of them;

"Disclaimed Landlord" means the landlord under any Disclaimed Lease and "Disclaimed Landlords" means all of them;

"Disclaimed Landlord Claims" means all Proven Claims of the Disclaimed Landlords for losses incurred by Disclaimed Landlords in relation to Disclaimed Leases, as provided for in Section 32(7) of the CCAA and "Disclaimed Landlord Claim" means any of them;

"Disputed Claim" means a Claim or that portion thereof that is the object of a Notice of Revision or Disallowance and, in either case, has not become a Proven Claim or a Disallowed Claim;

"Excluded Claim" means any right of any person against the Company in connection with any indebtedness, liability or obligation of any kind which arose with respect to transactions which occurred after the Determination Date and any interest thereon, including the Administration Claim, the DIP Claim, a KERP Claim and any obligation of the Company towards creditors who have supplied or shall supply services, utilities, goods or materials or who have or shall have advanced funds to the Company after the Determination Date, but only to the extent of their claims in respect of the supply of such services, utilities, goods, materials or funds after the Determination Date and to the extent that such claims are not otherwise affected by the Plan. For greater certainty, a Restructuring Claim is not an Excluded Claim;

"Excluded Creditor" means a Person having a claim in respect of an Excluded Claim but only in respect of such Excluded Claim and to the extent that the Plan does not otherwise affect such claim;

"Fisher Group" means Kalman Fisher, 3482731 Canada Inc., 9318-5494 Quebec Inc. (formerly 7735235 Canada Inc.) and any Person who/which is, or was at any time within the 12 consecutive months immediately preceding the Determination Date, a Related Person to Kalman Fisher;

"Fisher Group Claims" means any Claims for payment of principal on any loans or advances made to the Company (or its predecessor) at any time prior to the Determination Date by any member of the Fisher Group, which includes any Secured Claim by any member of the Fisher Group, but excludes any Claims or future claims by any member of the Fisher Group for (i) salary (or other remuneration), (ii) reimbursement of business expenses, or (iii) payment of principal on any loans or advances in lieu of salary (or other remuneration);

"Gift Card Claim" has the meaning set forth in Section 2.3 (a)(v) hereof;

"Governmental Authority" means any (i) multinational, national, provincial, state, regional, municipal, local or other government, governmental or public department, ministry, central bank, court, tribunal, arbitral body, commission, board, official, minister, bureau or agency, domestic or foreign, (ii) subdivision, agent, commission, board or authority of any of the foregoing; or (iii) quasi-governmental or private body, including any tribunal, commission, regulatory agency or self-regulatory organization, exercising any regulatory, expropriation or taxing authority under, or for the account of, any of the foregoing;

"Initial Order" means the "Initial Order" issued by the Court on August 12, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"KERP Charge" has the meaning ascribed to such term in the Initial Order;

"KERP Claim" means a claim or any other indebtedness or obligation secured by the KERP Charge and "KERP Claims" means all of them;

"Laws" means all laws, statutes, codes, ordinances, decrees, rules, regulations, by-laws, judicial or arbitral or administrative or ministerial or departmental or regulatory judgments, injunctions, orders or decisions of any Governmental Authority, statutory body or self-regulatory authority, including general principles of law having the force of law and the term "applicable" with respect to such Laws and, in the context that refers to any Person, means such Laws as are applicable to such Person or its business, undertaking, property or securities and emanate from a Governmental Authority or self-regulatory authority having jurisdiction over the Person or its business, undertaking, property or securities;

"Meeting Date" means the date fixed for the Creditors' Meeting in accordance with the Meeting Procedure Order, or any subsequent Order, or any subsequent date following any adjournment of that meeting, as the case may be;

"Meeting Procedure Order" means the "Meeting Procedure Order" issued or to be issued by the Court upon or following the filing of this Plan with the Court:

"Monitor" means KPMG Inc., in its capacity as Monitor, as appointed by the Court pursuant to the Initial Order;

"Notice of Revision or Disallowance" has the meaning as set forth in the Claims Procedure Order;

"Order" means any order of the Court in the CCAA Proceedings;

"Original Currency" has the meaning set forth in Section 7.7 hereof;

"Payment Date" and "Payment Dates" have the meaning set forth in Section 2.4 hereof;

"Person" means any individual, corporation, limited or unlimited liability company, general or limited partnership, association, trust, trustee, executor, administrator, legal personal representative, estate, unincorporated organization, joint venture, governmental body or agency, or any other entity;

"Plan" means the present "Plan of Compromise and Arrangement" of the Company pursuant to the provisions of the CCAA, as may be amended, varied or supplemented by the Company from time to time in accordance with its terms;

"Plan Implementation Conditions" has the meaning set forth in Section 8.1 hereof;

"Plan Implementation Date" means the date, on or prior to the Plan Implementation Deadline, upon which all of the Plan Implementation Conditions have occurred or have been satisfied or waived;

"Plan Implementation Deadline" has the meaning set forth in Section 8.1 hereof;

"Proof of Claim" has the meaning set forth in the Claims Procedure Order;

"Proven Claim" means, in respect of a Creditor, the amount of the Claim of such Creditor as finally determined for distribution purposes in accordance with the provisions of this Plan, the CCAA and the Claims Procedure Order;

"Related Persons" has the meaning set forth in Section 4 of the BIA;

"Released Party" and "Released Parties" have the meanings as set forth in Section 6.2 hereof;

"Replacement Lender(s)" means the Person(s) who will provide the Replacement Loan;

"Replacement Loan" means the loans and/or other credit facilities to be provided to the Company by the Replacement Lender(s) in order to completely discharge and release (i) any then remaining DIP Claim and the DIP Lender Charge, and (ii) the Salus Claim and the Salus Security;

"Required Majorities" means the affirmative vote of a majority in number of the Affected Creditors voting in the Affected Creditors Class, having Voting Claims and voting on the Plan (in person or by proxy) at the Creditors' Meeting and representing not less than 66-2/3% in value of the Voting Claims of the Affected Creditors voting (in person or by proxy) at the Creditors' Meeting;

"Reserve(s)" means the reserve(s) to be established and maintained under this Plan by the Monitor by holding, on account of Disputed Claims, an amount equal to the aggregate amount that the holders of Disputed Claims would be entitled to receive if all such Disputed Claims had been Proven Claims as of any Payment Date;

"Resolution" means, collectively, one or any of, the resolutions providing for the approval of the Plan by the Affected Creditors;

"Restructuring Claim" means any right or claim of any Person against the Company in connection with any indebtedness or obligation of any kind owed to such Person arising out of the Company's restructuring under the CCAA and includes any right or claim resulting, directly or indirectly, from the consequences and effects of the Plan's acceptance by the Affected Creditors, the Plan's sanctions by the Sanction Order, the Plan's implementation and any debt forgiveness resulting from any of the foregoing; provided, however, that a Restructuring Claim shall not include an Excluded Claim. Without limiting the generality of the foregoing, a Restructuring Claim shall include a Disclaimed Landlord Claim, but shall exclude any claim of any landlord in respect of any lease for real or immovable property other than claims of Disclaimed Landlords. For greater certainty, a Restructuring Claim is an Affected Claim;

"Salus" means Salus Capital Partners, LLC or any Person acting on its behalf, in its place or as its successor;

"Salus Claim" means, at any given time, all indebtedness and obligations owing by the Company to Salus;

"Salus Security" means all mortgages, hypothecs, pledges, charges, liens, privileges, security interests, guarantees, suretyships or other rights securing and/or guaranteeing the Salus Claim, held by or on behalf of Salus (i) over any and all present and future property of the Company, any of the Stayed Parties, and/or any other Person, and/or (ii) against the Company, any of the Stayed Parties and/or any other Person;

"Sanction Date" means the date on which the Sanction Order is issued:

"Sanction Order" means the Order to be made under the CCAA sanctioning the Plan, as such Order may be affirmed, amended or modified by the Court at any time prior to the Plan Implementation Date or, if appealed, then, unless such appeal is withdrawn, abandoned or denied, as affirmed or amended on appeal, in form and content which is satisfactory to the Petitioner acting reasonably;

"Section 6(5) Claim" means any claim of an employee or former employee described in Section 6(5) of the CCAA but only to the extent of such amounts as required to be paid under the CCAA;

"Section 19(2) Claims" means any claim described in Section 19(2) of the CCAA;

"Secured Claim" means the Claim of a Secured Creditor, to the extent of the value of such Secured Creditor's security;

"Secured Creditor" has the meaning set forth in the CCAA, but only to the extent that such Creditor's mortgage, hypothec, pledge, charge, lien, privilege, security interest or other rights over the property of the Company was valid, opposable, perfected, and could be set up against third parties, including a trustee to the Company's bankruptcy, on both the Determination Date and the Claims Bar Date, failing which that Secured Creditor will be deemed to be an "unsecured creditor", as defined in the CCAA;

"Stayed Parties" means Boutique Laura Canada Ltée/Laura's Shoppe Canada Ltd., 3482731 Canada Inc., 9318-5494 Quebec Inc. (formerly 7735235 Canada Inc.) and Kalman Fisher;

"Taxes" means all federal, state, provincial, territorial, county, municipal, local or foreign taxes, duties, imposts, levies, assessments, tariffs and other charges imposed, assessed or collected by a Governmental Authority, including (i) any gross income, net income, gross receipts, business, royalty, capital, capital gains, goods and services, value added, severance, stamp, franchise, occupation, premium, capital stock, sales and use, real property, land transfer, personal property, ad valorem, transfer, licence, profits, windfall profits, environmental, payroll, employment, employer health, pension plan, anti-dumping, countervail, excise, severance, stamp, occupation, or premium tax, (ii) all withholdings on amounts paid to or by the relevant Person, (iii) all employment insurance premiums, Canada, Quebec and any other pension plan contributions or premiums, (iv) any fine, penalty, interest, or addition to tax, (v) any tax imposed, assessed, or collected or payable pursuant to any tax-sharing agreement or any other contract relating to the sharing or payment of any such tax, levy, assessment, tariff, duty, deficiency, or fee, and (vi) any liability for any of the foregoing as a transferee, successor, guarantor, or by contract or by operation of law;

"Unaffected Claim" means any right of any Person in connection with any indebtedness, liability or obligation of any kind of the type described in Section 2.3 (a) hereof. For greater certainty, the Unaffected Claims include the Excluded Claims;

"Unaffected Creditor" means a Person having a Claim in respect of an Unaffected Claim, but only in respect of such Unaffected Claim, and for greater certainty, includes an Excluded Creditor;

"Voting Claim" means, in respect of a Creditor, the amount of such Creditor's claim which has been accepted for voting purposes in accordance with the provisions of this Plan, the Claims Procedure Order and the CCAA;

1.2 Interpretation

For purposes of this Plan:

- (a) any reference in this Plan to a contract, instrument, release, indenture, agreement or other document being in a particular form or on particular terms and conditions means that such document shall be substantially in such form or substantially on such terms and conditions;
- (b) any reference in this Plan to an existing document or exhibit filed or to be filed means such document or exhibit as it may have been or may be amended, modified or supplemented;
- (c) all references to currency and to "\$", "C\$" or "Dollars" are to Canadian dollars except as otherwise indicated;
- (d) all references in this Plan to Sections are references to Sections of this Plan;
- (e) unless otherwise specified, the words "hereof", "herein" and "hereto" refer to this Plan in its entirety rather than to any particular portion of this Plan;
- (f) the division of this Plan into Sections and paragraphs and the insertion of captions and headings to Sections and paragraphs are for convenience of reference only and are not intended to affect the interpretation of, or to be part of this Plan;
- (g) where the context requires, a word or words importing the singular shall include the plural and vice versa and a word or words importing one gender shall include all genders;
- (h) the words "includes" and "including" are not limiting; and
- (i) the word "or" is not exclusive.

1.3 Date For Any Action

In the event that any date (including any Payment Date) on which any action (including any payment) is required to be taken under this Plan by any of the parties is not a Business Day, that action (including any payment) shall be required to be taken on the next succeeding day which is a Business Day.

1.4 Statutory References

Any reference in this Plan to a statute includes all regulations made thereunder and all amendments to such statute or regulations in force, from time to time, or any statute or regulations that supplement or supersede such statute or regulations.

SECTION 2 COMPROMISE AND ARRANGEMENT

2.1. Persons Affected

The purpose of this Plan is to provide for the full and final settlement of all Affected Claims. Except as specifically provided for in this Plan, this Plan will become effective on the Plan Implementation Date in accordance with its terms, and all Affected Claims against the Company will be fully and finally, settled, compromised and released in this Plan upon the Aggregate Distribution being fully paid to the Monitor, it being understood that any Affected Claim that is paid in full pursuant to this Plan before such date shall be settled, compromised and released on the date of such payment. This Plan shall be binding on and enure to the benefit of the Company, the Affected Creditors, the Released Parties, any trustee, agent or other Person acting on behalf of any Affected Creditor and such other Persons who have received the benefit of, or are bound by any waivers, releases or indemnities hereunder.

2.2. Classes of Affected Claims

There shall only be one class of Affected Creditors for the purpose of voting on, and receiving distributions pursuant to this Plan, being the "Affected Creditors Class".

2.3. Unaffected Claims

- (a) This Plan does not affect the following claims (each, an "Unaffected Claim" and, collectively, the "Unaffected Claims"), the holders of which will not be entitled to vote at the Creditors' Meeting or receive any distributions under this Plan, namely:
 - (i) the Excluded Claims;
 - (ii) any Administration Claim;
 - (iii) any DIP Claim;
 - (iv) any KERP Claims;

- (v) any claim with respect to gift-cards, gift certificates, lay-away deposits and other customer certificates (collectively, "Gift Card Claims");
- (vi) Crown Priority Claims;
- (vii) Section 6(5) Claims;
- (viii) Section 19(2) Claims;
- (ix) Secured Claims:
- (x) the Salus Claim; and
- (xi) Fisher Group Claims,

all of which will be dealt with in Section 3 hereof;

(b) Nothing in this Plan shall affect the Company's rights and defences, both legal and equitable, with respect to any Unaffected Claim including any rights arising under or pursuant to the Claims Procedure Order or this Plan or any rights with respect to legal and equitable defences or entitlements to set-offs or recoupments against such Unaffected Claims.

2.4. <u>Treatment of Affected Claims</u>

By no later than each of the payment dates hereafter set forth (collectively the "Payment Dates" or individually a "Payment Date"), the aggregate sum of \$4 Million (the "Aggregate Distribution"), without any interest whatsoever thereon, shall be paid by the Company to the Monitor and shall thereafter be distributed by the Monitor to the Affected Creditors, according to the amount of their respective Proven Claims as hereafter set forth, namely:

- (a) by no later than April 30, 2016, \$1 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor as follows:
 - (i) an amount equal to the lesser of (A) the Proven Claim of each Affected Creditor, or (B) \$1,250, shall be distributed by the Monitor to the Affected Creditors, according to the amount of their Proven Claims; and
 - (ii) an amount equal to the difference between (A) \$1 Million, and (B) the aggregate amount to be distributed by the Monitor pursuant to Section 2.4 (a)(i) above, shall be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their

respective Proven Claims, less any amounts received in respect of the amounts set forth in Section 2.4 (a)(i) above;

- (b) by no later than July 31, 2016, \$1 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distribution under Section 2.4 (a) hereof;
- by no later than December 15, 2016, \$1.8 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distributions under Sections 2.4 (a) and 2.4 (b) hereof; and
- (d) by no later than February 15, 2017, \$200,000 shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distributions under Sections 2.4 (a), 2.4 (b) and 2.4 (c) hereof.

2.5. Settlement of Affected Claims

The payment of the entire Aggregate Distribution to the Monitor as set forth in Section 2 hereof shall constitute the full and final payment, reduction, settlement and transaction (as envisaged by the relevant provisions of the *Civil Code of Quebec*) of all of the Affected Claims of all of the Affected Creditors and engage the releases and other provisions set forth in this Plan.

2.6. Extension of Payment Dates

Any or all of the Payment Dates may be extended by the Company with the consent of the Monitor, by additional periods not exceeding, in each case, 60 days. On application by the Company or, the Monitor made prior to a Payment Date or, if applicable, prior to the expiry of any extension thereof as described above, the Court may issue Orders extending (or further extending) any or all of the Payment Dates for any additional period(s) deemed appropriate by the Court in the then existing circumstances.

2.7. Prepayment Rights

The Company shall, at all times prior to the Final Distribution Date, be entitled to pre-pay the whole or any amount of the Aggregate Distribution prior to any Payment Date, whereupon the Amount(s) of any such pre-payment(s) shall be distributed by the Monitor

to the Affected Creditors in accordance with the provisions of Section 2.4 hereof, mutatis mutandis.

SECTION 3 TREATMENT OF OTHER CLAIMS AND MATTERS

3.1 Treatment of the DIP Claim

On or before the earlier of (i) the Plan Implementation Date, or (ii) the date on which the DIP Claim becomes payable pursuant to the provisions of the DIP Facility Documents, the DIP Lender shall receive full payment of any remaining DIP Claim.

3.2 Treatment of Excluded Claims

Subject to **Section 3.1** hereof, the Excluded Claims (other than the Administration Claim, the DIP Claim and the KERP Claim) will remain in full force and effect in accordance with their terms after the Plan Implementation Date, and will be paid in full by the Company in the normal course of its business as and when they become due.

3.3 Treatment of Administration Claims

All Administration Claims, if any, will be paid in full by the Company as and when they become due, and any remaining balance will be settled as soon as practicable after the Plan Implementation Date.

3.4 Treatment of Gift Card Claims

Gift Card Claims will be honored in accordance with the terms of the relevant gift card, gift certificate, lay-away deposit or other customer certificates upon presentation of such gift card, gift certificate, lay-away deposit or other customer certificates by the holder at any of the Petitioner's retail stores.

3.5 <u>Treatment of Certain Crown Priority Claims</u>

All Crown Priority Claims described in Section 6(3) of the CCAA, if any, will be paid in full by the Petitioner within 6 months immediately following the Sanction Date;

3.6 <u>Treatment of Section 6(5) Claims</u>

All Section 6(5) Claims, if any, will be paid in such amounts as required under the CCAA immediately after the Sanction Date.

3.7 Treatment of Section 19(2) Claims

All Section 19(2) Claims, if any, will be paid by the Company as and when they become due, unaffected by this Plan.

3.8 Treatment of KERP Claims

Each KERP Claim will be paid by the Company as and when it becomes due.

3.9 <u>Treatment of Secured Claims</u>

Secured Claims (other than the Administration Claim, the DIP Claim, the KERP Claims and the Salus Claim) will be dealt with as provided for under agreements entered into or as may be hereafter entered into between the Company and the relevant Secured Creditor, or as may be ordered by the Court. For greater certainty, Secured Claims will be unaffected by this Plan.

3.10 Treatment of Salus Claim

The Salus Claim and the Salus Security will be dealt with as provided for under agreements entered into or as may be hereafter entered into between the Company and Salus, or as may be ordered by the Court. For greater certainty, the Salus Claim will be unaffected by this Plan.

3.11 Treatment of Fisher Group Claims

Conditional upon the enactment of a Resolution approving the Plan, issuance of the Sanction Order and implementation of the Plan as set forth in **Section 8** hereof, each member of the Fisher Group:

- subordinates and postpones all of the Fisher Group Claims to and in favour of full payment of the Distribution Amount to the Affected Creditors, such that no member of the Fisher Group shall be entitled to receive any payment from the Company of any principal on any of the Fisher Claims unless and until the full Aggregate Distribution shall have been paid to the Monitor as set forth in the Plan; and
- (b) waives and renounces to any right to be an Affected Creditor.

SECTION 4 VALUATION OF CLAIMS, CREDITORS' MEETING AND RELATED MATTERS

4.1 Conversion of Affected Claims into Canadian Currency

For the purposes of determining the value of Affected Claims denominated in currencies other than Canadian dollars for voting purposes, such Affected Claims shall be converted by the Monitor to Canadian dollars at the Bank of Canada noon spot rate of exchange for exchanging currency to Canadian dollars on the Determination Date (which in the case of US Dollars was C\$1.3080 for US\$1.00).

4.2 Affected Claims

Affected Creditors shall be entitled to prove their respective Affected Claims, vote their Voting Claims in respect of the Plan, and, if their Affected Claims become Proven Claims, receive the distributions, all as set forth in the Claims Procedure Order and this Plan.

All amounts recognized as Voting Claims or Proven Claims shall be net of any amount that the Company is entitled to offset, recoup, compensate or otherwise apply in reduction of such amounts.

4.3 Creditors' Meeting

The Creditors' Meeting shall be held in accordance with this Plan, the Meeting Procedure Order, and any further Order which may be made from time to time for the purposes of considering and voting on the Resolution or other matters to be considered at the Creditors' Meeting.

4.4 Approval by Affected Creditors

The Company will seek approval of the Plan by the affirmative vote of the Required Majorities. The Resolution to be voted on at the Creditors' Meeting will be decided by the Required Majorities on a vote as set forth in the Meeting Procedure Order. The result of any vote will be binding on all Affected Creditors, whether or not any such Affected Creditor is present and voting (in person or by proxy) at the Creditors' Meeting.

4.5 Claims Bar Date

If an Affected Creditor holding an Affected Claim has failed to file its Proof of Claim prior to the Claims Bar Date and has not been permitted to file a late claim pursuant to an Order, that Affected Creditor shall be barred from voting at the Creditors' Meeting, that Affected Creditor shall have no right to receive any distributions of the Aggregate

Distribution hereunder, the Company shall be released from the Affected Claims of such Creditor and Section 6.2 of this Plan shall apply to all such Affected Claims.

SECTION 5 PROCEDURE FOR RESOLVING DISTRIBUTIONS IN RESPECT OF DISPUTED CLAIMS

5.1 No Distributions Pending Allowance

Notwithstanding any other provision of this Plan, no distributions of the Aggregate Distribution hereunder shall be made by the Monitor with respect to a Disputed Claim unless and until it has become a Proven Claim. Disputed Claims shall be dealt with in accordance with the Claims Procedure Order and this Plan.

5.2 <u>Distribution From Reserve Once Disputed Claims Resolved</u>

The Monitor shall make allocations from the Reserve to holders of Disputed Claims following any of the Payment Dates in accordance with this Plan. To the extent that Disputed Claims become Proven Claims after any relevant Payment Date, the Monitor shall, from time to time at its sole discretion, distribute from the Reserve to the holders of such Proven Claims, the amount which they would have been entitled to receive in respect of such Proven Claims had such Claims been Proven Claims on any relevant Payment Date. To the extent that any Disputed Claim or a portion thereof has become a Disallowed Claim, then the Monitor shall, at any time which it deems appropriate in the circumstances, distribute, to the holders of Proven Claims, their *pro rata* share from the Reserve of such additional amount kept in the Reserve on account of any such Disallowed Claim.

SECTION 6 EFFECT OF THE PLAN AND RELEASES

6.1 Effect of the Plan

Upon the Aggregate Distribution being fully remitted to the Monitor, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Performance"), whereupon all Affected Claims shall be deemed to be fully and finally settled, compromised, released and transacted (as envisaged by the relevant provisions of the Civil Code of Quebec), as of the date of issuance of the Certificate of Performance, subject only to an Affected Creditor's right to recover the distributions under this Plan, except for Claims as may have been fully paid prior to such date, in which case those Affected Claims will be deemed to be settled as of such payment.

6.2 Releases

Upon the implementation of this Plan on the Plan Implementation Date, each of:

- (a) the Company;
- (b) the Monitor and its legal counsel in the CCAA Proceedings;
- (c) the Directors and all present and future officers and employees, legal counsel, financial advisors, consultants and agents of the Company (and their respective directors, officers and employees); and
- (d) the DIP Lender and its legal counsel in the CCAA Proceedings,

(each a "Released Party" and collectively the "Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Person (including any Secured Creditor in respect of each Released Party, except the Company and solely in respect to its Secured Claim), may be entitled to assert (including any and all Claims in respect of statutory liabilities and any Directors' Claims of all Directors and all present and former officers and employees of the Company and any alleged fiduciary or other duly), whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with Claims, the business and affairs of the Company, this Plan and the CCAA Proceedings, and all claims arising out of such actions or omissions shall be forever waived and released (other than the right to enforce the Company's obligations under the Plan or any related document), all to the full extent permitted by law.

6.3 Exception to Releases

Nothing set forth in Section 6.2 hereof shall:

- (a) release or discharge a Released Party from an Unaffected Claim;
- (b) release or discharge the Company from or in respect of its obligations under this Plan;
- (c) affect the right of any Person:
 - to recover an indemnity from any insurance coverage under which that Person is an insured; or

- (ii) to obtain recovery on a claim or liability against a Released Party from any insurance coverage pursuant to which that Released Party is an insured, but, for certainty, any claim or liability to which an insurer is or would otherwise be subrogated as against the Company is released hereunder and the recovery to which such Person shall be entitled under such insurance coverage shall be limited to the proceeds of insurance actually paid by the insurer with respect to such claim or liability;
- (d) release or discharge the Directors with respect to matters set out in Section 5.1(2) of the CCAA; or
- (e) release or discharge any Person (whether or not a Director, officer or employee of the Company) other than the Company from any obligations arising from any guarantee furnished by such Person to and in favour of any Affected Creditor.

6.4 Set-Off Rights

Notwithstanding the releases set forth under the Plan, in general, and Sections 6.1 and 6.2 hereof, in particular, all Claims shall remain subject to any rights of set-off that otherwise would be available to the Person against whom a Claim is asserted.

6.5 <u>Injunction Related to Releases</u>

The Sanction Order will enjoin the prosecution by or on behalf of any Person, whether directly, derivatively or otherwise, of any claim, obligation, suit, judgment, damage, demand, debt, right, cause of action, liability or interest released, discharged or terminated pursuant to this Plan.

6.6 Waiver of Defaults

From and after the Plan Implementation Date, all Persons shall be deemed to have waived any and all defaults of the Company then existing or previously committed by the Company or caused by the Company, directly or indirectly, or non-compliance with any covenant, positive or negative pledge, warranty, representation, term, provision, condition or obligation, express or implied, in any contract, credit document, agreement for sale, lease or other agreement, written or oral, and any and all amendments or supplements thereto, existing between such Person and the Company arising from the Company's insolvency, the Company's filing a notice of intention to make a proposal under the *BIA*, the filing by the Company under the *CCAA* or the transactions contemplated by this Plan or otherwise, and any and all notices of default and demands for payment under any instrument, including any guarantee arising from such default, shall be deemed to have been rescinded.

SECTION 7 PROVISIONS GOVERNING DISTRIBUTIONS

7.1 Partial Distributions for Claims Allowed

Except as otherwise provided herein or as ordered by the Court, distributions of the Aggregate Distribution shall be made by the Monitor at the time and in the manner deemed reasonable by the Monitor, and partial distributions may be made prior to receipt of the full amount of the Aggregate Distribution. In such a case, all such partial payments shall represent the *pro rata* amount of the distribution to which the holders of a Proven Claim would otherwise be entitled to receive.

7.2 <u>Currency to be used for the Distribution</u>

For the purposes of determining any Claims denominated in currencies other than Canadian Dollars for distribution purposes, such Claims shall be converted by the Monitor to Canadian dollars at the Bank of Canada noon spot rate of exchange for exchanging currency to Canadian dollars on the Determination Date (which in the case of US Dollars was C\$1.3080 for US\$1.00).

7.3 <u>Assignment of Claims</u>

For purposes of determining entitlement to receive a distribution pursuant to this Plan, the Company and the Monitor and each of their respective agents, successors and assigns shall have no obligation to recognize any transfer of Claims except as provided for under the Claims Procedure Order or the Meeting Procedure Order.

7.4 Interest on Affected Claims

Except as specifically provided in the Plan or the Sanction Order, interest shall not be treated as accruing on account of any Affected Claims for purposes of determining the allowance and distribution of such Affected Claim. To the extent that any Proven Claim to which a distribution under this Plan relates is comprised of indebtedness and accrued but unpaid interest thereon, such distribution shall, to the extent permitted by applicable Laws, be allocated for Tax purposes to the principal amount of such Proven Claim (including any secured and unsecured portion(s) of the principal amount of such Proven Claim) first and then, to the extent that the consideration exceeds the principal amount of the Proven Claim, to the portion of such Proven Claim representing accrued but unpaid interest (including interest in respect of any secured portion of such Proven Claim).

7.5 <u>Distributions by Monitor</u>

The Monitor shall make all distributions required under this Plan in accordance with the provisions of this Plan, in general, and the provisions of Section 5 and Section 7 hereof.

in particular. The Monitor shall receive, without further Court approval, reasonable compensation from the Company for distribution services rendered pursuant to the Plan.

7.6 <u>Delivery of Distributions</u>

- (a) Subject to Section 7.3 hereof, distributions shall be made by the Monitor (i) at the addresses set forth on the Proof of Claim form filed by the Affected Creditors (or at the last known addresses of such Affected Creditors if no Proof of Claim form was filed or if the Company or the Monitor has been notified in writing of a change of address), (ii) at the addresses set forth in any written notice of address change delivered to the Monitor after the date of any related Proof of Claim, or (iii) in a registered retirement savings plan account designated by any employee of the Company;
- (b) If any distribution to a Creditor is returned as undeliverable, no further distributions to such Creditor shall be made unless and until the Monitor is notified of the then-current address of such Creditor, at which time all missed distributions shall be made to such Creditor without interest. The Company shall make reasonable efforts to locate Affected Creditors for which distributions were undeliverable. Any claim for undeliverable distributions must be made on or before the later to occur of (i) 3 months after the Payment Date, or (ii) 3 months after such Creditor's Claim becomes a Proven Claim, after which date all unclaimed property shall revert to the Company free of any restrictions or claims thereon and the claim of any Creditor with respect to such property shall be discharged and forever barred, notwithstanding any applicable Laws to the contrary.

7.7 No Double Recovery

The aggregate recovery on account of any Proven Claim from all sources, regardless of whether on account of a theory of primary or secondary liability, by reason of guarantee, surety, indemnity, joint and several obligation or otherwise, shall not exceed (i) 100% of the underlying indebtedness, liability or obligation giving rise to such Claim or, (ii) where the underlying indebtedness, liability or obligation giving rise to such Claim is denominated in a currency (the "Original Currency") other than Canadian dollars, 100% of such underlying indebtedness, liability or obligation after conversion of the value of the distributions received in Canadian dollars back to the Original Currency at the Bank of Canada noon spot rate of exchange for exchanging Canadian dollars to the Original Currency on the Determination Date.

7.8 Withholding Requirements

In connection with the Plan, any distribution made hereunder by the Monitor shall be made net of all applicable Taxes. Notwithstanding any other provision of the Plan, each

Creditor that is to receive a distribution pursuant to the Plan shall have sole and exclusive responsibility for the satisfaction and payment of any Taxes imposed by any Governmental Authority (including income, withholding and other Taxes on account of such distribution). The Monitor, as necessary, shall be authorized to take any and all actions as may be necessary or appropriate to comply with such withholding and reporting requirements. All amounts withheld on account of Taxes shall be treated for all purposes as having been paid to the Creditor in respect of which such withholding was made, provided such withheld amounts are remitted to the appropriate Governmental Authority.

SECTION 8 IMPLEMENTATION OF THE PLAN

8.1 Conditions Precedent to Implementation of Plan

The implementation of this Plan by the Company is subject to the occurrence and/or fulfillment of each of the following conditions precedent (the "Plan Implementation Conditions") on or prior to the December 15, 2015 or such later date as may be ordered by the Court (the "Plan Implementation Deadline"), namely:

- (a) this Plan shall have been approved by Resolution enacted by the Required Majorities;
- (b) the Sanction Order sanctioning this Plan shall have been issued, and not then subject to any appeal therefrom, or the Sanction Order shall have been rendered executory notwithstanding any appeal, and the operation and effect of the Sanction Order shall not have been stayed, reversed or amended, and shall, among other things, declare and order that:
 - (i) this Plan, including the compromises and arrangements set out herein, is sanctioned and approved pursuant to Section 6 of the CCAA and, as at the Plan Implementation Date, will be effective and will enure to the benefit of and be binding upon the Company, the Affected Creditors and all other Persons referred to in this Plan or in the Sanction Order;
 - (ii) (A) this Plan has been approved by the Required Majorities of Affected Creditors of the Company in conformity with the CCAA; (B) the Company has complied with the provisions of the CCAA and the Orders of the Court made in the CCAA Proceedings in all respects; (C) the Court is satisfied that the Company has neither done nor purported to do anything that is not authorized by the CCAA; and (D) this Plan and the transactions contemplated thereby are fair and reasonable;

- (iii) the full and final release and discharge of the Claims will be effective and will enure to the benefit of and be binding upon the Company, the Affected Creditors and all other Persons referred to in this Plan upon the issuance of the Certificate of Performance;
- (iv) the Company and the Monitor are authorized to take all steps and actions necessary to implement this Plan;
- (v) all Proven Claims determined in accordance with the Claims Procedure Order are final and binding on the Company and all Affected Creditors;
- (vi) any Claims for which a Proof of Claim has not been filed by the Claims Bar Date shall be forever barred and extinguished;
- (vii) all distributions and payments by or at the direction of the Monitor, in each case on behalf of the Company, under the Plan are for the account of the Company and the fulfillment of its obligations under the Plan;
- (viii) the Company and the Monitor may apply to the Court for advice and direction in respect of any matters arising from or under the Plan;
- subject to the performance by the Company of its obligations under this Plan, all contracts, leases, agreements and arrangements to which the Company are a party and that have not been disclaimed, terminated or repudiated pursuant to the Initial Order and/or the CCAA will be and remain in full force and effect, unamended, as at the Plan Implementation Date, and no Person who is a party to any such contract, lease, agreement or other arrangement may accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of dilution or other remedy) or make any demand under or in respect of any such contract, lease, agreement or other arrangement and no automatic termination will have any validity or effect, by reason of:
 - (A) any event that occurred on or prior to the Plan Implementation Date and is not continuing that would have entitled such Person to enforce those rights or remedies (including defaults, events of default, or termination events arising as a result of the insolvency of the Company);
 - (B) the insolvency of the Company, the fact that the Company filed a notice of intention to make a proposal under the *BIA* or the fact that the Company sought or obtained relief under the *CCAA*; or

- (C) any compromises or arrangements effected pursuant to this Plan or any action taken or transaction effected pursuant to this Plan;
- the releases set forth in Section 6.2 hereof and the injunction set forth in Section 6.5 hereof are implemented and in effect; and
- (xi) the commencement or prosecution, whether directly, derivatively or otherwise, or any demands, claims, actions, causes of action, counterclaims, suits or any indebtedness, liability, obligation or cause of action released and discharged pursuant to this Plan are enjoined;
- (c) the Replacement Loan shall have been disbursed and each of the DIP Claim, the DIP Lender Charge, the Salus Claim and the Salus Security shall have been discharged and released.

8.2 Waiver of Conditions

Each of the Plan Implementation Conditions enure solely to the Company's benefit and, except for each of the conditions set forth in Sections 8.1 (a) and 8.1 (b)(i) hereof, may be waived in whole or in part either (i) solely by the Company (and no other Person and without the consent of any other Person), without any other notice to parties in interest or the Court and without a hearing, or (ii) by the Court. The failure to satisfy or waive any of the Plan Implementation Conditions prior to the Plan Implementation Date may be asserted by the Company regardless of the circumstances giving rise to the failure of such Plan Implementation Conditions to be satisfied (including any action or inaction by the Company). The failure of the Company to exercise any of the foregoing rights shall not be deemed a waiver of any other rights, and each such right shall be deemed an ongoing right that may be asserted at any time.

8.3 <u>Certificate of Implementation</u>

Upon the occurrence and/or fulfillment of all of the Plan Implementation Conditions (or waiver thereof, as the case may be) on or prior to the Plan Implementation Deadline, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Implementation").

8.4 Effect of Plan

Upon the issuance of the Certificate of Performance, the settlement of the Affected Claims in accordance with the Plan shall become final and binding on the Company, and all the Affected Creditors and their respective successors and assigns, and this Plan shall result in the full and final settlement of all Affected Claims, including for greater certainty, any Restructuring Claim and any Claims resulting, directly or indirectly, from the consequences and effects relating to the acceptance of the Plan by the Affected

Creditors, its sanction by the Court, or its implementation, and any indebtedness, obligations or undertakings which the Company may subsequently become subject to, directly or indirectly, as a result of an obligation, transaction or an event that occurred before the Determination Date, as well as any indebtedness, obligations or any undertakings that the Company may be subject to on any date whatsoever in connection with the Plan, the approval thereof by the Court or the implementation thereof. For greater certainty, this Plan shall not affect or impair any rights, remedies and recourses which the Company had, has or may have after the issuance of the Certificate of Performance in connection with transactions, facts or obligations existing prior to the Determination Date;

8.5 Nullity of Plan

In the event that all of the Plan Implementation Conditions have not occurred and/or been fulfilled (or waived, as the case may be) on or prior to the Plan Implementation Deadline, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Non-Implementation"). Upon the issuance of such Certificate of Non-Implementation, this Plan shall automatically become null, void and of no effect whatsoever and neither the Company, the Affected Creditors, the Monitor nor any other Person affected by this Plan shall be bound, obliged or affected by any of the provisions of this Plan.

SECTION 9 MISCELLANEOUS

9.1 <u>Confirmation of Plan</u>

Upon occurrence and/or fulfillment of all of the Plan Implementation Conditions (or waiver thereof, as the case may be) on or prior to the Plan Implementation Deadline, this Plan shall be binding upon the Company, the Affected Creditors and any and all other Persons affected by the provisions of this Plan as well as each of their respective successors and assigns.

9.2 Paramountcy

From and after the Plan Implementation Date, any conflict between this Plan and/or the covenants, warranties, representations, terms, conditions, provisions or obligations, express or implied, of any contract, mortgage, security agreement, and/or indenture, trust indenture, loan agreement, commitment letter, agreement for sale, the by-laws of the Company, lease or other agreement, undertaking or any other source of obligations, written or oral and any and all amendments or supplements thereto existing between one or more of the Affected Creditors and the Company as at the Plan Implementation Date will be deemed to be governed by the terms, conditions and provisions of this Plan and the Sanction Order, which shall take precedence and priority. For greater certainty,

all Affected Creditors shall be deemed irrevocably for all purposes to consent to all transactions contemplated in and by this Plan.

9.3 <u>Modification of Plan</u>

The Company:

- (a) in consultation with the Monitor and the DIP Lender (if any DIP Claim remains outstanding), reserves the right to file any modification of, or amendment or supplement to, this Plan by way of supplementary plan or plans of reorganization, compromise or arrangement (or any one or more thereof) at or before the Creditors' Meeting, in which case any such supplementary plan or plans of reorganization, compromise or arrangement (or an one or more thereof), shall, for all purposes, be and be deemed to form part of and be incorporated into this Plan. The Company shall file any supplementary plans with the Court as soon as practicable. The Company shall give notice to Affected Creditors of the details of any such modification, amendment or supplement at the Creditors' Meeting prior to the vote being taken to approve this Plan. The Company may give notice of a proposed modification, amendment or supplement to this Plan at or before the Creditors' Meeting by notice which shall be sufficient if given to those Affected Creditors present at such meeting in person or by proxy; and/or
- (b) after the Creditors' Meeting (and both prior to and subsequent to the obtaining of the Sanction Order), the Company, in consultation with the Monitor and, if prior to the Plan Implementation Date, the DIP Lender (if any DIP Claim then remains), at any time and from time to time vary, amend, modify or supplement this Plan, (except to reduce the amount of the Aggregate Distribution or extend any Payment Dates other than as expressly provided herein), without the need for obtaining an Order of the Court or providing notice to the Affected Creditors if the Monitor determines that such variation, amendment, modification or supplement would not be materially prejudicial to the interests of the Affected Creditors under this Plan or the Sanction Order and is necessary in order to give effect to the substance of this Plan or the Sanction Order. All of the foregoing shall not require any further vote by or approval by the Affected Creditors or any approval by the Court.

9.4 Deeming Provisions

In this Plan, any deeming provisions are not rebuttable and are conclusive and irrevocable.

9.5 <u>Sections 95 to 101 BIA</u>

Notwithstanding Section 36.1 of the *CCAA*, Section 38 and Sections 95 through 101 of the *BIA* shall not apply to this Plan. Accordingly, neither the Monitor, any Creditor nor any other Person shall be entitled to exercise any right, remedy or recourse, or to commence any action, motion, application or any other proceeding against any member of the Fisher Group, any Creditor or any other Person in relation to the Company, based on Section 38 and Sections 95 through 101 of the *BIA*.

9.6 Responsibilities of the Monitor

The Monitor is acting in its capacity as Monitor in the CCAA Proceedings with respect to the Company and not in its personal or corporate capacity and will not be responsible or liable for any responsibilities or obligations of the Company under this Plan or otherwise, including with respect to the making of distributions or the receipt of any distribution by any Creditor or any other Person pursuant to the Plan. The Monitor will have the powers and protections granted to it by this Plan, the CCAA, the Initial Order, the Claims Procedure Order, the Meeting Procedure Order, the Sanction Order and any other Orders.

9.7 <u>Liability Limitations</u>

The Monitor, the Company and their respective legal counsel and other professional advisors, shall not be liable to any Person for any act or omission in connection with, or arising out of, the CCAA Proceedings, the Plan, the pursuit of sanctioning of the Plan, the consummation and implementation of the Plan or the administration of the Plan or the funds to be distributed under the Plan, except for their own wilful misconduct or gross negligence.

9.8 Notices

(a) Any notices or communication to be made or given hereunder to the Company or the Monitor shall be in writing and shall refer to this Plan and may, subject as hereinafter provided, be made or given by personal delivery, by courier, by prepaid mail, by fax or by e-mail, addressed to the respective parties as follows:

(i) if to the Company:

LAURA'S SHOPPE (P.V.) INC.

2955 Jules-Brillant Laval, Quebec H7P 6B2

Attention:

Kalman Fisher and

Josh Fisher

E-Mails:

kfisher@laura.ca

josh.fisher@laura.ca

Fax:

(450) 973-6099

with copy to:

KUGLER KANDESTIN LLP

1 Place Ville Marie Suite 2101 Montreal, Quebec H3B 2C6

Attention:

Gerald F. Kandestin,

David Stolow and Jeremy Cuttler

E-Mail:

gkandestin@kklex.com

dstolow@kklex.com jcuttler@kklex.com

Fax:

(514) 875-8424

(ii) if to the Monitor:

KPMG LLP KPMG Tower 600 De Maisonneuve Blvd., West Suite 1500 Montreal, Quebec H3A 0A3

Attention:

Dev Coossa

E-Mail:

dcoossa@kpmg.ca

Fax:

(514) 840-2121

with copy to:

FISHMAN FLANZ MELAND PAQUIN, L.L.P. 1250 Rene-Levesque Blvd., West Suite 4100 Montreal, Quebec H3B 4W8

Attention: Av

Avram Fishman and

Mark Meland

E-Mails:

afishman@ffmp.ca

mmeland@ffmp.ca

Fax:

(514) 932-4170

or to such other address as any party may from time to time notify the others in accordance with this Section 9.8. In the event of any strike, lock-out or other event which interrupts postal service in any part of Canada, all notices and communications during such interruption may only be given or made by personal delivery, fax or e-mail and any notice or other communication given or made by prepaid mail within the 5 Business Day period immediately preceding the commencement of such interruption, unless actually received, shall be deemed not to have been given or made. All such notices and communications shall be deemed to have been received, in the case of notice by delivery, fax or e-mail prior to 5:00 p.m. (local time) on a Business Day, when received or if received after 5:00 p.m. (local time) on a Business Day or at any time on a non-Business Day, on the next following Business Day and, in the case of notice mailed as aforesaid, on the fourth Business Day following the date on which such notice or other communication is mailed. The unintentional failure by the Company or the Monitor to give any notice contemplated hereunder to any particular Creditor shall not invalidate this Plan or any action taken by any Person pursuant to this Plan.

(b) Any notices or communication to be made or given hereunder by the Monitor or the Company to a Creditor may be sent by fax, e-mail, ordinary mail, registered mail, courier or facsimile transmission to the e-mail address, address or fax number specified by such Creditor in its Proof of Claim, or in any subsequent written notice of change of address given to the Monitor. A Creditor shall be deemed to have received any document sent pursuant to this Plan four (4) Business Days after the document is sent by ordinary or registered mail and on the Business Day immediately following the day on which the document is sent by courier, fax or e-mail.

9.9 Severability of Plan Provisions

If, prior to the Sanction Date, any term or provision of this Plan is held by the Court to be invalid, void or unenforceable, the Court, at the request of the Company which request shall be made in consultation with the Monitor and the DIP Lender (if any DIP Claim then remains), shall have the power to either:

- (a) sever such term or provision from the balance of this Plan and provide the Company with the option to proceed with the implementation of the balance of this Plan as of and with effect from the Plan Implementation Date; or
- (b) alter and interpret such term or provision to make it valid and enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpreted.

Notwithstanding any such severing, holding, alteration or interpretation, and provided the Company proceeds with the implementation of this Plan, the remainder of the terms and provisions of this Plan shall remain in full force and effect and shall in no way be affected, impaired or invalidated by such severing, holding, alteration or interpretation.

9.10 Revocation, Withdrawal or Non-Consummation

The Company, upon consultation with the Monitor and the DIP Lender, reserves the right to revoke or withdraw this Plan at any time prior to the Sanction Date and to file subsequent plans of arrangement and compromise. If the Company revokes or withdraws this Plan, or if the Sanction Order is not issued:

- (a) this Plan shall be null, void and inoperative in all respects;
- (b) any Claim, any settlement or compromise embodied in this Plan (including the fixing or limiting of any Claim to an amount certain), assumption or termination by this Plan, and any document or agreement executed pursuant to this Plan shall be deemed null, void and inoperative; and
- (c) nothing contained in this Plan, and no act taken in preparation for consummation of this Plan, shall:
 - constitute or be deemed to constitute a waiver or release of any Claims by or against the Company or any other Person;
 - (ii) prejudice in any manner the rights of the Company or any Person in any further proceedings involving the Company; or

(iii) constitute an admission of any sort by the Company or any other Person.

9.11 Further Assurance

Notwithstanding that the transactions and events set out in this Plan shall occur and be deemed to occur in the order set out herein without any additional act or formality, each of the Persons affected hereby shall make, do and execute, or cause to be made, done and executed at the cost of the requesting party, all such further acts, deeds, agreements, transfers, assurances, instruments or documents as may reasonably be required by the Company in order to better implement this Plan.

9.12 Governing Law

This Plan shall be governed by and construed in accordance with the law of the Province of Quebec and the federal laws of Canada applicable therein. Any questions as to the interpretation or application of this Plan and all proceedings taken in connection with this Plan and its provisions shall be subject to the exclusive jurisdiction of the Court.

9.13 Successors and Assigns

This Plan shall be binding upon and shall enure to the benefit of the heirs, administrators, executors, legal personal representatives, successors and permitted assigns of the Company, the Directors, the Affected Creditors or any other Persons affected by or benefiting from the provisions of this Plan.

9.14 French Language Version of Plan

In the event of any discrepancy between any of the provisions of the English language version of this Plan and any French language version thereof, the provisions of the English version of this Plan shall, under all circumstances, prevail and govern.

9.15 Choice of Language

The Company and each of the hereafter enumerated members of the Fisher Group acknowledges that it (or he) has required that this Plan and all related documents be prepared, in English. La Compagnie et chacun des membres du "Fisher Group" ci-après énumerés reconnaît avoir exigé que ce Plan et tous les documents connexes soient rédigés en anglais.

(signature page to follow)

MONTREAL, Province of Quebec, this 16th day of October, 2015.

	LAURA'S SHOPPE (P.V.) INC. Per:
•	- Alm
	Kalman Fisher, President
EACH of the following members of the cognizance of the contents of this Plan and hereof.	
	Borr
	KALMAŃ FISHER
	3482731 CANADA INC. Per: //////
	Kalmán Fisher .
	9318-5494 QUEBEC INC. Per:

A N N E X (corrected) (TO PLAN OF COMPROMISE AND ARRANGEMENT OF LAURA'S SHOPPE (P.V.) INC.)

DISCLAIMED LEASES

	LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
1.	Calloway REIT (Halifax)	Bayers Lake Center	Laura
	Ltd.	208 Chain Lake Drive	
	(Smartcentres)	Halifax, Nova Scotia	
		B3S 1C5	
2.	The Boulevard Shopping	The Boulevard Shopping Centre	Laura
	Centre (Montreal) Limited Partnership	4270 Jean Talon Street East	
·		Montreal, Quebec	•
		H1S 1J7	
3.	Calloway Reit (Cambridge)	Cambridge Centre	Laura
	Inc. (Smartcentres)	28 Pinebush Road	
		Cambridge, Ontario	
4.	Riocan Holdings (Quebec) Inc.	Mega Centre Notre-Dame	Laura
		Notre-Dame and Autoroute 13	
		Laval, Quebec	
- ,		H7X 3V5	

		•	
	LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
5.	3945333 Manitoba Limited (Smartcentres)	SmartCentres Winnipeg Southwest	Laura
		1665 Kenaston Blvd.	
		Unit P3	
		Winnipeg, Manitoba	
		R3P 2M4	
6.	Riocan Property Services	Beacon Hill Centre	Laura
	Trust as agent for 2113362 Ontario Limited, as to an undivided one-half interest, and Trinity Properties Alberta Limited as to an undivided one-half interest	11834 Sarcee Trail NW	
		Calgary, Alberta	
		T3R 0A1	
	Beauward Shopping	Galeries Joliette	Laura
	Centres Ltd.	1075 Firestone Blvd.	
		Joliette, Quebec	
		J6E 6X6	
8.	Montez Core Income Fund Limited Partnership -and- Hillcrest Property Holdings Inc.	Hillcrest Mall	Laura
		9350 Yonge Street	
		Richmond Hill, Ontario	
		L4C 5G2	

	LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (If applicable)
9.	Carrefour Richelieu	Carrefour Angrignon	Laura Plus
	Realties Ltd.	7077 Newman Blvd.	
	(Westcliff)	LaSalle, Quebec	
		H8N 1X1	·
10.	Omers Realty	Square One Shopping Centre	Laura Plus
	Management Corporation, by its manager OPGI	100 City Centre Drive	
	Management GP Inc. (as general partner of the OPGI	Mississauga, Ontario	
	Management Limited Partnership)	L5B 2C9	
11.	Park Royal Shopping	Park Royal Shopping Centre	Melanie Lyne
	Centre Holdings Ltd.	2002 Park Royal South	
		West Vancouver, BC	
		V7T 2W4	
12.	Oxford Properties Group	Edmonton City Centre West	Melanie Lyne
	Incand- CPP Investment Board Real Estate Holdings Inc.	10020-102 Avenue	
		Edmonton, Alberta	
		T5J 4B7	
13.	Dartmouth Crossing	Dartmouth Crossing	Melanie Lyne
•	Limited	34 Logiealmond Close	**
		Dartmouth, Nova Scotia	
		B3B 0C8	

	LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
14.	RRL Burloak Incand-	RioCan Centre Burloak	Melanie Lyne
	2121049 Ontario Limited, by its agent RioCan Property	3543 Wyecroft Road	
Services Inc. (Trustee for RioCan Real Estate Investment Trust)	Oakville, Ontario		
15.	Riotrin Properties	Trinity Crossing	Melanie Lyne
	(Orleans) Inc.	4210 Innes Road	
		Orleans, Ontario	
		K4A 5E6	
16.	RioCan Holdings (TJV	Carrefour Champêtre Bromont	Laura
	Bromont) Incand-	105 boul. le Carrefour	
	1633272 Alberta ULC	Bromont, Quebec	
		J2L 3L1	
17.	1451945 Ontario Limited	Mayfield Common	Laura
	-and- Sun Life Assurance	288 Mayfield Common Northwest	
	Company of Canada	Edmonton, Alberta	
		T5P 4B3	
18.	RioCan Meadows	RioCan Meadows	Laura
	Shopping Centre Incand-	3918 - 17 St. NW	
	2118717 Ontario Limited	Edmonton, Alberta	
		T6T 0C2	

	LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
19.	Riotrin Properties Inc.	RioCan Signal Hill Center	Laura
		5715 Signal Hill Center SW	
		Calgary, Alberta	
		T3H 3P8	
20.	713949 Ontario Limited	St. Laurent Shopping Centre	Laura
		1200 St. Laurent Blvd.	
		Ottawa, Ontario	
		K1K 3B8	
21.	Montez Core Income Fund	Hillcrest Mall	Melanie Lyne
	Limited Partnership -and-	9350 Yonge Street	
	Hillcrest Property Holdings Inc.	Richmond Hill, Ontario	
		L4C 5G2	
22.	Les Factoeries Saint-	Les Factoreries Saint-Sauveur	Melanie Lyne
	Sauveur (2003) Inc.	160 chemin du Lac-Millette	•
		Saint-Sauveur, Quebec	
		JOR 1R6	
23.	9191-3723 Quebec Inc.	2955 rue Jules-Brillant	N/A
		Laval, Quebec	
	• • • • • • • • • • • • • • • • • • • •	H7P 6B2	

Page 6

	LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
24.	Desjardins Financial	151 City Centre Drive	N/A
	Security Life Assurance Company	Suites 603/701	
		Mississauga, Ontario	
		L5B 1M7	
25.	3608727 Canada Inc.	2519-2525 boul. Le Corbusier	N/A
		Laval, Quebec	·
		H7S 1Z4	

EXHIBIT P-3

(Re-Amended Plan dated November 18, 2015)

CANADA

PROVINCE OF QUEBEC DISTRICT OF MONTREAL

NO.: 500-11-049256-155

SUPERIOR COURT (Commercial Division)

(Sitting as a court designated pursuant to the Companies' Creditors Arrangement Act, R.S.C. c. C-36)

IN THE MATTER OF THE PLAN OF COMPROMISE AND ARRANGEMENT OF:

MAGASIN LAURA (P.V.) INC. / LAURA'S SHOPPE (P.V.) INC.

Debtor/Petitioner

-and-

BOUTIQUE LAURA CANADA LTÉE / LAURA'S SHOPPE CANADA LTD.

-and-

3482731 CANADA INC.

-and-

9318-5494 QUEBEC INC.

-and-

KALMAN FISHER

Stayed Parties

-and-

KPMG INC.

Monitor

RE-AMENDED PLAN OF COMPROMISE AND ARRANGEMENT
Under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36

TABLE OF CONTENTS

1. INIE	RPRETATION	<u> 1</u>
1.1.	Definitions	1
1.2.	Interpretation	9
1.3.	Date For Any Action	10
1.4.	Statutory References	10
2. COMP	PROMISE AND ARRANGEMENT	<u>10</u>
2.1.	Persons Affected	
2.2.	Classes of Affected Claims	10
2.3.	Disclaimed Landlord Claims	10
2.4.	Unaffected Claims	10
2.5.	Treatment of Affected Claims	11
2.6.	Settlement of Affected Claims	12
2.7.	Extension of Payment Dates	12
2.8.	Prepayment Rights	13
2.9.	No Acknowledgment or Admission	13
3. TREAT	TMENT OF OTHER CLAIMS AND MATTERS	13
3.1.	Treatment of the DIP Claim	13
3.2.	Treatment of Excluded Claims	13
3.3.	Treatment of Administration Claims	13
3.4.	Treatment of Gift Card Claims	14
3.5.	Treatment of Certain Crown Priority Claims	14
3.6.	Treatment of Section 6(5) Claims	14
3.7.	Treatment of Section 19(2) Claims	14
3.8.	Treatment of KERP Claims	14
3.9.	Treatment of Secured Claims	14
3.10.	Treatment of Replacement Financing Claim	14
3.11.	Treatment of Fisher Group Claims	14
4. VALUA	ATION OF CLAIMS, CREDITORS' MEETING AND RELATED MATTERS	15
4.1.	Conversion of Affected Claims into Canadian Currency	15
4.2.	Affected Claims	15
4.3.	Creditors' Meeting	

4.4.	Approval by Affected Creditors	15
4.5.	Amendment of Proofs of Claim	16
4.6.	Late Filing of Claims for Disclaimed Leases	16
4.7.	Claims Bar Date	16
5. PR	OCEDURE FOR RESOLVING DISTRIBUTIONS IN RESPECT OF DISPUTE	
5.1.	No Distributions Pending Allowance	
5.2.	Distribution From Reserve Once Disputed Claims Resolved	16
<u>6. EFF</u>	FECT OF THE PLAN AND RELEASES	17
6.1.	Effect of the Plan	
6.2.	Releases upon Implementation	17
6.3.	Releases after Performance	18
6.4.	Exception to Releases	18
6.5.	Set-Off Rights	19
6.6.	Injunction Related to Releases	19
6.7.	Waiver of Defaults	19
6.8.	Stay of Proceedings	20
7. PRC	OVISIONS GOVERNING DISTRIBUTIONS	20
7.1.	Partial Distributions for Claims Allowed	
7.2.	Currency to be used for the Distribution	20
7.3.	Assignment of Claims	20
7.4.	Interest on Affected Claims	20
7.5.	Distributions by Monitor	21
7.6.	Delivery of Distributions	21
7.7.	No Double Recovery	21
7.8.	Withholding Requirements	22
<u>8. IMPI</u>	LEMENTATION OF THE PLAN	
8.1.	Conditions Precedent to Implementation of Plan	
8.2.	Waiver of Conditions	24
8.3.	Certificate of Implementation	25
8.4.	Effect of Plan	
8,5,	Nullity of Plan	25
9. MISC	CELLANEOUS	
9.1.	Confirmation of Plan	

9.2.	Paramountcy	26
9.3.	Modification of Plan	
9.4.	Deeming Provisions	27
9.5.	Sections 38 and 95 to 101 BIA	27
9.6.	Responsibilities of the Monitor	27
9.7.	Liability Limitations	27
9.8.	Notices	28
9.9.	Severability of Plan Provisions	30
9.10.	Revocation, Withdrawal or Non-Consummation	30
9.11.	Further Assurance	31
9.12.	Governing Law	
9.13.	Successors and Assigns	31
9.14.	French Language Version of Plan	31
9.15.	Choice of Language	31

1. INTERPRETATION

1.1. Definitions

In this Plan, unless otherwise stated or the context otherwise requires:

"Administration Charge" has the meaning ascribed to such term in the Initial Order;

"Administration Claim" means a claim or any other indebtedness or obligation secured by the Administration Charge:

"Affected Claim" means any Claim other than an Unaffected Claim. For greater certainty, Affected Claims include Disclaimed Landlord Claims, any Claim of Her Majesty the Queen in right of Canada or any province (other than Crown Priority Claims), Directors Claims and Restructuring Claims;

"Affected Creditor" means a Creditor holding an Affected Claim, but only to the extent of its Affected Claim;

"Affected Creditors Class" has the meaning set forth in Section 2.2 hereof;

"Aggregate Distribution" has the meaning set forth in Section 2.5 hereof;

"BIA" means the Bankruptcy and Insolvency Act, R.S.C., 1985, c. B-3, as amended;

"Business Day" means a day, other than a Saturday, a Sunday, or a non-juridical day (as defined in article 6 of the Quebec Code of Civil Procedure, R.S.Q., c. C-25, as amended):

"CCAA" means the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended:

"CCAA Charge(s)" means the Administration Charge, the DIP Lender Charge and the KERP Charge;

"CCAA Proceedings" means proceedings in respect of the Company before the Court commenced, taken up and continued under the CCAA;

"Certificate of Implementation" has the meaning set forth in Section 8.3 hereof;

"Certificate of Performance" has the meaning set forth in Section 6.1 hereof;

"Certificate of Non-Implementation" has the meaning set forth in Section 8.5 hereof;

"Chair" means a representative of the Monitor, or any other individual designated by the Monitor to preside as chairperson at the Creditors' Meeting;

"Claim" means any right of any Person against the Company in connection with any indebtedness, liability or obligation of any kind of the Company owed to such Person and any interest, or penalties accrued thereon or costs payable in respect thereof, whether liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, unsecured, present, future, known or unknown, by guarantee, surety or otherwise, and whether or not such right is executory or anticipatory in nature, including the right or ability of any Person to advance a claim for contribution or indemnity or otherwise with respect to any matter, action, cause or chose in action, whether existing at present or commenced in the future, which indebtedness, liability or obligation is based in whole or in part on facts existing, or transactions which occurred, prior to the Determination Date, or which would have been claims provable in bankruptcy had the Company become bankrupt on the Determination Date, and, without limitation, including any Restructuring Claim, but excluding any Unaffected Claim and "Claims" means all of them;

"Claims Bar Date" has the meaning as set forth in the Claims Procedure Order;

"Claims Procedure Order" means the "Claims Procedure Order" issued by the Court on September 11, 2015, establishing, among other things, procedures for proving Claims, as amended or supplemented from time to time by further Order(s) of the Court;

"Company", "Debtor" or "Petitioner" means Magasin Laura (P.V.) Inc./Laura's Shoppe (P.V.) Inc.;

"Court" means the Commercial Division of the Quebec Superior Court for the District of Montreal;

"Creditor(s)" means any Person(s) having a Claim and may, where the context requires, include the assignee of a Claim or a trustee, interim receiver, receiver, receiver and manager, or other Person acting on behalf of such Person(s);

"Creditors' Meeting" means the meeting of Affected Creditors to be convened for the purposes of voting on the Plan, or any adjournment of such meeting;

"Crown Priority Claims" means any Claims of Her Majesty the Queen in right of Canada or in right of any province as described in Section 6(3) or Section 38(2) of the CCAA. For greater certainty, any Claim of Her Majesty the Queen in right of Canada or in right of any Province other than Crown Priority Claims shall be an Affected Claim hereunder:

"Determination Date" means July 31, 2015, being the date on which the Company commenced proceedings under the *BIA* by filing a notice of intention to make a proposal thereunder, which proceedings were subsequently taken up and continued by the *CCAA* Proceedings under the *CCAA*;

"DIP Claim" means any and all obligations of the Company to the DIP Lender pursuant to the DIP Facility Documents or the Order for New Financing as of or after the Plan Implementation Date;

"DIP Facility Documents" means the "committed term sheet" dated October 22, 2015 by and among, inter alia, the Company and the DIP Lender (as amended, modified, restated and/or supplemented from time to time) together with any related collateral, loan or security documents executed in connection therewith or which relate thereto;

"DIP Lender" means collectively, Third Eye Capital Corporation, acting as agent and hypothecary representative for and on behalf of certain affiliates and funds it manages as well as all of the "New Financing Parties" as defined in the Order for New Financing;

"DIP Lender Charge" means the "New Lender Charge" as set forth in the Order for New Financing;

"DIP Loan" means the loan(s) made and/or to be made by the DIP Lender as envisaged by the Order for New Financing;

"Directors" means all of the Company's past and present directors as well as any Persons who were or are deemed to be directors of the Company pursuant to any applicable Laws;

"Directors' Claims" means all Claims, of any nature or source whatsoever, of any of the Creditors against any or all of the Directors which arose before the Determination Date and which relate to obligations of the Company where such Directors are by law liable in their capacity as directors for payment of such obligations. Director's Liability shall, however, exclude all Claims which (i) relate to contractual rights of one or more Creditors arising from contracts with one or more Directors, including any guarantees given by any Directors relating to the Company's obligations, (ii) seek injunctive relieve against any of the Directors in relation to the Company, or (iii) are based on allegations of misrepresentation made by any Directors to Creditors or of wrongful or oppressive conduct by Directors;

"Disallowed Claim" means any Claim, or that portion thereof which has been revised or disallowed by the Monitor pursuant to the Claims Procedure Order or any other Order in respect of which an appeal by the Creditor has been dismissed or all appeal periods, if any, have expired;

"Disclaimed Leases" means each and every one of the Leases which (i) were disclaimed or resiliated by the Company pursuant to Section 65.2 of the BIA (as recognized by the Initial Order) or Section 32 of the CCAA, and (ii) may be disclaimed or resiliated by the Company at any time between the date of this Plan and the Sanction Date, and "Disclaimed Lease" means each of them:

"Disclaimed Landlord" means the landlord under any Disclaimed Lease and "Disclaimed Landlords" means all of them:

"Disclaimed Landlord Claims" means all Proven Claims of the Disclaimed Landlords for losses incurred by Disclaimed Landlords in relation to Disclaimed Leases, as provided for in Section 2.3 hereof and "Disclaimed Landlord Claim" means any of them;

"Disputed Claim" means a Claim or that portion thereof that is the object of a Notice of Revision or Disallowance and, in either case, has not become a Proven Claim or a Disallowed Claim;

"Excluded Claim" means any right of any person against the Company in connection with any indebtedness, liability or obligation of any kind which arose with respect to transactions which occurred after the Determination Date and any interest thereon, including the Administration Claim, the DIP Claim, a KERP Claim, the Replacement Financing Claim and any obligation of the Company towards creditors who have supplied or shall supply services, utilities, goods or materials or who have or shall have advanced funds to the Company after the Determination Date, but only to the extent of their claims in respect of the supply of such services, utilities, goods, materials or funds after the Determination Date and to the extent that such claims are not otherwise affected by the Plan. For greater certainty, a Restructuring Claim is not an Excluded Claim;

"Excluded Creditor" means a Person having a claim in respect of an Excluded Claim but only in respect of such Excluded Claim and to the extent that the Plan does not otherwise affect such claim;

"First Released Parties" has the meaning as set forth in Section 6.2 hereof;

"Fisher Group" means Kalman Fisher, 3482731 Canada Inc., 9318-5494 Quebec Inc. (formerly 7735235 Canada Inc.) and any Person who/which is, or was at any time within the 12 consecutive months immediately preceding the Determination Date, a Related Person to Kalman Fisher:

"Fisher Group Claims" means any Claims for payment of principal on any loans or advances made to the Company (or its predecessor) at any time prior to the Determination Date by any member of the Fisher Group, which includes any Secured

Claim by any member of the Fisher Group, but excludes any Claims or future claims by any member of the Fisher Group for (i) salary (or other remuneration), (ii) reimbursement of business expenses, or (iii) payment of principal on any loans or advances in lieu of salary (or other remuneration) or in lieu of interest;

"Gift Card Claim" has the meaning set forth in Section 2.4(a)(v) hereof;

"Governmental Authority" means any (i) multinational, national, provincial, state, regional, municipal, local or other government, governmental or public department, ministry, central bank, court, tribunal, arbitral body, commission, board, official, minister, bureau or agency, domestic or foreign, (ii) subdivision, agent, commission, board or authority of any of the foregoing; or (iii) quasi-governmental or private body, including any tribunal, commission, regulatory agency or self-regulatory organization, exercising any regulatory, expropriation or taxing authority under, or for the account of, any of the foregoing;

"Initial Order" means the "Initial Order" issued by the Court on August 12, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"KERP Charge" has the meaning ascribed to such term in the Initial Order;

"KERP Claim" means a claim or any other indebtedness or obligation secured by the KERP Charge and "KERP Claims" means all of them;

"Laws" means all laws, statutes, codes, ordinances, decrees, rules, regulations, by-laws, judicial or arbitral or administrative or ministerial or departmental or regulatory judgments, injunctions, orders or decisions of any Governmental Authority, statutory body or self-regulatory authority, including general principles of law having the force of law and the term "applicable" with respect to such Laws and, in the context that refers to any Person, means such Laws as are applicable to such Person or its business, undertaking, property or securities and emanate from a Governmental Authority or self-regulatory authority having jurisdiction over the Person or its business, undertaking, property or securities;

"Lease" means any lease and/or other agreement for the occupancy of real and immovable property as well as all amendments thereto and renewals thereof and "Leases" means all of them:

"Meeting Date" means the date fixed for the Creditors' Meeting in accordance with the Meeting Procedure Order, or any subsequent Order, or any subsequent date following any adjournment of that meeting, as the case may be;

"Meeting Procedure Order" means the "Meeting Procedure Order" issued by the Court on October 23, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"Monitor" means KPMG Inc., in its capacity as Monitor, as appointed by the Court pursuant to the Initial Order;

"Notice of Revision or Disallowance" has the meaning as set forth in the Claims Procedure Order:

"Order" means any order of the Court in the CCAA Proceedings:

"Order for New Financing" means the "Order for New Financing" issued by the Court on October 30, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"Original Currency" has the meaning set forth in Section 7.7 hereof;

"Payment Date" and "Payment Dates" have the meaning set forth in Section 2.5 hereof;

"Person" means any individual, corporation, limited or unlimited liability company, general or limited partnership, association, trust, trustee, executor, administrator, legal personal representative, estate, unincorporated organization, joint venture, governmental body or agency, or any other entity;

"Plan" means the present "Re-amended Plan of Compromise and Arrangement" of the Company pursuant to the provisions of the CCAA, as may be amended, varied or supplemented by the Company from time to time in accordance with its terms:

"Plan Implementation Conditions" has the meaning set forth in Section 8.1 hereof;

"Plan Implementation Date" means the date, on or prior to the Plan Implementation Deadline, upon which all of the Plan Implementation Conditions have occurred or have been satisfied or waived:

"Plan Implementation Deadline" has the meaning set forth in Section 8.1 hereof;

"Proof of Claim" has the meaning set forth in the Claims Procedure Order;

"Proven Claim" means, in respect of a Creditor, the amount of the Claim of such Creditor as finally determined for distribution purposes in accordance with the provisions of this Plan, the CCAA and the Claims Procedure Order and "Proven Claims" means all of them;

"Related Persons" has the meaning set forth in Section 4 of the BIA;

"Replacement Financing" means the new senior credit facilities to be provided to the Company by the DIP Lender, including a term loan facility and a revolving asset-based loan facility on terms and conditions acceptable to the DIP Lender and the Company, which shall completely repay and/or replace the DIP Loan;

"Replacement Financing Claim" means any and all indebtedness and obligations of the Company and the DIP Lender under the Replacement Financing;

"Required Majorities" means the affirmative vote of a majority in number of the Affected Creditors voting in the Affected Creditors Class, having Voting Claims and voting on the Plan (in person or by proxy) at the Creditors' Meeting and representing not less than 66-2/3% in value of the Voting Claims of the Affected Creditors voting (in person or by proxy) at the Creditors' Meeting;

"Reserve(s)" means the reserve(s) to be established and maintained under this Plan by the Monitor by holding, on account of Disputed Claims, an amount equal to the aggregate amount that the holders of Disputed Claims would be entitled to receive if all such Disputed Claims had been Proven Claims as of any Payment Date;

"Resolution" means, collectively, one or any of, the resolutions providing for the approval of the Plan by the Affected Creditors;

"Restructuring Claim" means any right or claim of any Person against the Company in connection with any indebtedness or obligation of any kind owed to such Person arising out of the Company's disclaimer, resiliation or renegotiation (at any time after the Determination Date) of any Leases (excluding the Disclaimed Landlord Claims, which shall be dealt with pursuant to Section 2.3 hereof), agreements or other contracts or the Company's restructuring under the CCAA, and includes any right or claim resulting, directly or indirectly, from the consequences and effects of the Plan's acceptance by the Affected Creditors, the Plan's sanction by the Sanction Order, the Plan's implementation and any debt forgiveness resulting from any of the foregoing; provided, however, that a Restructuring Claim shall not include an Excluded Claim. For greater certainty, a Restructuring Claim is an Affected Claim;

"Sanction Date" means the date on which the Sanction Order is issued;

"Sanction Order" means the Order to be made under the CCAA sanctioning the Plan, as such Order may be affirmed, amended or modified by the Court at any time prior to the Plan Implementation Date or, if appealed, then, unless such appeal is withdrawn, abandoned or denied, as affirmed or amended on appeal, in form and content which is satisfactory to the Petitioner acting reasonably;

"Second Released Party" and "Second Released Parties" have the meanings as set forth in Section 6.3 hereof;

"Section 6(5) Claim" means any claim of an employee or former employee described in Section 6(5) of the CCAA but only to the extent of such amounts as required to be paid under the CCAA;

"Section 19(2) Claims" means any claim described in Section 19(2) of the CCAA;

"Secured Claim" means the Claim of a Secured Creditor, to the extent of the value of such Secured Creditor's security;

"Secured Creditor" has the meaning set forth in the CCAA, but only to the extent that such Creditor's mortgage, hypothec, pledge, charge, lien, privilege, security interest or other rights over the property of the Company was valid, opposable, perfected, and could be set up against third parties, including a trustee to the Company's bankruptcy, on both the Determination Date and the Claims Bar Date, failing which that Secured Creditor will be deemed to be an "unsecured creditor", as defined in the CCAA:

"Stayed Parties" means Boutique Laura Canada Ltée/Laura's Shoppe Canada Ltd., 3482731 Canada Inc., 9318-5494 Quebec Inc. (formerly 7735235 Canada Inc.) and Kalman Fisher;

"Taxes" means all federal, state, provincial, territorial, county, municipal, local or foreign taxes, duties, imposts, levies, assessments, tariffs and other charges imposed, assessed or collected by a Governmental Authority, including (i) any gross income, net income, gross receipts, business, royalty, capital, capital gains, goods and services, value added, severance, stamp, franchise, occupation, premium, capital stock, sales and use, real property, land transfer, personal property, ad valorem, transfer, licence, profits, windfall profits, environmental, payroll, employment, employer health, pension plan, anti-dumping, countervail, excise, severance, stamp, occupation, or premium tax, (ii) all withholdings on amounts paid to or by the relevant Person, (iii) all employment insurance premiums, Canada, Quebec and any other pension plan contributions or premiums, (iv) any fine, penalty, interest, or addition to tax, (v) any tax imposed, assessed, or collected or payable pursuant to any tax-sharing agreement or any other contract relating to the sharing or payment of any such tax, levy, assessment, tariff, duty, deficiency, or fee, and (vi) any liability for any of the foregoing as a transferee, successor, guarantor, or by contract or by operation of law;

"Unaffected Claim" means any right of any Person in connection with any indebtedness, liability or obligation of any kind of the type described in Section 2.4(a) hereof. For greater certainty, the Unaffected Claims include the Excluded Claims;

"Unaffected Creditor" means a Person having a Claim in respect of an Unaffected Claim, but only in respect of such Unaffected Claim, and for greater certainty, includes an Excluded Creditor;

"Voting Claim" means, in respect of an Affected Creditor, the amount of such Affected Creditor's claim which has been accepted for voting purposes in accordance with the provisions of this Plan, the Claims Procedure Order and the CCAA and "Voting Claims" means all of them;

1.2. Interpretation

For purposes of this Plan:

- (a) any reference in this Plan to a contract, instrument, release, indenture, agreement or other document being in a particular form or on particular terms and conditions means that such document shall be substantially in such form or substantially on such terms and conditions;
- (b) any reference in this Plan to an existing document or exhibit filed or to be filed means such document or exhibit as it may have been or may be amended, modified or supplemented;
- (c) all references to currency and to "\$", "C\$" or "Dollars" are to Canadian dollars except as otherwise indicated;
- (d) all references in this Plan to Sections are references to Sections of this Plan;
- (e) unless otherwise specified, the words "hereof", "herein" and "hereto" refer to this Plan in its entirety rather than to any particular portion of this Plan;
- (f) the division of this Plan into Sections and paragraphs and the insertion of captions and headings to Sections and paragraphs are for convenience of reference only and are not intended to affect the interpretation of, or to be part of this Plan;
- (g) where the context requires, a word or words importing the singular shall include the plural and vice versa and a word or words importing one gender shall include all genders;
- (h) the words "includes" and "including" are not limiting; and
- (i) the word "or" is not exclusive.

1.3. Date For Any Action

In the event that any date (including any Payment Date) on which any action (including any payment) is required to be taken under this Plan by any of the parties is not a Business Day, that action (including any payment) shall be required to be taken on the next succeeding day which is a Business Day.

1.4. Statutory References

Any reference in this Plan to a statute includes all regulations made thereunder and all amendments to such statute or regulations in force, from time to time, or any statute or regulations that supplement or supersede such statute or regulations.

2. <u>COMPROMISE AND ARRANGEMENT</u>

2.1. Persons Affected

The purpose of this Plan is to provide for the full and final settlement of all Affected Claims. Except as specifically provided for in this Plan, this Plan will become effective on the Plan Implementation Date in accordance with its terms, and all Affected Claims against the Company will be fully and finally, settled, compromised and released in this Plan upon the Aggregate Distribution being fully paid to the Monitor, it being understood that any Affected Claim that is paid in full pursuant to this Plan before such date shall be settled, compromised and released on the date of such payment. This Plan shall be binding on and enure to the benefit of the Company, the Affected Creditors, the First Released Parties, the Second Released Parties, any trustee, agent or other Person acting on behalf of any Affected Creditor and such other Persons who have received the benefit of, or are bound by any waivers, releases or indemnities hereunder.

2.2. Classes of Affected Claims

There shall only be one class of Affected Creditors for the purpose of voting on, and receiving distributions pursuant to this Plan, being the "Affected Creditors Class".

2.3. Disclaimed Landlord Claims

Disclaimed Landlord Claims shall be Affected Claims under this Plan and shall be determined in accordance with Section 32(7) CCAA.

2.4. Unaffected Claims

(a) This Plan does not affect the following claims (each, an "Unaffected Claim" and, collectively, the "Unaffected Claims"), the holders of which will not be entitled to

vote at the Creditors' Meeting or receive any distributions under this Plan, namely:

- (i) the Excluded Claims;
- (ii) any Administration Claim;
- (iii) any DIP Claim;
- (iv) any KERP Claims;
- (v) any claim with respect to gift-cards, gift certificates, lay-away deposits and other customer certificates (collectively, "Gift Card Claims");
- (vi) Crown Priority Claims;
- (vii) Section 6(5) Claims;
- (viii) Section 19(2) Claims;
- (ix) Secured Claims:
- (x) the Replacement Financing Claim; and
- (xi) Fisher Group Claims,

all of which will be dealt with in Section 3 hereof;

(b) Nothing in this Plan shall affect the Company's rights and defences, both legal and equitable, with respect to any Unaffected Claim including any rights arising under or pursuant to the Claims Procedure Order or this Plan or any rights with respect to legal and equitable defences or entitlements to set-offs or recoupments against such Unaffected Claims.

2.5. Treatment of Affected Claims

By no later than each of the payment dates hereafter set forth (collectively the "Payment Dates" or individually a "Payment Date"), the aggregate sum of \$4 Million (the "Aggregate Distribution"), without any interest whatsoever thereon, shall be paid by the Company to the Monitor and shall thereafter be distributed by the Monitor to the Affected Creditors, according to the amount of their respective Proven Claims as hereafter set forth, namely:

- (a) by no later than April 30, 2016, \$1 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor as follows:
 - (i) an amount equal to the lesser of (A) the Proven Claim of each Affected Creditor, or (B) \$1,250, shall be distributed by the Monitor to the Affected Creditors, according to the amount of their Proven Claims; and
 - (ii) an amount equal to the difference between (A) \$1 Million, and (B) the aggregate amount to be distributed by the Monitor pursuant to Section 2.5(a)(i) above, shall be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the amounts set forth in Section 2.5(a)(i) above;
- (b) by no later than July 31, 2016, \$1 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distribution under Section 2.5(a) hereof;
- (c) by no later than December 15, 2016, \$1.8 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distributions under Sections 2.5(a) and 2.5(b) hereof; and
- (d) by no later than February 15, 2017, \$200,000 shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distributions under Sections 2.5(a), 2.5(b) and 2.5(c) hereof.

2.6. Settlement of Affected Claims

The payment of the entire Aggregate Distribution to the Monitor as set forth in Section 2 hereof shall constitute the full and final payment, reduction, settlement and transaction (as envisaged by the relevant provisions of the Civil Code of Quebec) of all of the Affected Claims of all of the Affected Creditors and engage the releases and other provisions set forth in this Plan.

2.7. Extension of Payment Dates

Any or all of the Payment Dates may be extended by the Company with the consent of the Monitor and the DIP Lender (if any portion of the DIP Loan or the Replacement Financing then remains outstanding), by additional periods not exceeding, in each case, 60 days. On application by the Company or the Monitor (in either case, with the consent of the DIP Lender, if any portion of the DIP Loan or the Replacement Financing remains outstanding) made prior to a Payment Date or, if applicable, prior to the expiry of any extension thereof as described above, the Court may issue Orders extending (or further extending) any or all of the Payment Dates for any additional period(s) deemed appropriate by the Court in the then existing circumstances.

2.8. Prepayment Rights

The Company (with the consent of the DIP Lender, if any portion of the DIP Loan or the Replacement Financing then remains outstanding) shall, at all times prior to the Final Distribution Date, be entitled to pre-pay the whole or any amount of the Aggregate Distribution prior to any Payment Date, whereupon the amount(s) of any such pre-payment(s) shall be distributed by the Monitor to the Affected Creditors in accordance with the provisions of **Section 2.5** hereof, *mutatis mutandis*.

2.9. No Acknowledgment or Admission

Nothing contained in this Plan shall constitute, under any circumstances whatsoever, an acknowledgment or admission by the Company as to the existence or validity of any Claims or Restructuring Claims.

3. TREATMENT OF OTHER CLAIMS AND MATTERS

3.1. <u>Treatment of the DIP Claim</u>

On or before the earlier of (i) the Plan Implementation Date, or (ii) the date on which the DIP Claim becomes payable pursuant to the provisions of the DIP Facility Documents, the DIP Lender shall receive full payment of any remaining DIP Claim.

3.2. Treatment of Excluded Claims

Subject to Section 3.1 hereof, the Excluded Claims (other than the Administration Claim, the DIP Claim and any KERP Claim) will remain in full force and effect in accordance with their terms after the Plan Implementation Date, and will be paid in full by the Company in the normal course of its business as and when they become due.

3.3. <u>Treatment of Administration Claims</u>

All Administration Claims, if any, will be paid in full by the Company as and when they become due, and any remaining balance will be settled as soon as practicable after the Plan Implementation Date.

3.4. Treatment of Gift Card Claims

Gift Card Claims will be honored in accordance with the terms of the relevant gift card, gift certificate, lay-away deposit or other customer certificates upon presentation of such gift card, gift certificate, lay-away deposit or other customer certificates by the holder at any of the Petitioner's retail stores.

3.5. Treatment of Certain Crown Priority Claims

All Crown Priority Claims described in Section 6(3) of the CCAA, if any, will be paid in full by the Petitioner within 6 months immediately following the Sanction Date.

3.6. Treatment of Section 6(5) Claims

All Section 6(5) Claims, if any, will be paid in such amounts as required under the CCAA immediately after the Sanction Date.

3.7. Treatment of Section 19(2) Claims

All Section 19(2) Claims, if any, will be paid by the Company as and when they become due, unaffected by this Plan.

3.8. Treatment of KERP Claims

Each KERP Claim will be paid by the Company as and when it becomes due.

3.9. Treatment of Secured Claims

Secured Claims (other than the Administration Claim, the DIP Claim, the Replacement Financing Claim and KERP Claims) will be dealt with as provided for under agreements entered into or as may be hereafter entered into between the Company and the relevant Secured Creditor, or as may be ordered by the Court. For greater certainty, Secured Claims will be unaffected by this Plan.

3.10. <u>Treatment of Replacement Financing Claim</u>

The Replacement Financing Claim shall be paid by the Company to the DIP Lender in accordance with the provisions, terms and conditions of the Replacement Financing.

3.11. Treatment of Fisher Group Claims

Conditional upon the enactment of a Resolution approving the Plan, issuance of the Sanction Order and implementation of the Plan as set forth in **Section 8** hereof, each member of the Fisher Group:

- (a) subordinates and postpones all of the Fisher Group Claims to and in favour of full payment of the Aggregate Distribution to the Affected Creditors, such that no member of the Fisher Group shall be entitled to receive any payment from the Company of any principal on any of the Fisher Group Claims unless and until the full Aggregate Distribution shall have been paid to the Monitor as set forth in the Plan; and
- (b) waives and renounces to any right to be an Affected Creditor.

4. VALUATION OF CLAIMS, CREDITORS' MEETING AND RELATED MATTERS

4.1. Conversion of Affected Claims into Canadian Currency

For the purposes of determining the value of Affected Claims denominated in currencies other than Canadian dollars for voting purposes, such Affected Claims shall be converted by the Monitor to Canadian dollars at the Bank of Canada noon spot rate of exchange for exchanging currency to Canadian dollars on the Determination Date (which in the case of US Dollars was C\$1.3080 for US\$1.00).

4.2. Affected Claims

Affected Creditors shall be entitled to prove their respective Affected Claims, vote their Voting Claims in respect of the Plan, and, if their Affected Claims become Proven Claims, receive the distributions, all as set forth in the Claims Procedure Order and this Plan.

All amounts recognized as Voting Claims or Proven Claims shall be net of any amount that the Company is entitled to offset, recoup, compensate or otherwise apply in reduction of such amounts.

4.3. Creditors' Meeting

The Creditors' Meeting shall be held in accordance with this Plan, the Meeting Procedure Order, and any further Order which may be made from time to time for the purposes of considering and voting on the Resolution or other matters to be considered at the Creditors' Meeting.

4.4. Approval by Affected Creditors

The Company will seek approval of the Plan by the affirmative vote of the Required Majorities. The Resolution to be voted on at the Creditors' Meeting will be decided by the Required Majorities on a vote as set forth in the Meeting Procedure Order. The result of any vote will be binding on all Affected Creditors, whether or not any such Affected Creditor is present and voting (in person or by proxy) at the Creditors' Meeting.

4.5. Amendment of Proofs of Claim

Any Affected Creditor who filed its Proof of Claim prior to the Claims Bar Date or who has been permitted to file a late claim pursuant to an Order, shall be entitled to amend its Proof of Claim in order for its Proof of Claim to conform to the provisions of this Plan.

4.6. Late Filing of Claims for Disclaimed Leases

Notwithstanding the Claims Bar Date and its effect or anything contained in this Plan or the Claims Procedure Order, any Disclaimed Landlord in respect of a Disclaimed Lease which may be disclaimed or resiliated between the date of this Plan and the Sanction Date shall be entitled, during (but not after) a period of 30 days following the date of the disclaimer or resiliation of such Lease, to file a Disclaimed Landlord Claim with the Monitor in conformity with Section 2.3 hereof. If such Disclaimed Landlord fails to file such Disclaimed Landlord Claim within such 30 day period, such Disclaimed Landlord shall have no right to receive any distribution of the Aggregate Distribution and the Company shall be released from such Disclaimed Landlord Claim and Sections 6.2 and 6.3 of the Plan shall apply to such Disclaimed Landlord Claim.

4.7. Claims Bar Date

Subject to the provisions of **Sections 4.5** and **4.6** hereof, if an Affected Creditor holding an Affected Claim has failed to file its Proof of Claim prior to the Claims Bar Date and has not been permitted to file a late claim pursuant to an Order, that Affected Creditor shall be barred from voting at the Creditors' Meeting, that Affected Creditor shall have no right to receive any distributions of the Aggregate Distribution hereunder, the Company shall be released from the Affected Claims of such Creditor and **Sections 6.2** and **6.3** of this Plan shall apply to all such Affected Claims.

5. PROCEDURE FOR RESOLVING DISTRIBUTIONS IN RESPECT OF DISPUTED CLAIMS

5.1. No Distributions Pending Allowance

Notwithstanding any other provision of this Plan, no distributions of the Aggregate Distribution hereunder shall be made by the Monitor with respect to a Disputed Claim unless and until it has become a Proven Claim. Disputed Claims shall be dealt with in accordance with the Claims Procedure Order and this Plan.

5.2. <u>Distribution From Reserve Once Disputed Claims Resolved</u>

The Monitor shall make allocations from the Reserve to holders of Disputed Claims following any of the Payment Dates in accordance with this Plan. To the extent that Disputed Claims become Proven Claims after any relevant Payment Date, the Monitor

shall, from time to time at its sole discretion, distribute from the Reserve to the holders of such Proven Claims, the amount which they would have been entitled to receive in respect of such Proven Claims had such Claims been Proven Claims on any relevant Payment Date. To the extent that any Disputed Claim or a portion thereof has become a Disallowed Claim, then the Monitor shall, at any time which it deems appropriate in the circumstances, distribute, to the holders of Proven Claims, their *pro rata* share from the Reserve of such additional amount kept in the Reserve on account of any such Disallowed Claim.

6. <u>EFFECT OF THE PLAN AND RELEASES</u>

6.1. Effect of the Plan

Upon the Aggregate Distribution being fully remitted to the Monitor, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Performance"), whereupon all Affected Claims shall be deemed to be fully and finally settled, compromised, released and transacted (as envisaged by the relevant provisions of the Civil Code of Quebec), as of the date of issuance of the Certificate of Performance, subject only to an Affected Creditor's right to recover the distributions under this Plan, except for Claims as may have been fully paid prior to such date, in which case those Affected Claims will be deemed to be settled as of such payment.

6.2. Releases upon Implementation

Upon the Implementation of this Plan on the Plan Implementation Date, each of:

- (a) the Monitor and its legal counsel in the CCAA proceedings;
- (b) the Company's legal counsel, financial advisors, consultants and agents of the Company (and their respective directors, officers and employees); and
- (c) the DIP Lender and its legal counsel,

(collectively the "First Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Person may be entitled to assert whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with the business and affairs of the Company, this Plan and the CCAA Proceedings, and all

claims arising out of such actions or omissions shall be forever waived and released, all to the full extent permitted by law.

6.3. Releases after Performance

Upon the Aggregate Distribution being fully remitted to the Monitor and the Monitor's issuance and filing with the Court of the Certificate of Performance, each of:

- (a) the Company; and
- (b) the Directors and all present and future officers and employees of the Company,

(each a "Second Released Party" and collectively the "Second Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability. obligation, demand or cause of action of whatever nature which any Person (including any Secured Creditor in respect of each Second Released Party, except the Company and solely in respect to its Secured Claim), may be entitled to assert (including any and all Claims in respect of statutory liabilities and any Directors' Claims of all Directors and all present and former officers and employees of the Company and any alleged fiduciary or other duty), whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with Claims, the business and affairs of the Company, this Plan and the CCAA Proceedings, and all claims arising out of such actions or omissions shall be forever waived and released (other than the right to enforce the Company's obligations under the Plan or any related document), all to the full extent permitted by law.

6.4. Exception to Releases

Nothing set forth in Sections 6.2 and 6.3 hereof shall:

- (a) release or discharge a Second Released Party from an Unaffected Claim;
- (b) release or discharge the Company from or in respect of its obligations under this Plan:
- (c) affect the right of any Person:
 - (i) to recover an indemnity from any insurance coverage under which that Person is an insured; or

- (ii) to obtain recovery on a claim or liability against a Second Released Party from any insurance coverage pursuant to which that Second Released Party is an insured, but, for certainty, any claim or liability to which an insurer is or would otherwise be subrogated as against the Company is released hereunder and the recovery to which such Person shall be entitled under such insurance coverage shall be limited to the proceeds of insurance actually paid by the insurer with respect to such claim or liability;
- (d) release or discharge the Directors with respect to matters set out in Section 5.1(2) of the CCAA; or
- (e) release or discharge any Person (whether or not a Director, officer or employee of the Company) other than the Company from any obligations arising from any guarantee furnished by such Person to and in favour of any Affected Creditor.

6.5. Set-Off Rights

Notwithstanding the releases set forth under the Plan, in general, and **Sections 6.1**, **6.2** and **6.3** hereof, in particular, all Claims shall remain subject to any rights of set-off that otherwise would be available to the Person against whom a Claim is asserted.

6.6. <u>Injunction Related to Releases</u>

The Sanction Order will enjoin the prosecution by or on behalf of any Person, whether directly, derivatively or otherwise, of any claim, obligation, suit, judgment, damage, demand, debt, right, cause of action, liability or interest released, discharged or terminated pursuant to this Plan.

6.7. Waiver of Defaults

From and after the Plan Implementation Date, all Persons shall be deemed to have waived any and all defaults or occurrences of the Company then existing or previously committed by the Company, caused by the Company or arising, directly or indirectly, or non-compliance with any covenant, positive or negative pledge, warranty, representation, term, provision, condition or obligation, express or implied, in any contract, credit document, agreement for sale, lease or other agreement, written or oral, and any and all amendments or supplements thereto, existing between such Person and the Company arising from the Company's insolvency, the Company's filling a notice of intention to make a proposal under the BIA, the filing by the Company under the CCAA or the transactions contemplated by this Plan or otherwise, and any and all notices of default and demands for payment under any instrument, including any guarantee arising from such default, shall be deemed to have been rescinded.

6.8. Stay of Proceedings

The stay of proceedings set forth and ordered in the Initial Order (except to the extent modified by any other Order) in respect of the Company, the Directors and all other Stayed Parties shall be and remain in full force and effect up until and including the Plan Implementation Date and the Affected Creditors hereby consent to the Court's issuance of an Order to such effect.

7. PROVISIONS GOVERNING DISTRIBUTIONS

7.1. Partial Distributions for Claims Allowed

Except as otherwise provided herein or as ordered by the Court, distributions of the Aggregate Distribution shall be made by the Monitor at the time and in the manner deemed reasonable by the Monitor, and partial distributions may be made prior to receipt of the full amount of the Aggregate Distribution. In such a case, all such partial payments shall represent the *pro rata* amount of the distribution to which the holders of a Proven Claim would otherwise be entitled to receive.

7.2. Currency to be used for the Distribution

For the purposes of determining any Claims denominated in currencies other than Canadian Dollars for distribution purposes, such Claims shall be converted by the Monitor to Canadian dollars at the Bank of Canada noon spot rate of exchange for exchanging currency to Canadian dollars on the Determination Date (which in the case of US Dollars was C\$1.3080 for US\$1.00).

7.3. Assignment of Claims

For purposes of determining entitlement to receive a distribution pursuant to this Plan, the Company and the Monitor and each of their respective agents, successors and assigns shall have no obligation to recognize any transfer of Claims except as provided for under the Claims Procedure Order or the Meeting Procedure Order.

7.4. Interest on Affected Claims

Except as specifically provided in the Plan or the Sanction Order, interest shall not be treated as accruing on account of any Affected Claims for purposes of determining the allowance and distribution of such Affected Claim. To the extent that any Proven Claim to which a distribution under this Plan relates is comprised of indebtedness and accrued but unpaid interest thereon, such distribution shall, to the extent permitted by applicable Laws, be allocated for tax purposes to the principal amount of such Proven Claim (including any secured and unsecured portion(s) of the principal amount of such Proven Claim) first and then, to the extent that the consideration exceeds the principal amount of

the Proven Claim, to the portion of such Proven Claim representing accrued but unpaid interest (including interest in respect of any secured portion of such Proven Claim).

7.5. <u>Distributions by Monitor</u>

The Monitor shall make all distributions required under this Plan in accordance with the provisions of this Plan, in general, and the provisions of **Section 5** and **Section 7** hereof, in particular. The Monitor shall receive, without further Court approval, reasonable compensation from the Company for distribution services rendered pursuant to the Plan.

7.6. Delivery of Distributions

- (a) Subject to Section 7.3 hereof, distributions shall be made by the Monitor (i) at the addresses set forth on the Proof of Claim form filed by the Affected Creditors (or at the last known addresses of such Affected Creditors if no Proof of Claim form was filed or if the Company or the Monitor has been notified in writing of a change of address), (ii) at the addresses set forth in any written notice of address change delivered to the Monitor after the date of any related Proof of Claim, or (iii) in a registered retirement savings plan account designated by any employee of the Company;
- (b) If any distribution to a Creditor is returned as undeliverable, no further distributions to such Creditor shall be made unless and until the Monitor is notified of the then-current address of such Creditor, at which time all missed distributions shall be made to such Creditor without interest. The Company shall make reasonable efforts to locate Affected Creditors for which distributions were undeliverable. Any claim for undeliverable distributions must be made on or before the later to occur of (i) 3 months after the Payment Date, or (ii) 3 months after such Creditor's Claim becomes a Proven Claim, after which date all unclaimed property shall revert to the Company free of any restrictions or claims thereon and the claim of any Creditor with respect to such property shall be discharged and forever barred, notwithstanding any applicable Laws to the contrary.

7.7. No Double Recovery

The aggregate recovery on account of any Proven Claim from all sources, regardless of whether on account of a theory of primary or secondary liability, by reason of guarantee, surety, indemnity, joint and several obligation or otherwise, shall not exceed (i) 100% of the underlying indebtedness, liability or obligation giving rise to such Claim or, (ii) where the underlying indebtedness, liability or obligation giving rise to such Claim is denominated in a currency (the "Original Currency") other than Canadian dollars, 100% of such underlying indebtedness, liability or obligation after conversion of the value of the distributions received in Canadian dollars back to the Original Currency at the Bank of

Canada noon spot rate of exchange for exchanging Canadian dollars to the Original Currency on the Determination Date.

7.8. Withholding Requirements

In connection with the Plan, any distribution made hereunder by the Monitor shall be made net of all applicable Taxes. Notwithstanding any other provision of the Plan, each Creditor that is to receive a distribution pursuant to the Plan shall have sole and exclusive responsibility for the satisfaction and payment of any Taxes imposed by any Governmental Authority (including income, withholding and other Taxes on account of such distribution). The Monitor, as necessary, shall be authorized to take any and all actions as may be necessary or appropriate to comply with such withholding and reporting requirements. All amounts withheld on account of Taxes shall be treated for all purposes as having been paid to the Creditor in respect of which such withholding was made, provided such withheld amounts are remitted to the appropriate Governmental Authority.

8. <u>IMPLEMENTATION OF THE PLAN</u>

8.1. Conditions Precedent to Implementation of Plan

The implementation of this Plan by the Company is subject to the occurrence and/or fulfillment of each of the following conditions precedent (the "Plan Implementation Conditions") on or prior to December 18, 2015 or such later date as may be ordered by the Court (the "Plan Implementation Deadline"), namely:

- (a) this Plan shall have been approved by Resolution enacted by the Required Majorities;
- (b) the Sanction Order sanctioning this Plan shall have been issued, and shall either (A) order its provisional execution notwithstanding appeal, or (B) not then be subject to any appeal therefrom, and the operation and effect of the Sanction Order shall not have been stayed, reversed or amended, and shall, among other things, declare and order that:
 - (i) this Plan, including the compromises and arrangements set out herein, is sanctioned and approved pursuant to Section 6 of the CCAA and, as at the Plan Implementation Date, will be effective and will enure to the benefit of and be binding upon the Company, the Affected Creditors and all other Persons referred to in this Plan or in the Sanction Order;
 - (ii) (A) this Plan has been approved by the Required Majorities of Affected Creditors of the Company in conformity with the CCAA; (B) the Company has complied with the provisions of the CCAA and the Orders of the Court

made in the CCAA Proceedings in all respects; (C) the Court is satisfied that the Company has neither done nor purported to do anything that is not authorized by the CCAA; and (D) this Plan and the transactions contemplated thereby are fair and reasonable;

- (iii) the full and final release and discharge of the Claims will be effective and will enure to the benefit of and be binding upon the Company, the Affected Creditors and all other Persons referred to in this Plan upon the issuance of the Certificate of Performance;
- (iv) the Company and the Monitor are authorized to take all steps and actions necessary to implement this Plan;
- (v) all Proven Claims determined in accordance with the Claims Procedure Order and this Plan are final and binding on the Company and all Affected Creditors;
- (vi) subject to Sections 4.5 and 4.6 hereof, any Claims for which a Proof of Claim has not been filed by the Claims Bar Date shall be forever barred and extinguished;
- (vii) all distributions and payments by or at the direction of the Monitor, in each case on behalf of the Company, under the Plan are for the account of the Company and the fulfillment of its obligations under the Plan;
- (viii) the Company and the Monitor may apply to the Court for advice and direction in respect of any matters arising from or under the Plan;
- subject to the performance by the Company of its obligations under this Plan, all contracts, leases, agreements and arrangements to which the Company are a party and that have not been disclaimed, terminated or repudiated pursuant to the Initial Order and/or the CCAA will be and remain in full force and effect, unamended, as at the Plan Implementation Date, and no Person who is a party to any such contract, lease, agreement or other arrangement may accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of dilution or other remedy) or make any demand under or in respect of any such contract, lease, agreement or other arrangement and no automatic termination will have any validity or effect, by reason of:
 - (A) any event that occurred on or prior to the Plan Implementation Date and is not continuing that would have entitled such Person to enforce those rights or remedies (including defaults, events of

default, or termination events arising as a result of the insolvency of the Company);

- (B) the insolvency of the Company, the fact that the Company filed a notice of intention to make a proposal under the BIA or the fact that the Company sought or obtained relief under the CCAA; or
- (C) any compromises or arrangements effected pursuant to this Plan or any action taken or transaction effected pursuant to this Plan;
- (x) the releases set forth in **Section 6.2** hereof and the injunction set forth in **Section 6.6** hereof are implemented and in effect;
- (xi) the commencement or prosecution, whether directly, derivatively or otherwise, or any demands, claims, actions, causes of action, counterclaims, suits or any indebtedness, liability, obligation or cause of action released and discharged pursuant to this Plan are enjoined; and
- (xii) the stay of proceedings as set forth in Section 6.8 hereof is and remains in full force and effect up until and including the Plan Implementation Date; and
- (c) the Replacement Financing shall have occurred.

8.2. Waiver of Conditions

Except for the condition set forth at Section 8.1(c) hereof, each of the Plan Implementation Conditions enure solely to the Company's benefit and, except for each of the conditions set forth in Sections 8.1(a) and 8.1(b)(i) hereof, may be waived in whole or in part either (i) solely by the Company (and no other Person and without the consent of any other Person), without any other notice to parties in interest or the Court and without a hearing, or (ii) by the Court. The failure to satisfy or waive any of the Plan Implementation Conditions prior to the Plan Implementation Date may be asserted by the Company regardless of the circumstances giving rise to the failure of such Plan Implementation Conditions to be satisfied (including any action or inaction by the Company). The failure of the Company to exercise any of the foregoing rights shall not be deemed a waiver of any other rights, and each such right shall be deemed an ongoing right that may be asserted at any time. Notwithstanding the foregoing, the condition set forth at Section 8.1(c) hereof may only be waived in writing by the Company and the DIP Lender.

8.3. Certificate of Implementation

Upon the occurrence and/or fulfillment of all of the Plan Implementation Conditions (or waiver thereof, as the case may be) on or prior to the Plan Implementation Deadline, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Implementation").

8.4. Effect of Plan

Upon the issuance of the Certificate of Performance, the settlement of the Affected Claims in accordance with the Plan shall become final and binding on the Company, and all the Affected Creditors and their respective successors and assigns, and this Plan shall, upon the Aggregate Distribution being fully remitted to the Monitor and the Monitor's issuance and filing with this Court of the Certificate of Performance, result in the full and final settlement of all Affected Claims, including for greater certainty, any Restructuring Claim and any Claims resulting, directly or indirectly, from the consequences and effects relating to the acceptance of the Plan by the Affected Creditors, its sanction by the Court, or its implementation, and any indebtedness, obligations or undertakings which the Company may subsequently become subject to, directly or indirectly, as a result of an obligation, transaction or an event that occurred before the Determination Date, as well as any indebtedness, obligations or any undertakings that the Company may be subject to on any date whatsoever in connection with the Plan, the approval thereof by the Court or the implementation thereof. For greater certainty, this Plan shall not affect or impair any rights, remedies and recourses which the Company had, has or may have after the issuance of the Certificate of Performance in connection with transactions, facts or obligations existing prior to the Determination Date.

8.5. Nullity of Plan

In the event that all of the Plan Implementation Conditions have not occurred and/or been fulfilled (or waived, as the case may be) on or prior to the Plan Implementation Deadline, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Non-Implementation"). Upon the issuance of such Certificate of Non-Implementation, any settlement of the Affected Claims shall automatically become null, void and of no effect whatsoever and shall remain owing by the Company and neither the Company, the Affected Creditors, the Monitor nor any other Person affected by this Plan shall be bound, obliged or affected by any of the provisions of this Plan.

9. <u>MISCELLANEOUS</u>

9.1. Confirmation of Plan

Upon occurrence and/or fulfillment of all of the Plan Implementation Conditions (or waiver thereof, as the case may be) on or prior to the Plan Implementation Deadline, this Plan shall be binding upon the Company, the Affected Creditors and any and all other Persons affected by the provisions of this Plan as well as each of their respective successors and assigns.

9.2. Paramountcy

From and after the Plan Implementation Date, any conflict between this Plan and/or the covenants, warranties, representations, terms, conditions, provisions or obligations, express or implied, of any contract, mortgage, security agreement, and/or indenture, trust indenture, loan agreement, commitment letter, agreement for sale, the by-laws of the Company, lease or other agreement, undertaking or any other source of obligations, written or oral and any and all amendments or supplements thereto existing between one or more of the Affected Creditors and the Company as at the Plan Implementation Date will be deemed to be governed by the terms, conditions and provisions of this Plan and the Sanction Order, which shall take precedence and priority. For greater certainty, all Affected Creditors shall be deemed irrevocably for all purposes to consent to all transactions contemplated in and by this Plan.

9.3. Modification of Plan

The Company:

in consultation with the Monitor and the DIP Lender (if any DIP Claim remains outstanding), reserves the right to file any modification of, or amendment or supplement to, this Plan by way of supplementary plan or plans of reorganization, compromise or arrangement (or any one or more thereof) at or before the Creditors' Meeting, in which case any such supplementary plan or plans of reorganization, compromise or arrangement (or an one or more thereof), shall, for all purposes, be and be deemed to form part of and be incorporated into this Plan. The Company shall file any supplementary plans with the Court as soon as practicable. The Company shall give notice to Affected Creditors of the details of any such modification, amendment or supplement at the Creditors' Meeting prior to the vote being taken to approve this Plan. The Company may give notice of a proposed modification, amendment or supplement to this Plan at or before the Creditors' Meeting by notice which shall be sufficient if given to those Affected Creditors present at such meeting in person or by proxy; and/or

(b) after the Creditors' Meeting (and both prior to and subsequent to the obtaining of the Sanction Order), the Company, in consultation with the Monitor and, if prior to the Plan Implementation Date, the DIP Lender (if any DIP Claim then remains), at any time and from time to time vary, amend, modify or supplement this Plan, (except to reduce the amount of the Aggregate Distribution or extend any Payment Dates other than as expressly provided herein), without the need for obtaining an Order of the Court or providing notice to the Affected Creditors if the Monitor determines that such variation, amendment, modification or supplement would not be materially prejudicial to the interests of the Affected Creditors under this Plan or the Sanction Order and is necessary in order to give effect to the substance of this Plan or the Sanction Order. All of the foregoing shall not require any further vote by or approval by the Affected Creditors or any approval by the Court.

9.4. <u>Deeming Provisions</u>

In this Plan, any deeming provisions are not rebuttable and are conclusive and irrevocable.

9.5. <u>Sections 38 and 95 to 101 BIA</u>

Notwithstanding Section 36.1 of the *CCAA*, Section 38 and Sections 95 through 101 of the *BIA* shall not apply to this Plan. Accordingly, neither the Monitor, any Creditor nor any other Person shall be entitled to exercise any right, remedy or recourse, or to commence any action, motion, application or any other proceeding against any member of the Fisher Group, any Creditor or any other Person in relation to the Company, based on Section 38 and Sections 95 through 101 of the *BIA*.

9.6. Responsibilities of the Monitor

The Monitor is acting in its capacity as Monitor in the CCAA Proceedings with respect to the Company and not in its personal or corporate capacity and will not be responsible or liable for any responsibilities or obligations of the Company under this Plan or otherwise, including with respect to the making of distributions or the receipt of any distribution by any Creditor or any other Person pursuant to the Plan. The Monitor will have the powers and protections granted to it by this Plan, the CCAA, the Initial Order, the Claims Procedure Order, the Meeting Procedure Order, the Sanction Order and any other Orders.

9.7. Liability Limitations

The Monitor, the Company and their respective legal counsel and other professional advisors, shall not be liable to any Person for any act or omission in connection with, or arising out of, the CCAA Proceedings, the Plan, the pursuit of sanctioning of the Plan,

the consummation and implementation of the Plan or the administration of the Plan or the funds to be distributed under the Plan, except for their own wilful misconduct or gross negligence.

9.8. Notices

- (a) Any notices or communication to be made or given hereunder to the Company or the Monitor shall be in writing and shall refer to this Plan and may, subject as hereinafter provided, be made or given by personal delivery, by courier, by prepaid mail, by fax or by e-mail, addressed to the respective parties as follows:
 - (i) if to the Company:

LAURA'S SHOPPE (P.V.) INC.

2955 Jules-Brillant Laval, Quebec H7P 6B2

Attention:

Kalman Fisher and

Josh Fisher

E-Mails:

kfisher@laura.ca

josh.fisher@laura.ca

Fax:

(450) 973-6099

with copy to:

KUGLER KANDESTIN LLP

1 Place Ville Marie Suite 2101 Montreal, Quebec H3B 2C6

Attention:

Gerald F. Kandestin,

David Stolow and

Jeremy Cuttler

E-Mail:

gkandestin@kklex.com

dstolow@kklex.com

jcuttler@kklex.com

Fax:

(514) 875-8424

(ii) if to the Monitor:

KPMG LLP
KPMG Tower
600 De Maisonneuve Blvd., West
Suite 1500
Montreal, Quebec
H3A 0A3

Attention:

Dev Coossa

E-Mail:

dcoossa@kpmq.ca

Fax:

(514) 840-2121

with copy to:

FISHMAN FLANZ MELAND PAQUIN, L.L.P. 1250 Rene-Levesque Blvd., West Suite 4100 Montreal, Quebec H3B 4W8

Attention:

Avram Fishman and

Mark Meland

E-Mails:

afishman@ffmp.ca

mmeland@ffmp.ca

Fax:

(514) 932-4170

or to such other address as any party may from time to time notify the others in accordance with this Section 9.8. In the event of any strike, lock-out or other event which interrupts postal service in any part of Canada, all notices and communications during such interruption may only be given or made by personal delivery, fax or e-mail and any notice or other communication given or made by prepaid mail within the 5 Business Day period immediately preceding the commencement of such interruption, unless actually received, shall be deemed not to have been given or made. All such notices and communications shall be deemed to have been received, in the case of notice by delivery, fax or e-mail prior to 5:00 p.m. (local time) on a Business Day, when received or if received after 5:00 p.m. (local time) on a Business Day or at any time on a non-Business Day, on the next following Business Day and, in the case of notice mailed as aforesaid, on the fourth Business Day following the date on which such notice or other communication is mailed. The unintentional failure by the Company or the Monitor to give any notice contemplated hereunder to any particular Creditor shall not invalidate this Plan or any action taken by any Person pursuant to this Plan.

(b) Any notices or communication to be made or given hereunder by the Monitor or the Company to a Creditor may be sent by fax, e-mail, ordinary mail, registered mail, courier or facsimile transmission to the e-mail address, address or fax number specified by such Creditor in its Proof of Claim, or in any subsequent written notice of change of address given to the Monitor. A Creditor shall be deemed to have received any document sent pursuant to this Plan 4 Business Days after the document is sent by ordinary or registered mail and on the Business Day immediately following the day on which the document is sent by courier, fax or e-mail.

9.9. Severability of Plan Provisions

If, prior to the Sanction Date, any term or provision of this Plan is held by the Court to be invalid, void or unenforceable, the Court, at the request of the Company which request shall be made in consultation with the Monitor and the DIP Lender (if any DIP Claim then remains), shall have the power to either:

- (a) sever such term or provision from the balance of this Plan and provide the Company with the option to proceed with the implementation of the balance of this Plan as of and with effect from the Plan Implementation Date; or
- (b) alter and interpret such term or provision to make it valid and enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpreted.

Notwithstanding any such severing, holding, alteration or interpretation, and provided the Company proceeds with the implementation of this Plan, the remainder of the terms and provisions of this Plan shall remain in full force and effect and shall in no way be affected, impaired or invalidated by such severing, holding, alteration or interpretation.

9.10. Revocation, Withdrawal or Non-Consummation

The Company, upon consultation with the Monitor and the DIP Lender, reserves the right to revoke or withdraw this Plan at any time prior to the Sanction Date and to file subsequent plans of arrangement and compromise. If the Company revokes or withdraws this Plan, or if the Sanction Order is not issued:

- (a) this Plan shall be null, void and inoperative in all respects:
- (b) any Claim, any settlement or compromise embodied in this Plan (including the fixing or limiting of any Claim to an amount certain), assumption or termination by this Plan, and any document or agreement executed pursuant to this Plan shall be deemed null, void and inoperative; and

- (c) nothing contained in this Plan, and no act taken in preparation for consummation of this Plan, shall:
 - (i) constitute or be deemed to constitute a waiver or release of any Claims by or against the Company or any other Person;
 - (ii) prejudice in any manner the rights of the Company or any Person in any further proceedings involving the Company; or
 - (iii) constitute an admission of any sort by the Company or any other Person.

9.11. Further Assurance

Notwithstanding that the transactions and events set out in this Plan shall occur and be deemed to occur in the order set out herein without any additional act or formality, each of the Persons affected hereby shall make, do and execute, or cause to be made, done and executed at the cost of the requesting party, all such further acts, deeds, agreements, transfers, assurances, instruments or documents as may reasonably be required by the Company in order to better implement this Plan.

9.12. Governing Law

This Plan shall be governed by and construed in accordance with the law of the Province of Quebec and the federal laws of Canada applicable therein. Any questions as to the interpretation or application of this Plan and all proceedings taken in connection with this Plan and its provisions shall be subject to the exclusive jurisdiction of the Court.

9.13. Successors and Assigns

This Plan shall be binding upon and shall enure to the benefit of the heirs, administrators, executors, legal personal representatives, successors and permitted assigns of the Company, the Directors, the Affected Creditors or any other Persons affected by or benefiting from the provisions of this Plan.

9.14. French Language Version of Plan

In the event of any discrepancy between any of the provisions of the English language version of this Plan and any French language version thereof, the provisions of the English version of this Plan shall, under all circumstances, prevail and govern.

9.15. Choice of Language

The Company and each of the hereafter enumerated members of the Fisher Group acknowledges that it (or he) has required that this Plan and all related documents be

prepared, in English. La Compagnie et chacun des membres du "Fisher Group" ci-après énumerés reconnaît avoir exigé que ce Plan et tous les documents connexes soient rédigés en anglais.

(signature page to follow)

٠

takei on 3.1
1-14-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-

EXHIBIT P-4

(Compared version between the Initial Plan and the Re-Amended Plan)

COMPARED: ORIGINAL TO RE-AMENDED

CANADA

PROVINCE OF QUEBEC DISTRICT OF MONTREAL

NO.: 500-11-049256-155

SUPERIOR COURT (Commercial Division)

(Sitting as a court designated pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. c. C-36)

IN THE MATTER OF THE PLAN OF COMPROMISE AND ARRANGEMENT OF:

MAGASIN LAURA (P.V.) INC. / LAURA'S SHOPPE (P.V.) INC.

Debtor/Petitioner

-and-

BOUTIQUE LAURA CANADA LTÉE / LAURA'S SHOPPE CANADA LTD.

-and-

3482731 CANADA INC.

-and-

9318-5494 QUEBEC INC.

-and-

KALMAN FISHER

Stayed Parties

-and-

KPMG INC.

Monitor

RE-AMENDED PLAN OF COMPROMISE AND ARRANGEMENT Under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36

SECTION 1 TABLE OF CONTENTS

<u>1.</u>	INTERPRETATION
<u>1.1.</u>	
<u>1.2.</u>	Interpretation10
<u>1.3.</u>	Date For Any Action10
<u>1.4.</u>	Statutory References11
<u>2.</u>	COMPROMISE AND ARRANGEMENT 1
<u>2.1.</u>	
<u>2.2.</u>	Classes of Affected Claims11
<u>2.3.</u>	Disclaimed Landlord Claims11
<u>2.4.</u>	Unaffected Claims11
<u>2.5.</u>	Treatment of Affected Claims12
<u>2.6.</u>	Settlement of Affected Claims13
<u>2.7.</u>	Extension of Payment Dates13
2.8.	Prepayment Rights14
<u>2.9.</u>	No Acknowledgment or Admission14
3.	TREATMENT OF OTHER CLAIMS AND MATTERS 14
<u>3.1.</u>	Treatment of the DIP Claim14
<u>3.2.</u>	Treatment of Excluded Claims14
<u>3.3.</u>	Treatment of Administration Claims14
3.4.	Treatment of Gift Card Claims14
<u>3.5.</u>	Treatment of Certain Crown Priority Claims15
<u>3.6.</u>	Treatment of Section 6(5) Claims
3.7.	Treatment of Section 19(2) Claims
<u>3.8.</u>	Treatment of KERP Claims15
3.9.	Treatment of Secured Claims15
3.10	
3.11	. Treatment of Fisher Group Claims
4.	VALUATION OF CLAIMS, CREDITORS' MEETING AND RELATED MATTERS 16
4.1.	Conversion of Affected Claims into Canadian Currency

<u>4.2.</u>	Affected Claims	<u></u> 16
<u>4.3.</u>	Creditors' Meeting	<u></u> 16
4.4.	Approval by Affected Creditors	<u></u> 16
<u>4.5.</u>	Amendment of Proofs of Claim	<u></u> 17
<u>4.6.</u>	Late Filing of Claims for Disclaimed Leases	<u></u> 17
4.7.	Claims Bar Date	<u></u> 17
<u>5. PRO</u>	OCEDURE FOR RESOLVING DISTRIBUTIONS IN RESPECT OF DISPUTE	D CLAIMS 17
<u>5.1.</u>	No Distributions Pending Allowance	17
5.2.	Distribution From Reserve Once Disputed Claims Resolved	<u></u> 18
6. EFF	FECT OF THE PLAN AND RELEASES	18
6.1.	Effect of the Plan	<u></u> 18
6.2.	Releases upon Implementation	<u></u> 18
6.3.	Releases after Performance	<u></u> 19
6.4.	Exception to Releases	<u></u> 19
6.5.	Set-Off Rights	<u></u> 20
6.6.	Injunction Related to Releases	<u></u> 20
<u>6.7.</u>	Waiver of Defaults	<u></u> 20
6.8.	Stay of Proceedings	<u></u> 21
7. PRC	DVISIONS GOVERNING DISTRIBUTIONS	21
7.1.	Partial Distributions for Claims Allowed	<u></u> 21
7.2.	Currency to be used for the Distribution	<u></u> 21
<u>7.3.</u>	Assignment of Claims	<u></u> 21
7.4.	Interest on Affected Claims	<u></u> 22
7.5.	Distributions by Monitor	<u></u> 22
7.6.	Delivery of Distributions	<u></u> 22
7.7.	No Double Recovery	<u></u> 23
7.8.	Withholding Requirements	23
8. IMPI	LEMENTATION OF THE PLAN	<u> 23</u>
8.1.	Conditions Precedent to Implementation of Plan	23
8.2.	Waiver of Conditions	<u>.</u> 25
8.3.	Certificate of Implementation	<u></u> 26
8.4.	Effect of Plan	<u></u> 26
8.5.	Nullity of Plan	26

9.	MISCELLANEOUS27
<u>9.1.</u>	Confirmation of Plan27
<u>9.2.</u>	Paramountcy27
<u>9.3.</u>	Modification of Plan27
9.4.	Deeming Provisions28
9.5.	Sections 38 and 95 to 101 BIA28
9.6.	Responsibilities of the Monitor28
9.7.	Liability Limitations29
9.8.	Notices29
9.9.	Severability of Plan Provisions31
9.10	. Revocation, Withdrawal or Non-Consummation31
<u>9.11</u>	. Further Assurance32
9.1 <u>2</u>	. Governing Law32
9.13	
9.14	. French Language Version of Plan
9.15	. Choice of Language

1. <u>INTERPRETATION</u>

1.1. Definitions

In this Plan, unless otherwise stated or the context otherwise requires:

"Administration Charge" has the meaning ascribed to such term in the Initial Order;

"Administration Claim" means a claim or any other indebtedness or obligation secured by the Administration Charge;

"Affected Claim" means any Claim other than an Unaffected Claim. For greater certainty, Affected Claims include Disclaimed Landlord Claims, any Claim of Her Majesty the Queen in right of Canada or any province (other than Crown Priority Claims), Directors Claims and Restructuring Claims;

"Affected Creditor" means a Creditor holding an Affected Claim, but only to the extent of its Affected Claim;

"Affected Creditors Class" has the meaning set forth in Section 2.22.2 hereof;

"Aggregate Distribution" has the meaning set forth in Section 2.42.52.5 hereof;

"BIA" means the Bankruptcy and Insolvency Act, R.S.C., 1985, c. B-3, as amended;

"Business Day" means a day, other than a Saturday, a Sunday, or a non-juridical day (as defined in article 6 of the Quebec Code of Civil Procedure, R.S.Q., c. C-25, as amended);

"CCAA" means the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended;

"CCAA Charge(s)" means the Administration Charge, the DIP Lender Charge and the KERP Charge;

"CCAA Proceedings" means proceedings in respect of the Company before the Court commenced, taken up and continued under the CCAA;

"Certificate of Implementation" has the meaning set forth in Section 8.38.3 hereof;

"Certificate of Performance" has the meaning set forth in Section 6.16.4 hereof;

"Certificate of Non-Implementation" has the meaning set forth in Section 8.58.5 hereof;

"Chair" means a representative of the Monitor, or any other individual designated by the Monitor to preside as chairperson at the Creditors' Meeting;

"Claim" means any right of any Person against the Company in connection with any indebtedness, liability or obligation of any kind of the Company owed to such Person and any interest, or penalties accrued thereon or costs payable in respect thereof, whether liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, unsecured, present, future, known or unknown, by guarantee, surety or otherwise, and whether or not such right is executory or anticipatory in nature, including the right or ability of any Person to advance a claim for contribution or indemnity or otherwise with respect to any matter, action, cause or chose in action, whether existing at present or commenced in the future, which indebtedness, liability or obligation is based in whole or in part on facts existing, or transactions which occurred, prior to the Determination Date, or which would have been claims provable in bankruptcy had the Company become bankrupt on the Determination Date, and, without limitation, including any Restructuring Claim, but excluding any Unaffected Claim and "Claims" means all of them:

"Claims Bar Date" has the meaning as set forth in the Claims Procedure Order;

"Claims Procedure Order" means the "Claims Procedure Order" issued by the Court on September 11, 2015, establishing, among other things, procedures for proving Claims, as amended or supplemented from time to time by further Order(s) of the Court;

"Company", "Debtor" or "Petitioner" means Magasin Laura (P.V.) Inc./Laura's Shoppe (P.V.) Inc.;

"Court" means the Commercial Division of the Quebec Superior Court for the District of Montreal;

"Creditor(s)" means any Person(s) having a Claim and may, where the context requires, include the assignee of a Claim or a trustee, interim receiver, receiver, receiver and manager, or other Person acting on behalf of such Person(s);

"Creditors' Meeting" means the meeting of Affected Creditors to be convened for the purposes of voting on the Plan, or any adjournment of such meeting;

"Crown Priority Claims" means any Claims of Her Majesty the Queen in right of Canada or in right of any province as described in Section 6(3) or Section 38(2) of the CCAA. For greater certainty, any Claim of Her Majesty the Queen in right of Canada or in right of any Province other than Crown Priority Claims shall be an Affected Claim hereunder;

"Determination Date" means July 31, 2015, being the date on which the Company commenced proceedings under the *BIA* by filing a notice of intention to make a proposal thereunder, which proceedings were subsequently taken up and continued by the *CCAA* Proceedings under the *CCAA*;

"DIP Claim" means any and all obligations of the Company to the DIP Lender pursuant to the DIP Facility Documents or the Initial-Order for New Financing as of or after the Plan Implementation Date;

"DIP Facility Documents" means the "Interim Financingcommitted term sheet" dated August 10October 22, 2015 by and among, inter alia, the Company and the DIP Lender (as amended, modified, restated and/or supplemented from time to time) together with any related collateral, loan or security documents executed in connection therewith or which relate thereto;

"DIP Lender" means The Cadillac Fairview Corporation Limited;

"DIP Lender" means collectively, Third Eye Capital Corporation, acting as agent and hypothecary representative for and on behalf of certain affiliates and funds it manages as well as all of the "New Financing Parties" as defined in the Order for New Financing:

"DIP Lender Charge" means the "InterimNew Lender Charge" as set forth in the Initial Order for New Financing;

"DIP Loan" means the loan(s) made and/or to be made by the DIP Lender as envisaged by the Order for New Financing;

"Directors" means all of the Company's past and present directors as well as any Persons who were or are deemed to be directors of the Company pursuant to any applicable Laws;

"Directors' Claims" means all Claims, of any nature or source whatsoever, of any of the Creditors against any or all of the Directors which arose before the Determination Date and which relate to obligations of the Company where such Directors are by law liable in their capacity as directors for payment of such obligations. Director's Liability shall, however, exclude all Claims which (i) relate to contractual rights of one or more Creditors arising from contracts with one or more Directors, including any guarantees given by any Directors relating to the Company's obligations, (ii) seek injunctive relieve against any of the Directors in relation to the Company, or (iii) are based on allegations of misrepresentation made by any Directors to Creditors or of wrongful or oppressive conduct by Directors;

"Disallowed Claim" means any Claim, or that portion thereof which has been revised or disallowed by the Monitor pursuant to the Claims Procedure Order or any other Order in

respect of which an appeal by the Creditor has been dismissed or all appeal periods, if any, have expired;

"Disclaimed Leases" means each and every one of the Leases and/or other agreements for the occupancy of real or immovable property (as well as all amendments thereto and renewals thereof) which (i) were disclaimed or resiliated by the Company pursuant to Section 60 (1.1)65.2 of the BIA (as recognized by the Initial Order) or Section 32 of the CCAA, as exhaustively enumerated in the annex forming part of this Plan, and (ii) may hereafter be disclaimed or resiliated by the Company pursuant to Section 32 of the CCAA, to the extent (in either case) thatat any such disclaimers or resiliations have not been or shall not hereafter be rescinded by the Companytime between the date of this Plan and the Sanction Date, and "Disclaimed Lease" means each of them;

"Disclaimed Landlord" means the landlord under any Disclaimed Lease and "Disclaimed Landlords" means all of them:

"Disclaimed Landlord Claims" means all Proven Claims of the Disclaimed Landlords for losses incurred by Disclaimed Landlords in relation to Disclaimed Leases, as provided for in Section 32(7) of the CCAA2.32.3 hereof and "Disclaimed Landlord Claim" means any of them;

"Disputed Claim" means a Claim or that portion thereof that is the object of a Notice of Revision or Disallowance and, in either case, has not become a Proven Claim or a Disallowed Claim:

"Excluded Claim" means any right of any person against the Company in connection with any indebtedness, liability or obligation of any kind which arose with respect to transactions which occurred after the Determination Date and any interest thereon, including the Administration Claim, the DIP Claim, a KERP Claim, the Replacement Financing Claim and any obligation of the Company towards creditors who have supplied or shall supply services, utilities, goods or materials or who have or shall have advanced funds to the Company after the Determination Date, but only to the extent of their claims in respect of the supply of such services, utilities, goods, materials or funds after the Determination Date and to the extent that such claims are not otherwise affected by the Plan. For greater certainty, a Restructuring Claim is not an Excluded Claim;

"Excluded Creditor" means a Person having a claim in respect of an Excluded Claim but only in respect of such Excluded Claim and to the extent that the Plan does not otherwise affect such claim;

"First Released Parties" has the meaning as set forth in Section 6.26.2 hereof;

"Fisher Group" means Kalman Fisher, 3482731 Canada Inc., 9318-5494 Quebec Inc. (formerly 7735235 Canada Inc.) and any Person who/which is, or was at any time within the 12 consecutive months immediately preceding the Determination Date, a Related Person to Kalman Fisher;

"Fisher Group Claims" means any Claims for payment of principal on any loans or advances made to the Company (or its predecessor) at any time prior to the Determination Date by any member of the Fisher Group, which includes any Secured Claim by any member of the Fisher Group, but excludes any Claims or future claims by any member of the Fisher Group for (i) salary (or other remuneration), (ii) reimbursement of business expenses, or (iii) payment of principal on any loans or advances in lieu of salary (or other remuneration);) or in lieu of interest:

"Gift Card Claim" has the meaning set forth in Section 2.3 (a)(v)2.4(a)(v) hereof;

"Governmental Authority" means any (i) multinational, national, provincial, state, regional, municipal, local or other government, governmental or public department, ministry, central bank, court, tribunal, arbitral body, commission, board, official, minister, bureau or agency, domestic or foreign, (ii) subdivision, agent, commission, board or authority of any of the foregoing; or (iii) quasi-governmental or private body, including any tribunal, commission, regulatory agency or self-regulatory organization, exercising any regulatory, expropriation or taxing authority under, or for the account of, any of the foregoing;

"Initial Order" means the "Initial Order" issued by the Court on August 12, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"KERP Charge" has the meaning ascribed to such term in the Initial Order;

"KERP Claim" means a claim or any other indebtedness or obligation secured by the KERP Charge and "KERP Claims" means all of them;

"Laws" means all laws, statutes, codes, ordinances, decrees, rules, regulations, by-laws, judicial or arbitral or administrative or ministerial or departmental or regulatory judgments, injunctions, orders or decisions of any Governmental Authority, statutory body or self-regulatory authority, including general principles of law having the force of law and the term "applicable" with respect to such Laws and, in the context that refers to any Person, means such Laws as are applicable to such Person or its business, undertaking, property or securities and emanate from a Governmental Authority or self-regulatory authority having jurisdiction over the Person or its business, undertaking, property or securities;

"Lease" means any lease and/or other agreement for the occupancy of real and immovable property as well as all amendments thereto and renewals thereof and "Leases" means all of them;

"Meeting Date" means the date fixed for the Creditors' Meeting in accordance with the Meeting Procedure Order, or any subsequent Order, or any subsequent date following any adjournment of that meeting, as the case may be;

"Meeting Procedure Order" means the "Meeting Procedure Order" issued or to be issued by the Court upon or following the filing of this Plan withon October 23, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"Monitor" means KPMG Inc., in its capacity as Monitor, as appointed by the Court pursuant to the Initial Order;

"Notice of Revision or Disallowance" has the meaning as set forth in the Claims Procedure Order:

"Order" means any order of the Court in the CCAA Proceedings;

"Order for New Financing" means the "Order for New Financing" issued by the Court on October 30, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"Original Currency" has the meaning set forth in Section 7.77.7 hereof;

"Payment Date" and "Payment Dates" have the meaning set forth in Section 2.42.52.5 hereof:

"Person" means any individual, corporation, limited or unlimited liability company, general or limited partnership, association, trust, trustee, executor, administrator, legal personal representative, estate, unincorporated organization, joint venture, governmental body or agency, or any other entity;

"Plan" means the present "Re-amended Plan of Compromise and Arrangement" of the Company pursuant to the provisions of the CCAA, as may be amended, varied or supplemented by the Company from time to time in accordance with its terms;

"Plan Implementation Conditions" has the meaning set forth in Section 8.18.4 hereof;

"Plan Implementation Date" means the date, on or prior to the Plan Implementation Deadline, upon which all of the Plan Implementation Conditions have occurred or have been satisfied or waived;

"Plan Implementation Deadline" has the meaning set forth in Section 8.18.4 hereof;

"Proof of Claim" has the meaning set forth in the Claims Procedure Order;

"Proven Claim" means, in respect of a Creditor, the amount of the Claim of such Creditor as finally determined for distribution purposes in accordance with the provisions of this Plan, the CCAA and the Claims Procedure Order and "Proven Claims" means all of them;

"Related Persons" has the meaning set forth in Section 4 of the BIA;

"Released Party" and "Released Parties" have the meanings as set forth in Section 6.2 hereof:

"Replacement Lender(s)"Financing" means the Person(s) who will provide the Replacement Loan;

"Replacement Loan" means the loans and/or othernew senior credit facilities to be provided to the Company by the Replacement Lender(s) in order DIP Lender, including a term loan facility and a revolving asset-based loan facility on terms and conditions acceptable to the DIP Lender and the Company, which shall completely discharge and release (i) any then remaining DIP Claim and the DIP Lender Charge, and (ii) the Salus Claim and the Salus Securityrepay and/or replace the DIP Loan;

"Replacement Financing Claim" means any and all indebtedness and obligations of the Company and the DIP Lender under the Replacement Financing;

"Required Majorities" means the affirmative vote of a majority in number of the Affected Creditors voting in the Affected Creditors Class, having Voting Claims and voting on the Plan (in person or by proxy) at the Creditors' Meeting and representing not less than 66-2/3% in value of the Voting Claims of the Affected Creditors voting (in person or by proxy) at the Creditors' Meeting;

"Reserve(s)" means the reserve(s) to be established and maintained under this Plan by the Monitor by holding, on account of Disputed Claims, an amount equal to the aggregate amount that the holders of Disputed Claims would be entitled to receive if all such Disputed Claims had been Proven Claims as of any Payment Date;

"Resolution" means, collectively, one or any of, the resolutions providing for the approval of the Plan by the Affected Creditors;

"Restructuring Claim" means any right or claim of any Person against the Company in connection with any indebtedness or obligation of any kind owed to such Person arising out of the Company's restructuring under the CCAAdisclaimer, resiliation or

renegotiation (at any time after the Determination Date) of any Leases (excluding the Disclaimed Landlord Claims, which shall be dealt with pursuant to Section 2.32.3 hereof), agreements or other contracts or the Company's restructuring under the CCAA, and includes any right or claim resulting, directly or indirectly, from the consequences and effects of the Plan's acceptance by the Affected Creditors, the Plan's sanctions sanction by the Sanction Order, the Plan's implementation and any debt forgiveness resulting from any of the foregoing; provided, however, that a Restructuring Claim shall not include an Excluded Claim. Without limiting the generality of the foregoing, a Restructuring Claim shall include a Disclaimed Landlord Claim, but shall exclude any claim of any landlord in respect of any lease for real or immovable property other than claims of Disclaimed Landlords. For greater certainty, a Restructuring Claim is an Affected Claim;

"Salus" means Salus Capital Partners, LLC or any Person acting on its behalf, in its place or as its successor;

"Salus Claim" means, at any given time, all indebtedness and obligations owing by the Company to Salus;

"Salus Security" means all mortgages, hypothess, pledges, charges, liens, privileges, security interests, guarantees, suretyships or other rights securing and/or guaranteeing the Salus Claim, held by or on behalf of Salus (i) over any and all present and future property of the Company, any of the Stayed Parties, and/or any other Person, and/or (ii) against the Company, any of the Stayed Parties and/or any other Person;

"Sanction Date" means the date on which the Sanction Order is issued;

"Sanction Order" means the Order to be made under the CCAA sanctioning the Plan, as such Order may be affirmed, amended or modified by the Court at any time prior to the Plan Implementation Date or, if appealed, then, unless such appeal is withdrawn, abandoned or denied, as affirmed or amended on appeal, in form and content which is satisfactory to the Petitioner acting reasonably;

"Second Released Party" and "Second Released Parties" have the meanings as set forth in Section 6.36-3 hereof;

"Section 6(5) Claim" means any claim of an employee or former employee described in Section 6(5) of the CCAA but only to the extent of such amounts as required to be paid under the CCAA;

"Section 19(2) Claims" means any claim described in Section 19(2) of the CCAA;

"Secured Claim" means the Claim of a Secured Creditor, to the extent of the value of such Secured Creditor's security;

"Secured Creditor" has the meaning set forth in the CCAA, but only to the extent that such Creditor's mortgage, hypothec, pledge, charge, lien, privilege, security interest or other rights over the property of the Company was valid, opposable, perfected, and could be set up against third parties, including a trustee to the Company's bankruptcy, on both the Determination Date and the Claims Bar Date, failing which that Secured Creditor will be deemed to be an "unsecured creditor", as defined in the CCAA;

"Stayed Parties" means Boutique Laura Canada Ltée/Laura's Shoppe Canada Ltd., 3482731 Canada Inc., 9318-5494 Quebec Inc. (formerly 7735235 Canada Inc.) and Kalman Fisher;

"Taxes" means all federal, state, provincial, territorial, county, municipal, local or foreign taxes, duties, imposts, levies, assessments, tariffs and other charges imposed, assessed or collected by a Governmental Authority, including (i) any gross income, net income, gross receipts, business, royalty, capital, capital gains, goods and services, value added, severance, stamp, franchise, occupation, premium, capital stock, sales and use, real property, land transfer, personal property, ad valorem, transfer, licence, profits, windfall profits, environmental, payroll, employment, employer health, pension plan, anti-dumping, countervail, excise, severance, stamp, occupation, or premium tax, (ii) all withholdings on amounts paid to or by the relevant Person, (iii) all employment insurance premiums, Canada, Quebec and any other pension plan contributions or premiums, (iv) any fine, penalty, interest, or addition to tax, (v) any tax imposed, assessed, or collected or payable pursuant to any tax-sharing agreement or any other contract relating to the sharing or payment of any such tax, levy, assessment, tariff, duty, deficiency, or fee, and (vi) any liability for any of the foregoing as a transferee, successor, guarantor, or by contract or by operation of law;

"Unaffected Claim" means any right of any Person in connection with any indebtedness, liability or obligation of any kind of the type described in Section 2.3 (a)2.4(a)2.4(a) hereof. For greater certainty, the Unaffected Claims include the Excluded Claims;

"Unaffected Creditor" means a Person having a Claim in respect of an Unaffected Claim, but only in respect of such Unaffected Claim, and for greater certainty, includes an Excluded Creditor;

"Voting Claim" means, in respect of a<u>an Affected</u> Creditor, the amount of such <u>Affected</u> Creditor's claim which has been accepted for voting purposes in accordance with the provisions of this Plan, the Claims Procedure Order and the *CCAA* and "Voting Claims" means all of them;

1.2. Interpretation

For purposes of this Plan:

- (a) any reference in this Plan to a contract, instrument, release, indenture, agreement or other document being in a particular form or on particular terms and conditions means that such document shall be substantially in such form or substantially on such terms and conditions;
- (b) any reference in this Plan to an existing document or exhibit filed or to be filed means such document or exhibit as it may have been or may be amended, modified or supplemented;
- (c) all references to currency and to "\$", "C\$" or "Dollars" are to Canadian dollars except as otherwise indicated;
- (d) all references in this Plan to Sections are references to Sections of this Plan;
- (e) unless otherwise specified, the words "hereof", "herein" and "hereto" refer to this Plan in its entirety rather than to any particular portion of this Plan;
- (f) the division of this Plan into Sections and paragraphs and the insertion of captions and headings to Sections and paragraphs are for convenience of reference only and are not intended to affect the interpretation of, or to be part of this Plan;
- (g) where the context requires, a word or words importing the singular shall include the plural and vice versa and a word or words importing one gender shall include all genders;
- (h) the words "includes" and "including" are not limiting; and
- (i) the word "or" is not exclusive.

1.3. Date For Any Action

In the event that any date (including any Payment Date) on which any action (including any payment) is required to be taken under this Plan by any of the parties is not a Business Day, that action (including any payment) shall be required to be taken on the next succeeding day which is a Business Day.

1.4. Statutory References

Any reference in this Plan to a statute includes all regulations made thereunder and all amendments to such statute or regulations in force, from time to time, or any statute or regulations that supplement or supersede such statute or regulations.

SECTION 2

2. <u>COMPROMISE AND ARRANGEMENT</u>

2.1. Persons Affected

The purpose of this Plan is to provide for the full and final settlement of all Affected Claims. Except as specifically provided for in this Plan, this Plan will become effective on the Plan Implementation Date in accordance with its terms, and all Affected Claims against the Company will be fully and finally, settled, compromised and released in this Plan upon the Aggregate Distribution being fully paid to the Monitor, it being understood that any Affected Claim that is paid in full pursuant to this Plan before such date shall be settled, compromised and released on the date of such payment. This Plan shall be binding on and enure to the benefit of the Company, the Affected Creditors, the First Released Parties, the Second Released Parties, any trustee, agent or other Person acting on behalf of any Affected Creditor and such other Persons who have received the benefit of, or are bound by any waivers, releases or indemnities hereunder.

2.2. Classes of Affected Claims

There shall only be one class of Affected Creditors for the purpose of voting on, and receiving distributions pursuant to this Plan, being the "Affected Creditors Class".

2.3. Disclaimed Landlord Claims

<u>Disclaimed Landlord Claims shall be Affected Claims under this Plan and shall be determined in accordance with Section 32(7) CCAA.</u>

2.3.2.4. Unaffected Claims

- (a) This Plan does not affect the following claims (each, an "Unaffected Claim" and, collectively, the "Unaffected Claims"), the holders of which will not be entitled to vote at the Creditors' Meeting or receive any distributions under this Plan, namely:
 - (i) the Excluded Claims;
 - (ii) any Administration Claim;

- (iii) any DIP Claim;
- (iv) any KERP Claims;
- any claim with respect to gift-cards, gift certificates, lay-away deposits and other customer certificates (collectively, "Gift Card Claims");
- (vi) Crown Priority Claims;
- (vii) Section 6(5) Claims;
- (viii) Section 19(2) Claims;
- (ix) Secured Claims;
- (x) the Salus Replacement Financing Claim; and
- (xi) Fisher Group Claims,

all of which will be dealt with in Section 33 hereof;

(b) Nothing in this Plan shall affect the Company's rights and defences, both legal and equitable, with respect to any Unaffected Claim including any rights arising under or pursuant to the Claims Procedure Order or this Plan or any rights with respect to legal and equitable defences or entitlements to set-offs or recoupments against such Unaffected Claims.

2.4.2.5. Treatment of Affected Claims

By no later than each of the payment dates hereafter set forth (collectively the "Payment Dates" or individually a "Payment Date"), the aggregate sum of \$4 Million (the "Aggregate Distribution"), without any interest whatsoever thereon, shall be paid by the Company to the Monitor and shall thereafter be distributed by the Monitor to the Affected Creditors, according to the amount of their respective Proven Claims as hereafter set forth, namely:

- (a) by no later than April 30, 2016, \$1 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor as follows:
 - (i) an amount equal to the lesser of (A) the Proven Claim of each Affected Creditor, or (B) \$1,250, shall be distributed by the Monitor to the Affected Creditors, according to the amount of their Proven Claims; and

- (ii) an amount equal to the difference between (A) \$1 Million, and (B) the aggregate amount to be distributed by the Monitor pursuant to Section 2.4 (a)(i)2.52.5(a)(a)(i)(i) above, shall be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the amounts set forth in Section 2.4 (a)(i)2.52.5(a)(a)(i)(i) above;
- (b) by no later than July 31, 2016, \$1 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distribution under Section 2.4 (a)2.52.5(a)(a) hereof;
- (c) by no later than December 15, 2016, \$1.8 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distributions under Sections 2.4 (a) and 2.4 (b)2.52.5(a)(a) and 2.52.5(b)(b) hereof; and
- (d) by no later than February 15, 2017, \$200,000 shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a *pro rata* basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distributions under Sections 2.4 (a), 2.4 (b) and 2.4 (e)2.52.5(a)(a), 2.52.5(b)(b) and 2.52.5(c)(e) hereof.

2.5.2.6. Settlement of Affected Claims

The payment of the entire Aggregate Distribution to the Monitor as set forth in **Section 22** hereof shall constitute the full and final payment, reduction, settlement and transaction (as envisaged by the relevant provisions of the *Civil Code of Quebec*) of all of the Affected Claims of all of the Affected Creditors and engage the releases and other provisions set forth in this Plan.

2.6.2.7. Extension of Payment Dates

Any or all of the Payment Dates may be extended by the Company with the consent of the Monitor; and the DIP Lender (if any portion of the DIP Loan or the Replacement Financing then remains outstanding), by additional periods not exceeding, in each case, 60 days. On application by the Company or, the Monitor (in either case, with the consent of the DIP Lender, if any portion of the DIP Loan or the Replacement Financing remains outstanding) made prior to a Payment Date or, if applicable, prior to the expiry of any extension thereof as described above, the Court may issue Orders extending (or

further extending) any or all of the Payment Dates for any additional period(s) deemed appropriate by the Court in the then existing circumstances.

2.7.2.8. Prepayment Rights

The Company The Company (with the consent of the DIP Lender, if any portion of the DIP Loan or the Replacement Financing then remains outstanding) shall, at all times prior to the Final Distribution Date, be entitled to pre-pay the whole or any amount of the Aggregate Distribution prior to any Payment Date, whereupon the amount(s) of any such pre-payment(s) shall be distributed by the Monitor to the Affected Creditors in accordance with the provisions of **Section 2.42.52.5** hereof, mutatis mutandis.

2.9. SECTION 3No Acknowledgment or Admission

Nothing contained in this Plan shall constitute, under any circumstances whatsoever, an acknowledgment or admission by the Company as to the existence or validity of any Claims or Restructuring Claims.

3. TREATMENT OF OTHER CLAIMS AND MATTERS

3.1. Treatment of the DIP Claim

On or before the earlier of (i) the Plan Implementation Date, or (ii) the date on which the DIP Claim becomes payable pursuant to the provisions of the DIP Facility Documents, the DIP Lender shall receive full payment of any remaining DIP Claim.

3.2. Treatment of Excluded Claims

Subject to Section 3.13.4 hereof, the Excluded Claims (other than the Administration Claim, the DIP Claim and the any KERP Claim) will remain in full force and effect in accordance with their terms after the Plan Implementation Date, and will be paid in full by the Company in the normal course of its business as and when they become due.

3.3. <u>Treatment of Administration Claims</u>

All Administration Claims, if any, will be paid in full by the Company as and when they become due, and any remaining balance will be settled as soon as practicable after the Plan Implementation Date.

3.4. Treatment of Gift Card Claims

Gift Card Claims will be honored in accordance with the terms of the relevant gift card, gift certificate, lay-away deposit or other customer certificates upon presentation of such gift card, gift certificate, lay-away deposit or other customer certificates by the holder at any of the Petitioner's retail stores.

3.5. Treatment of Certain Crown Priority Claims

All Crown Priority Claims described in Section 6(3) of the CCAA, if any, will be paid in full by the Petitioner within 6 months immediately following the Sanction Date;

3.6. Treatment of Section 6(5) Claims

All Section 6(5) Claims, if any, will be paid in such amounts as required under the CCAA immediately after the Sanction Date.

3.7. Treatment of Section 19(2) Claims

All Section 19(2) Claims, if any, will be paid by the Company as and when they become due, unaffected by this Plan.

3.8. Treatment of KERP Claims

Each KERP Claim will be paid by the Company as and when it becomes due.

3.9. Treatment of Secured Claims

Secured Claims (other than the Administration Claim, the DIP Claim, the Replacement Financing Claim and KERP Claims—and the Salus Claim) will be dealt with as provided for under agreements entered into or as may be hereafter entered into between the Company and the relevant Secured Creditor, or as may be ordered by the Court. For greater certainty, Secured Claims will be unaffected by this Plan.

3.10. Treatment of Salus Replacement Financing Claim

The Salus Claim and the Salus Security will be dealt with as provided for under agreements entered into or as may be hereafter entered into between the Company and Salus, or as may be ordered by the Court. For greater certainty, the Salus Claim will be unaffected by this Plan.

The Replacement Financing Claim shall be paid by the Company to the DIP Lender in accordance with the provisions, terms and conditions of the Replacement Financing.

3.11. Treatment of Fisher Group Claims

Conditional upon the enactment of a Resolution approving the Plan, issuance of the Sanction Order and implementation of the Plan as set forth in **Section 88** hereof, each member of the Fisher Group:

- (a) subordinates and postpones all of the Fisher Group Claims to and in favour of full payment of the <u>Aggregate</u> Distribution—Amount to the Affected Creditors, such that no member of the Fisher Group shall be entitled to receive any payment from the Company of any principal on any of the Fisher <u>Group</u> Claims unless and until the full Aggregate Distribution shall have been paid to the Monitor as set forth in the Plan; and
- (b) waives and renounces to any right to be an Affected Creditor.

SECTION-4

4. <u>VALUATION OF CLAIMS, CREDITORS' MEETING AND RELATED MATTERS</u>

4.1. Conversion of Affected Claims into Canadian Currency

For the purposes of determining the value of Affected Claims denominated in currencies other than Canadian dollars for voting purposes, such Affected Claims shall be converted by the Monitor to Canadian dollars at the Bank of Canada noon spot rate of exchange for exchanging currency to Canadian dollars on the Determination Date (which in the case of US Dollars was C\$1.3080 for US\$1.00).

4.2. Affected Claims

Affected Creditors shall be entitled to prove their respective Affected Claims, vote their Voting Claims in respect of the Plan, and, if their Affected Claims become Proven Claims, receive the distributions, all as set forth in the Claims Procedure Order and this Plan.

All amounts recognized as Voting Claims or Proven Claims shall be net of any amount that the Company is entitled to offset, recoup, compensate or otherwise apply in reduction of such amounts.

4.3. <u>Creditors' Meeting</u>

The Creditors' Meeting shall be held in accordance with this Plan, the Meeting Procedure Order, and any further Order which may be made from time to time for the purposes of considering and voting on the Resolution or other matters to be considered at the Creditors' Meeting.

4.4. Approval by Affected Creditors

The Company will seek approval of the Plan by the affirmative vote of the Required Majorities. The Resolution to be voted on at the Creditors' Meeting will be decided by the Required Majorities on a vote as set forth in the Meeting Procedure Order. The result

of any vote will be binding on all Affected Creditors, whether or not any such Affected Creditor is present and voting (in person or by proxy) at the Creditors' Meeting.

4.2 Claims Bar Date

4.5. Amendment of Proofs of Claim

Any Affected Creditor who filed its Proof of Claim prior to the Claims Bar Date or who has been permitted to file a late claim pursuant to an Order, shall be entitled to amend its Proof of Claim in order for its Proof of Claim to conform to the provisions of this Plan.

4.6. Late Filing of Claims for Disclaimed Leases

Notwithstanding the Claims Bar Date and its effect or anything contained in this Plan or the Claims Procedure Order, any Disclaimed Landlord in respect of a Disclaimed Lease which may be disclaimed or resiliated between the date of this Plan and the Sanction Date shall be entitled, during (but not after) a period of 30 days following the date of the disclaimer or resiliation of such Lease, to file a Disclaimed Landlord Claim with the Monitor in conformity with Section 2.32.3 hereof. If such Disclaimed Landlord fails to file such Disclaimed Landlord Claim within such 30 day period, such Disclaimed Landlord shall have no right to receive any distribution of the Aggregate Distribution and the Company shall be released from such Disclaimed Landlord Claim and Sections 6.26.2 and 6.36.3 of the Plan shall apply to such Disclaimed Landlord Claim.

4.7. Claims Bar Date

Subject to the provisions of Sections 00 and 4.64.6 hereof, if an Affected Creditor holding an Affected Claim has failed to file its Proof of Claim prior to the Claims Bar Date and has not been permitted to file a late claim pursuant to an Order, that Affected Creditor shall be barred from voting at the Creditors' Meeting, that Affected Creditor shall have no right to receive any distributions of the Aggregate Distribution hereunder, the Company shall be released from the Affected Claims of such Creditor and Section 6.26.2 and 6.36.3 of this Plan shall apply to all such Affected Claims.

SECTION-5

5. PROCEDURE FOR RESOLVING DISTRIBUTIONS IN RESPECT OF DISPUTED CLAIMS

5.1. No Distributions Pending Allowance

Notwithstanding any other provision of this Plan, no distributions of the Aggregate Distribution hereunder shall be made by the Monitor with respect to a Disputed Claim unless and until it has become a Proven Claim. Disputed Claims shall be dealt with in accordance with the Claims Procedure Order and this Plan.

5.2. <u>Distribution From Reserve Once Disputed Claims Resolved</u>

The Monitor shall make allocations from the Reserve to holders of Disputed Claims following any of the Payment Dates in accordance with this Plan. To the extent that Disputed Claims become Proven Claims after any relevant Payment Date, the Monitor shall, from time to time at its sole discretion, distribute from the Reserve to the holders of such Proven Claims, the amount which they would have been entitled to receive in respect of such Proven Claims had such Claims been Proven Claims on any relevant Payment Date. To the extent that any Disputed Claim or a portion thereof has become a Disallowed Claim, then the Monitor shall, at any time which it deems appropriate in the circumstances, distribute, to the holders of Proven Claims, their *pro rata* share from the Reserve of such additional amount kept in the Reserve on account of any such Disallowed Claim.

SECTION 6

6. <u>EFFECT OF THE PLAN AND RELEASES</u>

6.1. Effect of the Plan

Upon the Aggregate Distribution being fully remitted to the Monitor, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Performance"), whereupon all Affected Claims shall be deemed to be fully and finally settled, compromised, released and transacted (as envisaged by the relevant provisions of the *Civil Code of Quebec*), as of the date of issuance of the Certificate of Performance, subject only to an Affected Creditor's right to recover the distributions under this Plan, except for Claims as may have been fully paid prior to such date, in which case those Affected Claims will be deemed to be settled as of such payment.

6.2. Releases upon Implementation

Upon the Implementation of this Plan on the Plan Implementation Date, each of:

- (a) the Company;
- (a) the Monitor and its legal counsel in the CCAA proceedings;
- (b) the Directors and all present and future officers and employees, the Company's legal counsel, financial advisors, consultants and agents of the Company (and their respective directors, officers and employees); and
- (c) the DIP Lender and its legal counsel-in-the CCAA Proceedings,

(collectively the "First Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Person may be entitled to assert whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with the business and affairs of the Company, this Plan and the CCAA Proceedings, and all claims arising out of such actions or omissions shall be forever waived and released, all to the full extent permitted by law.

6.3. Releases after Performance

<u>Upon the Aggregate Distribution being fully remitted to the Monitor and the Monitor's issuance and filing with the Court of the Certificate of Performance, each of:</u>

(a) the Company; and

(b) the Directors and all present and future officers and employees of the Company,

(each a "Second Released Party" and collectively the "Second Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability. obligation, demand or cause of action of whatever nature which any Person (including any Secured Creditor in respect of each Second Released Party, except the Company and solely in respect to its Secured Claim), may be entitled to assert (including any and all Claims in respect of statutory liabilities and any Directors' Claims of all Directors and all present and former officers and employees of the Company and any alleged fiduciary or other duty), whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with Claims, the business and affairs of the Company, this Plan and the CCAA Proceedings, and all claims arising out of such actions or omissions shall be forever waived and released (other than the right to enforce the Company's obligations under the Plan or any related document), all to the full extent permitted by law.

6.3.6.4. Exception to Releases

Nothing set forth in Section 6.2 Sections 6.2 6.2 and 6.3 6.3 hereof shall:

- (a) release or discharge a <u>Second</u> Released Party from an Unaffected Claim;
- (b) release or discharge the Company from or in respect of its obligations under this Plan;
- (c) affect the right of any Person:
 - to recover an indemnity from any insurance coverage under which that Person is an insured; or
 - (ii) to obtain recovery on a claim or liability against a <u>Second Released Party</u> from any insurance coverage pursuant to which that <u>Second Released Party</u> is an insured, but, for certainty, any claim or liability to which an insurer is or would otherwise be subrogated as against the Company is released hereunder and the recovery to which such Person shall be entitled under such insurance coverage shall be limited to the proceeds of insurance actually paid by the insurer with respect to such claim or liability;
- (d) release or discharge the Directors with respect to matters set out in Section 5.1(2) of the CCAA; or
- (e) release or discharge any Person (whether or not a Director, officer or employee of the Company) other than the Company from any obligations arising from any guarantee furnished by such Person to and in favour of any Affected Creditor.

6.4.6.5. Set-Off Rights

Notwithstanding the releases set forth under the Plan, in general, and **Sections 6.1**-and **6.2**6.16.1, 6.26.2 and 6.36.3 hereof, in particular, all Claims shall remain subject to any rights of set-off that otherwise would be available to the Person against whom a Claim is asserted.

6.5.6.6. Injunction Related to Releases

The Sanction Order will enjoin the prosecution by or on behalf of any Person, whether directly, derivatively or otherwise, of any claim, obligation, suit, judgment, damage, demand, debt, right, cause of action; liability or interest released, discharged or terminated pursuant to this Plan.

6.6.6.7. Waiver of Defaults

From and after the Plan Implementation Date, all Persons shall be deemed to have waived any and all defaults or occurrences of the Company then existing or previously committed by the Company-er, caused by the Company or arising, directly or indirectly, or non-compliance with any covenant, positive or negative pledge, warranty, representation, term, provision, condition or obligation, express or implied, in any contract, credit document, agreement for sale, lease or other agreement, written or oral, and any and all amendments or supplements thereto, existing between such Person and the Company arising from the Company's insolvency, the Company's filing a notice of intention to make a proposal under the BIA, the filing by the Company under the CCAA or the transactions contemplated by this Plan or otherwise, and any and all notices of default and demands for payment under any instrument, including any guarantee arising from such default, shall be deemed to have been rescinded.

6.8. SECTION-7Stay of Proceedings

The stay of proceedings set forth and ordered in the Initial Order (except to the extent modified by any other Order) in respect of the Company, the Directors and all other Stayed Parties shall be and remain in full force and effect up until and including the Plan Implementation Date and the Affected Creditors hereby consent to the Court's issuance of an Order to such effect.

7. PROVISIONS GOVERNING DISTRIBUTIONS

7.1. Partial Distributions for Claims Allowed

Except as otherwise provided herein or as ordered by the Court, distributions of the Aggregate Distribution shall be made by the Monitor at the time and in the manner deemed reasonable by the Monitor, and partial distributions may be made prior to receipt of the full amount of the Aggregate Distribution. In such a case, all such partial payments shall represent the *pro rata* amount of the distribution to which the holders of a Proven Claim would otherwise be entitled to receive.

7.2. Currency to be used for the Distribution

For the purposes of determining any Claims denominated in currencies other than Canadian Dollars for distribution purposes, such Claims shall be converted by the Monitor to Canadian dollars at the Bank of Canada noon spot rate of exchange for exchanging currency to Canadian dollars on the Determination Date (which in the case of US Dollars was C\$1.3080 for US\$1.00).

7.3. Assignment of Claims

For purposes of determining entitlement to receive a distribution pursuant to this Plan, the Company and the Monitor and each of their respective agents, successors and

assigns shall have no obligation to recognize any transfer of Claims except as provided for under the Claims Procedure Order or the Meeting Procedure Order.

7.4. Interest on Affected Claims

Except as specifically provided in the Plan or the Sanction Order, interest shall not be treated as accruing on account of any Affected Claims for purposes of determining the allowance and distribution of such Affected Claim. To the extent that any Proven Claim to which a distribution under this Plan relates is comprised of indebtedness and accrued but unpaid interest thereon, such distribution shall, to the extent permitted by applicable Laws, be allocated for tax purposes to the principal amount of such Proven Claim (including any secured and unsecured portion(s) of the principal amount of such Proven Claim) first and then, to the extent that the consideration exceeds the principal amount of the Proven Claim, to the portion of such Proven Claim representing accrued but unpaid interest (including interest in respect of any secured portion of such Proven Claim).

7.5. <u>Distributions by Monitor</u>

The Monitor shall make all distributions required under this Plan in accordance with the provisions of this Plan, in general, and the provisions of **Section 55** and **Section 77** hereof, in particular. The Monitor shall receive, without further Court approval, reasonable compensation from the Company for distribution services rendered pursuant to the Plan.

7.6. <u>Delivery of Distributions</u>

- (a) Subject to Section 7.37-3 hereof, distributions shall be made by the Monitor (i) at the addresses set forth on the Proof of Claim form filed by the Affected Creditors (or at the last known addresses of such Affected Creditors if no Proof of Claim form was filed or if the Company or the Monitor has been notified in writing of a change of address), (ii) at the addresses set forth in any written notice of address change delivered to the Monitor after the date of any related Proof of Claim, or (iii) in a registered retirement savings plan account designated by any employee of the Company;
- (b) If any distribution to a Creditor is returned as undeliverable, no further distributions to such Creditor shall be made unless and until the Monitor is notified of the then-current address of such Creditor, at which time all missed distributions shall be made to such Creditor without interest. The Company shall make reasonable efforts to locate Affected Creditors for which distributions were undeliverable. Any claim for undeliverable distributions must be made on or before the later to occur of (i) 3 months after the Payment Date, or (ii) 3 months after such Creditor's Claim becomes a Proven Claim, after which date all

unclaimed property shall revert to the Company free of any restrictions or claims thereon and the claim of any Creditor with respect to such property shall be discharged and forever barred, notwithstanding any applicable Laws to the contrary.

7.7. No Double Recovery

The aggregate recovery on account of any Proven Claim from all sources, regardless of whether on account of a theory of primary or secondary liability, by reason of guarantee, surety, indemnity, joint and several obligation or otherwise, shall not exceed (i) 100% of the underlying indebtedness, liability or obligation giving rise to such Claim or, (ii) where the underlying indebtedness, liability or obligation giving rise to such Claim is denominated in a currency (the "Original Currency") other than Canadian dollars, 100% of such underlying indebtedness, liability or obligation after conversion of the value of the distributions received in Canadian dollars back to the Original Currency at the Bank of Canada noon spot rate of exchange for exchanging Canadian dollars to the Original Currency on the Determination Date.

7.8. Withholding Requirements

In connection with the Plan, any distribution made hereunder by the Monitor shall be made net of all applicable Taxes. Notwithstanding any other provision of the Plan, each Creditor that is to receive a distribution pursuant to the Plan shall have sole and exclusive responsibility for the satisfaction and payment of any Taxes imposed by any Governmental Authority (including income, withholding and other Taxes on account of such distribution). The Monitor, as necessary, shall be authorized to take any and all actions as may be necessary or appropriate to comply with such withholding and reporting requirements. All amounts withheld on account of Taxes shall be treated for all purposes as having been paid to the Creditor in respect of which such withholding was made, provided such withheld amounts are remitted to the appropriate Governmental Authority.

SECTION-8

8. <u>IMPLEMENTATION OF THE PLAN</u>

8.1. Conditions Precedent to Implementation of Plan

The implementation of this Plan by the Company is subject to the occurrence and/or fulfillment of each of the following conditions precedent (the "Plan Implementation Conditions") on or prior to the December 4518, 2015 or such later date as may be ordered by the Court (the "Plan Implementation Deadline"), namely:

 this Plan shall have been approved by Resolution enacted by the Required Majorities;

- (b) the Sanction Order sanctioning this Plan shall have been issued, and shall either (A) order its provisional execution notwithstanding appeal, or (B) not then be subject to any appeal therefrom, or the Sanction Order shall have been rendered executory notwithstanding any appeal, and the operation and effect of the Sanction Order shall not have been stayed, reversed or amended, and shall, among other things, declare and order that:
 - (i) this Plan, including the compromises and arrangements set out herein, is sanctioned and approved pursuant to Section 6 of the CCAA and, as at the Plan Implementation Date, will be effective and will enure to the benefit of and be binding upon the Company, the Affected Creditors and all other Persons referred to in this Plan or in the Sanction Order;
 - (ii) (A) this Plan has been approved by the Required Majorities of Affected Creditors of the Company in conformity with the CCAA; (B) the Company has complied with the provisions of the CCAA and the Orders of the Court made in the CCAA Proceedings in all respects; (C) the Court is satisfied that the Company has neither done nor purported to do anything that is not authorized by the CCAA; and (D) this Plan and the transactions contemplated thereby are fair and reasonable;
 - (iii) the full and final release and discharge of the Claims will be effective and will enure to the benefit of and be binding upon the Company, the Affected Creditors and all other Persons referred to in this Plan upon the issuance of the Certificate of Performance;
 - (iv) the Company and the Monitor are authorized to take all steps and actions necessary to implement this Plan;
 - all Proven Claims determined in accordance with the Claims Procedure Order and this Plan are final and binding on the Company and all Affected Creditors;
 - (vi) <u>subject to Sections 00 and 4.64.6 hereof</u>, any Claims for which a Proof of Claim has not been filed by the Claims Bar Date shall be forever barred and extinguished;
 - (vii) all distributions and payments by or at the direction of the Monitor, in each case on behalf of the Company, under the Plan are for the account of the Company and the fulfillment of its obligations under the Plan;
 - (viii) the Company and the Monitor may apply to the Court for advice and direction in respect of any matters arising from or under the Plan;

- subject to the performance by the Company of its obligations under this Plan, all contracts, leases, agreements and arrangements to which the Company are a party and that have not been disclaimed, terminated or repudiated pursuant to the Initial Order and/or the CCAA will be and remain in full force and effect, unamended, as at the Plan Implementation Date, and no Person who is a party to any such contract, lease, agreement or other arrangement may accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of dilution or other remedy) or make any demand under or in respect of any such contract, lease, agreement or other arrangement and no automatic termination will have any validity or effect, by reason of:
 - (A) any event that occurred on or prior to the Plan Implementation Date and is not continuing that would have entitled such Person to enforce those rights or remedies (including defaults, events of default, or termination events arising as a result of the insolvency of the Company);
 - (B) the insolvency of the Company, the fact that the Company filed a notice of intention to make a proposal under the BIA or the fact that the Company sought or obtained relief under the CCAA; or
 - (C) any compromises or arrangements effected pursuant to this Plan or any action taken or transaction effected pursuant to this Plan;
- (x) the releases set forth in **Section** <u>6.26.2</u> hereof and the injunction set forth in **Section** <u>6.56.66.6</u> hereof are implemented and in effect; and
- (xi) the commencement or prosecution, whether directly, derivatively or otherwise, or any demands, claims, actions, causes of action, counterclaims, suits or any indebtedness, liability, obligation or cause of action released and discharged pursuant to this Plan are enjoined; and
- (xii) the stay of proceedings as set forth in Section 6.8 hereof is and remains in full force and effect up until and including the Plan Implementation Date; and
- (c) the Replacement Loan Financing shall have been disbursed and each of the DIP Claim, the DIP Lender Charge, the Salus Claim and the Salus Security shall have been discharged and released occurred.

8.2. Waiver of Conditions

Except for the condition set forth at Section 8.1(c)8.1(e) hereof, each of the Plan Implementation Conditions enure solely to the Company's benefit and, except for each of the conditions set forth in Sections 8.1 (a) and 8.1 (b)(i)8.1(a)8.1(a) and 8.1(b)(i)8.1(b)(i) hereof, may be waived in whole or in part either (i) solely by the Company (and no other Person and without the consent of any other Person), without any other notice to parties in interest or the Court and without a hearing, or (ii) by the Court. The failure to satisfy or waive any of the Plan Implementation Conditions prior to the Plan Implementation Date may be asserted by the Company regardless of the circumstances giving rise to the failure of such Plan Implementation Conditions to be satisfied (including any action or inaction by the Company). The failure of the Company to exercise any of the foregoing rights shall not be deemed a waiver of any other rights, and each such right shall be deemed an on-going right that may be asserted at any time. Notwithstanding the foregoing, the condition set forth at Section 8.1(c)8.1(e) hereof may only be waived in writing by the Company and the DIP Lender.

8.3. Certificate of Implementation

Upon the occurrence and/or fulfillment of all of the Plan Implementation Conditions (or waiver thereof, as the case may be) on or prior to the Plan Implementation Deadline, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Implementation").

8.4. Effect of Plan

Upon the issuance of the Certificate of Performance, the settlement of the Affected Claims in accordance with the Plan shall become final and binding on the Company, and all the Affected Creditors and their respective successors and assigns, and this Plan shall, upon the Aggregate Distribution being fully remitted to the Monitor and the Monitor's issuance and filing with this Court of the Certificate of Performance, result in the full and final settlement of all Affected Claims, including for greater certainty, any Restructuring Claim and any Claims resulting, directly or indirectly, from the consequences and effects relating to the acceptance of the Plan by the Affected Creditors, its sanction by the Court, or its implementation, and any indebtedness. obligations or undertakings which the Company may subsequently become subject to. directly or indirectly, as a result of an obligation, transaction or an event that occurred before the Determination Date, as well as any indebtedness, obligations or any undertakings that the Company may be subject to on any date whatsoever in connection with the Plan, the approval thereof by the Court or the implementation thereof. For greater certainty, this Plan shall not affect or impair any rights, remedies and recourses which the Company had, has or may have after the issuance of the Certificate of Performance in connection with transactions, facts or obligations existing prior to the Determination Date:

8.5. Nullity of Plan

In the event that all of the Plan Implementation Conditions have not occurred and/or been fulfilled (or waived, as the case may be) on or prior to the Plan Implementation Deadline, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Non-Implementation"). Upon the issuance of such Certificate of Non-Implementation, this Planany settlement of the Affected Claims shall automatically become null, void and of no effect whatsoever and shall remain owing by the Company and neither the Company, the Affected Creditors, the Monitor nor any other Person affected by this Plan shall be bound, obliged or affected by any of the provisions of this Plan.

SECTION 9

9. MISCELLANEOUS

9.1. Confirmation of Plan

Upon occurrence and/or fulfillment of all of the Plan Implementation Conditions (or waiver thereof, as the case may be) on or prior to the Plan Implementation Deadline, this Plan shall be binding upon the Company, the Affected Creditors and any and all other Persons affected by the provisions of this Plan as well as each of their respective successors and assigns.

9.2. Paramountcy

From and after the Plan Implementation Date, any conflict between this Plan and/or the covenants, warranties, representations, terms, conditions, provisions or obligations, express or implied, of any contract, mortgage, security agreement, and/or indenture, trust indenture, loan agreement, commitment letter, agreement for sale, the by-laws of the Company, lease or other agreement, undertaking or any other source of obligations, written or oral and any and all amendments or supplements thereto existing between one or more of the Affected Creditors and the Company as at the Plan Implementation Date will be deemed to be governed by the terms, conditions and provisions of this Plan and the Sanction Order, which shall take precedence and priority. For greater certainty, all Affected Creditors shall be deemed irrevocably for all purposes to consent to all transactions contemplated in and by this Plan.

9.3. Modification of Plan

The Company:

(a) in consultation with the Monitor and the DIP Lender (if any DIP Claim remains outstanding), reserves the right to file any modification of, or amendment or supplement to, this Plan by way of supplementary plan or plans of reorganization, compromise or arrangement (or any one or more thereof) at or before the Creditors' Meeting, in which case any such supplementary plan or plans of reorganization, compromise or arrangement (or an one or more thereof), shall, for all purposes, be and be deemed to form part of and be incorporated into this Plan. The Company shall file any supplementary plans with the Court as soon as practicable. The Company shall give notice to Affected Creditors of the details of any such modification, amendment or supplement at the Creditors' Meeting prior to the vote being taken to approve this Plan. The Company may give notice of a proposed modification, amendment or supplement to this Plan at or before the Creditors' Meeting by notice which shall be sufficient if given to those Affected Creditors present at such meeting in person or by proxy; and/or

(b) after the Creditors' Meeting (and both prior to and subsequent to the obtaining of the Sanction Order), the Company, in consultation with the Monitor and, if prior to the Plan Implementation Date, the DIP Lender (if any DIP Claim then remains), at any time and from time to time vary, amend, modify or supplement this Plan, (except to reduce the amount of the Aggregate Distribution or extend any Payment Dates other than as expressly provided herein), without the need for obtaining an Order of the Court or providing notice to the Affected Creditors if the Monitor determines that such variation, amendment, modification or supplement would not be materially prejudicial to the interests of the Affected Creditors under this Plan or the Sanction Order and is necessary in order to give effect to the substance of this Plan or the Sanction Order. All of the foregoing shall not require any further vote by or approval by the Affected Creditors or any approval by the Court.

9.4. Deeming Provisions

In this Plan, any deeming provisions are not rebuttable and are conclusive and irrevocable.

9.5. Sections 38 and 95 to 101 BIA

Notwithstanding Section 36.1 of the *CCAA*, Section 38 and Sections 95 through 101 of the *BIA* shall not apply to this Plan. Accordingly, neither the Monitor, any Creditor nor any other Person shall be entitled to exercise any right, remedy or recourse, or to commence any action, motion, application or any other proceeding against any member of the Fisher Group, any Creditor or any other Person in relation to the Company, based on Section 38 and Sections 95 through 101 of the *BIA*.

9.6. Responsibilities of the Monitor

The Monitor is acting in its capacity as Monitor in the CCAA Proceedings with respect to the Company and not in its personal or corporate capacity and will not be responsible or liable for any responsibilities or obligations of the Company under this Plan or otherwise, including with respect to the making of distributions or the receipt of any distribution by any Creditor or any other Person pursuant to the Plan. The Monitor will have the powers and protections granted to it by this Plan, the *CCAA*, the Initial Order, the Claims Procedure Order, the Meeting Procedure Order, the Sanction Order and any other Orders.

9.7. Liability Limitations

The Monitor, the Company and their respective legal counsel and other professional advisors, shall not be liable to any Person for any act or omission in connection with, or arising out of, the CCAA Proceedings, the Plan, the pursuit of sanctioning of the Plan, the consummation and implementation of the Plan or the administration of the Plan or the funds to be distributed under the Plan, except for their own wilful misconduct or gross negligence.

9.8. Notices

- (a) Any notices or communication to be made or given hereunder to the Company or the Monitor shall be in writing and shall refer to this Plan and may, subject as hereinafter provided, be made or given by personal delivery, by courier, by prepaid mail, by fax or by e-mail, addressed to the respective parties as follows:
 - (i) if to the Company:

LAURA'S SHOPPE (P.V.) INC.

2955 Jules-Brillant Laval, Quebec H7P 6B2

Attention:

Kalman Fisher and

Josh Fisher

E-Mails:

kfisher@laura.ca

iosh.fisher@laura.ca

Fax:

(450) 973-6099

with copy to:

KUGLER KANDESTIN LLP

1 Place Ville Marie Suite 2101 Montreal, Quebec H3B 2C6 Attention: Gerald F. Kandestin,

David Stolow and Jeremy Cuttler

E-Mail:

gkandestin@kklex.com

dstolow@kklex.com jcuttler@kklex.com

Fax:

(514) 875-8424

(ii) if to the Monitor:

KPMG LLP

KPMG Tower 600 De Maisonneuve Blvd., West Suite 1500 Montreal, Quebec H3A 0A3

Attention:

Dev Coossa

E-Mail:

dcoossa@kpmg.ca

Fax:

(514) 840-2121

with copy to:

FISHMAN FLANZ MELAND PAQUIN, L.L.P.

1250 Rene-Levesque Blvd., West Suite 4100 Montreal, Quebec H3B 4W8

Attention:

Avram Fishman and

Mark Meland

E-Mails:

afishman@ffmp.ca

mmeland@ffmp.ca

Fax:

(514) 932-4170

or to such other address as any party may from time to time notify the others in accordance with this Section 9.89.8. In the event of any strike, lock-out or other event which interrupts postal service in any part of Canada, all notices and communications during such interruption may only be given or made by personal delivery, fax or e-mail and any notice or other communication given or made by prepaid mail within the 5 Business Day period immediately preceding the commencement of such interruption, unless actually received, shall be deemed not to have been given or made. All such notices and communications shall be deemed to have been received, in the case of notice by delivery, fax or e-mail prior to 5:00 p.m. (local time) on a Business Day, when received or if received

after 5:00 p.m. (local time) on a Business Day or at any time on a non-Business Day, on the next following Business Day and, in the case of notice mailed as aforesaid, on the fourth Business Day following the date on which such notice or other communication is mailed. The unintentional failure by the Company or the Monitor to give any notice contemplated hereunder to any particular Creditor shall not invalidate this Plan or any action taken by any Person pursuant to this Plan.

(b) Any notices or communication to be made or given hereunder by the Monitor or the Company to a Creditor may be sent by fax, e-mail, ordinary mail, registered mail, courier or facsimile transmission to the e-mail address, address or fax number specified by such Creditor in its Proof of Claim, or in any subsequent written notice of change of address given to the Monitor. A Creditor shall be deemed to have received any document sent pursuant to this Plan four (4) Business Days after the document is sent by ordinary or registered mail and on the Business Day immediately following the day on which the document is sent by courier, fax or e-mail.

9.9. Severability of Plan Provisions

If, prior to the Sanction Date, any term or provision of this Plan is held by the Court to be invalid, void or unenforceable, the Court, at the request of the Company which request shall be made in consultation with the Monitor and the DIP Lender (if any DIP Claim then remains), shall have the power to either:

- (a) sever such term or provision from the balance of this Plan and provide the Company with the option to proceed with the implementation of the balance of this Plan as of and with effect from the Plan Implementation Date; or
- (b) alter and interpret such term or provision to make it valid and enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpreted.

Notwithstanding any such severing, holding, alteration or interpretation, and provided the Company proceeds with the implementation of this Plan, the remainder of the terms and provisions of this Plan shall remain in full force and effect and shall in no way be affected, impaired or invalidated by such severing, holding, alteration or interpretation.

9.10. Revocation, Withdrawal or Non-Consummation

The Company, upon consultation with the Monitor and the DIP Lender, reserves the right to revoke or withdraw this Plan at any time prior to the Sanction Date and to file

subsequent plans of arrangement and compromise. If the Company revokes or withdraws this Plan, or if the Sanction Order is not issued:

- (a) this Plan shall be null, void and inoperative in all respects;
- (b) any Claim, any settlement or compromise embodied in this Plan (including the fixing or limiting of any Claim to an amount certain), assumption or termination by this Plan, and any document or agreement executed pursuant to this Plan shall be deemed null, void and inoperative; and
- (c) nothing contained in this Plan, and no act taken in preparation for consummation of this Plan, shall:
 - constitute or be deemed to constitute a waiver or release of any Claims by or against the Company or any other Person;
 - (ii) prejudice in any manner the rights of the Company or any Person in any further proceedings involving the Company; or
 - (iii) constitute an admission of any sort by the Company or any other Person.

9.11. Further Assurance

Notwithstanding that the transactions and events set out in this Plan shall occur and be deemed to occur in the order set out herein without any additional act or formality, each of the Persons affected hereby shall make, do and execute, or cause to be made, done and executed at the cost of the requesting party, all such further acts, deeds, agreements, transfers, assurances, instruments or documents as may reasonably be required by the Company in order to better implement this Plan.

9.12. Governing Law

This Plan shall be governed by and construed in accordance with the law of the Province of Quebec and the federal laws of Canada applicable therein. Any questions as to the interpretation or application of this Plan and all proceedings taken in connection with this Plan and its provisions shall be subject to the exclusive jurisdiction of the Court.

9.13. Successors and Assigns

This Plan shall be binding upon and shall enure to the benefit of the heirs, administrators, executors, legal personal representatives, successors and permitted assigns of the Company, the Directors, the Affected Creditors or any other Persons affected by or benefiting from the provisions of this Plan.

9.14. French Language Version of Plan

In the event of any discrepancy between any of the provisions of the English language version of this Plan and any French language version thereof, the provisions of the English version of this Plan shall, under all circumstances, prevail and govern.

9.15. Choice of Language

The Company and each of the hereafter enumerated members of the Fisher Group acknowledges that it (or he) has required that this Plan and all related documents be prepared, in English. La Compagnie et chacun des membres du "Fisher Group" ci-après énumerés reconnaît avoir exigé que ce Plan et tous les documents connexes soient rédigés en anglais.

(signature page to follow)

MONTREAL, Province of Quebec, this	day of, 2015.
	LAURA'S SHOPPE (P.V.) INC. Per:
	Kalman Fisher, President
	e Fisher Group acknowledges haven taken
Section 3.113.11 hereof.	Plan and consents to the provisions of
	Plan and consents to the provisions of KALMAN FISHER
	KALMAN FISHER 3482731 CANADA INC.
	KALMAN FISHER 3482731 CANADA INC. Per:

A-N-N-E-X (TO-PLAN-OF-COMPROMISE AND ARRANGEMENT OF LAURA'S SHOPPE (P.V.) INC.)

DISCLAIMED LEASES

		LANDLORD	STORE/OFFICE-LOCATION	STORE BANNER (if applicable)
	1	Galloway-REIT (Halifax) Ltd. (Smartcentres)	Bayers Lake Center 208 Chain Lake Drive Halifax, Nova Scotia B3S 1C5	Laura
	2.—	The Boulevard Shopping Centre (Montreal) Limited Partnership	The Boulevard Shopping Centre 4270 Jean Talon Street East Montreal, Quebec H1S-1J7	Laura
	3.—	Calloway-Reit (Cambridge) Inc. (Smartcentres)	Cambridge Centre 28 Pinebush Road Cambridge, Ontario	Laura
	4	Riocan Holdings (Quebec) Inc.	Mega-Gentre Notre-Dame Notre-Dame and Autoroute 13 Laval, Quebec H7X-3V5	Laura
	5. -	3945333 Manitoba Limited (Smartcentres)	SmartCentres Winnipeg Southwest 1665 Kenaston Blvd. Unit P3 Winnipeg, Manitoba R3P 2M4	. Laura
	6	Riocan Property Services Trust as agent for 2113362 Ontario Limited, as to an undivided one-half interest, and Trinity Properties Alberta Limited as to an undivided one-half interest	Beacon-Hill-Centre 11834-Sarcee Trail NW Calgary, Alberta T3R-0A1	Laura
***************************************	7	Beauward-Shopping	Galeries Joliette	Laura

		LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
,		Centres Ltd.	1075 Firestone Blvd. Joliette, Quebec J6E-6X6	
	8	Montez Core Income Fund Limited Partnership and Hillcrest Property Holdings Inc.	Hillcrest Mall 9350 Yonge Street Richmond Hill, Ontario L4C 5G2	Laura
***************************************	9,	Carrefour Richelieu Realties Ltd. (Westcliff)	Carrefour Angrignon 7077 Newman Blvd. LaSalle, Quebec H8N 1X1	Laura Plus
	10.	Omers-Realty Management-Corporation, by its manager-OPGI Management GP Inc. (as general partner of the OPGI Management-Limited Partnership)	Square One Shopping Centre 100 City Centre Drive Mississauga, Ontario L5B 2C9	Laura Plus
delinida estima esperante proprio por la companidad del companidad	11.	Park Royal Shopping Centre Holdings Ltd.	Park Royal Shopping Centre 2002 Park Royal South West Vancouver, BC V7T 2W4	Melanie Lyne
APP PER PER PER PER PER PER PER PER PER P	12.	Oxford-Properties Group Inc. and CPP Investment Board Real Estate Holdings Inc.	Edmonton City Centre West 10020-102 Avenue Edmonton, Alberta T5J-4B7	Melanie Lyne
	13.	Dartmouth Crossing Limited	Dartmouth-Crossing 34 Logicalmond-Close Dartmouth, Nova Scotia B3B-0C8	Melanie-Lyne
	14.	RRL Burloak Inc. and 2121049 Ontario Limited, by its agent RioCan Property Services Inc. (Trustee for	RioCan Centre Burloak 3543 Wyecroft Road Oakville, Ontario	Melanie Lyne

		LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
		RioCan Real Estate Investment Trust)		
	15. —	Riotrin Properties (Orleans) Inc.	Trinity Crossing 4210 Innes Road Orleans, Ontario K4A-5E6	Melanie Lyne
	16.	RioCan Holdings (TJV Bromont) Inc. and- 1633272 Alberta ULC	Carrefour Champêtre Bromont 105 boul. le Carrefour Bromont, Quebec J2L 3L1	Laura
	17.	1451945 Ontario Limited and Sun Life Assurance Company of Canada	Mayfield Common 288 Mayfield Common Northwest Edmonton, Alberta T5P-4B3	Laura
	18	RioCan Meadows Shopping Centre Inc. and- 2118717 Ontario Limited	RioCan Meadows 3918 - 17 St. NW Edmonton, Alberta T6T-0C2	Laura
	19.	Riotrin Properties Inc.	RioCan Signal Hill Center 5715 Signal Hill Center SW Calgary, Alberta T3H 3P8	Laura
***************************************	20.	713949 Ontario Limited	St. Laurent-Shopping Centre 1200 St. Laurent-Blvd. Ottawa, Ontario K1K-3B8	Laura
***************************************	21	Montez Core Income Fund Limited Partnership and- Hillcrest Property Holdings Inc.	Hillcrest Mall 9350 Yonge Street Richmond Hill, Ontario L4C-5G2	Melanie Lyne
***************************************	22.	Les Factoeries Saint- Sauveur (2003) Inc.	Les Factoreries Saint-Sauveur 160 chemin du Lac-Millette Saint-Sauveur, Quebec JOR-1R6	Melanie Lyne

	LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
23.	9191-3723 Quebec Inc.	2955 rue Jules-Brillant Laval, Quebec H7P 6B2	N/A
24	Desjardins Financial Security Life Assurance Company	151 City Centre Drive Suites 603/701 Mississauga, Ontario L5B 1M7	AWA
25. –	3608727 Canada Inc.	2519-2525 boul. Le Corbusier Laval, Quebec H7S-1Z4	N/A

EXHIBIT P-5

(Compared version between the Amended Plan and the Re-Amended Plan)

COMPARED: AMENDED TO RE-AMENDED

CANADA

PROVINCE OF QUEBEC DISTRICT OF MONTREAL

NO.: 500-11-049256-155

SUPERIOR COURT (Commercial Division)

(Sitting as a court designated pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. c. C-36)

IN THE MATTER OF THE PLAN OF COMPROMISE AND ARRANGEMENT OF:

MAGASIN LAURA (P.V.) INC. / LAURA'S SHOPPE (P.V.) INC.

Debtor/Petitioner

-and-

BOUTIQUE LAURA CANADA LTÉE / LAURA'S SHOPPE CANADA LTD.

-and-

3482731 CANADA INC.

-and-

9318-5494 QUEBEC INC.

-and-

KALMAN FISHER

Stayed Parties

-and-

KPMG INC.

Monitor

RE-AMENDED PLAN OF COMPROMISE AND ARRANGEMENT Under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36

TABLE OF CONTENTS

1.	INTERPRETATION	
1.1.		
1.2.	Interpretation	9
1.3.	Date For Any Action	. 10
1.4.	Statutory References	. 10
<u>2.</u>	COMPROMISE AND ARRANGEMENT	10
2.1.	Persons Affected	. 10
2.2.	Classes of Affected Claims	. 10
2.3.	Disclaimed Landlord Claims	. 10
2.4.	Unaffected Claims	. 11
2.5.	Treatment of Affected Claims	. 12
2.6.	Settlement of Affected Claims	. 13
2.7.	Extension of Payment Dates	. 13
2.8.	Prepayment Rights	. 13
2.9.	No Acknowledgment or Admission	13
3.	TREATMENT OF OTHER CLAIMS AND MATTERS	. 13
3.1.	Treatment of the DIP Claim	13
3.2.	Treatment of Excluded Claims	14
3.3.	Treatment of Administration Claims	14
3.4.	Treatment of Gift Card Claims	14
3.5.	Treatment of Certain Crown Priority Claims	14
3.6.	Treatment of Section 6(5) Claims	14
3.7.	Treatment of Section 19(2) Claims	14
3.8.	Treatment of KERP Claims	14
3.9.	Treatment of Secured Claims	14
3.10	Treatment of Replacement Financing Claim	15
3.11	Treatment of Fisher Group Claims	15
4.	VALUATION OF CLAIMS, CREDITORS' MEETING AND RELATED MATTERS	
4.1.	Conversion of Affected Claims into Canadian Currency	15
4.2.	Affected Claims	15
43	Creditors' Meeting	16

4.4.	Approval by Affected Creditors	16
4.5.	Amendment of Proofs of Claim	16
4.6.	Late Filing of Claims for Disclaimed Leases	. 16
4.7.	Claims Bar Date	. 16
<u>5.</u>	PROCEDURE FOR RESOLVING DISTRIBUTIONS IN RESPECT OF DISPUTED CLAIMS	17
5.1.	No Distributions Pending Allowance	. 17
5.2.	Distribution From Reserve Once Disputed Claims Resolved	. 17
<u>6.</u>	EFFECT OF THE PLAN AND RELEASES	<u> 17</u>
6.1.	Effect of the Plan	. 17
6.2.	Releases upon Implementation	. 17
6.3.	Releases after Performance	. 18
6.4.	Exception to Releases	. 19
6.5.	Set-Off Rights	. 19
6.6.	Injunction Related to Releases	. 19
6.7.	Waiver of Defaults	. 20
6.8.	Stay of Proceedings	. 20
<u>7. </u>	PROVISIONS GOVERNING DISTRIBUTIONS	20
7.1.	Partial Distributions for Claims Allowed	. 20
7.2.	Currency to be used for the Distribution	20
7.3.	Assignment of Claims	21
7.4.	Interest on Affected Claims	21
7.5.	Distributions by Monitor	21
7.6.	Delivery of Distributions	21
7.7.	No Double Recovery	22
7.8.	Withholding Requirements	22
8.	IMPLEMENTATION OF THE PLAN	. 22
8.1.	Conditions Precedent to Implementation of Plan	22
8.2.	Waiver of Conditions	25
8.3.	Certificate of Implementation	25
8.4.	Effect of Plan	25
8.5.	Nullity of Plan	
9.	MISCELLANEOUS	. 26
9.1.	Confirmation of Plan	

9.2.	Paramountcy	26
9.3.	Modification of Plan	26
9.4.	Deeming Provisions	27
9.5.	Sections 38 and 95 to 101 BIA	27
9.6.	Responsibilities of the Monitor	27
9.7.	Liability Limitations	28
9.8.	Notices	28
9.9.	Severability of Plan Provisions	30
9.10.	Revocation, Withdrawal or Non-Consummation	30
9.11.	Further Assurance	31
9.12.	Governing Law	31
9.13.	Successors and Assigns	31
9.14.	French Language Version of Plan	32
9.15.	Choice of Language	32

1. <u>INTERPRETATION</u>

1.1. Definitions

In this Plan, unless otherwise stated or the context otherwise requires:

"Administration Charge" has the meaning ascribed to such term in the Initial Order;

"Administration Claim" means a claim or any other indebtedness or obligation secured by the Administration Charge;

"Affected Claim" means any Claim other than an Unaffected Claim. For greater certainty, Affected Claims include Disclaimed Landlord Claims, any Claim of Her Majesty the Queen in right of Canada or any province (other than Crown Priority Claims), Directors Claims and Restructuring Claims;

"Affected Creditor" means a Creditor holding an Affected Claim, but only to the extent of its Affected Claim;

"Affected Creditors Class" has the meaning set forth in Section 2.2 hereof;

"Aggregate Distribution" has the meaning set forth in Section 2.5 hereof;

"BIA" means the Bankruptcy and Insolvency Act, R.S.C., 1985, c. B-3, as amended;

"Business Day" means a day, other than a Saturday, a Sunday, or a non-juridical day (as defined in article 6 of the Quebec Code of Civil Procedure, R.S.Q., c. C-25, as amended);

"CCAA" means the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended;

"CCAA Charge(s)" means the Administration Charge, the DIP Lender Charge and the KERP Charge;

"CCAA Proceedings" means proceedings in respect of the Company before the Court commenced, taken up and continued under the CCAA;

"Certificate of Implementation" has the meaning set forth in Section 8.3 hereof;

"Certificate of Performance" has the meaning set forth in Section 6.1 hereof;

"Certificate of Non-Implementation" has the meaning set forth in Section 8.5 hereof;

"Chair" means a representative of the Monitor, or any other individual designated by the Monitor to preside as chairperson at the Creditors' Meeting;

"Claim" means any right of any Person against the Company in connection with any indebtedness, liability or obligation of any kind of the Company owed to such Person and any interest, or penalties accrued thereon or costs payable in respect thereof, whether liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, unsecured, present, future, known or unknown, by guarantee, surety or otherwise, and whether or not such right is executory or anticipatory in nature, including the right or ability of any Person to advance a claim for contribution or indemnity or otherwise with respect to any matter, action, cause or chose in action, whether existing at present or commenced in the future, which indebtedness, liability or obligation is based in whole or in part on facts existing, or transactions which occurred, prior to the Determination Date, or which would have been claims provable in bankruptcy had the Company become bankrupt on the Determination Date, and, without limitation, including any Restructuring Claim, but excluding any Unaffected Claim and "Claims" means all of them:

"Claims Bar Date" has the meaning as set forth in the Claims Procedure Order;

"Claims Procedure Order" means the "Claims Procedure Order" issued by the Court on September 11, 2015, establishing, among other things, procedures for proving Claims, as amended or supplemented from time to time by further Order(s) of the Court;

"Company", "Debtor" or "Petitioner" means Magasin Laura (P.V.) Inc./Laura's Shoppe (P.V.) Inc.;

"Court" means the Commercial Division of the Quebec Superior Court for the District of Montreal;

"Creditor(s)" means any Person(s) having a Claim and may, where the context requires, include the assignee of a Claim or a trustee, interim receiver, receiver, receiver and manager, or other Person acting on behalf of such Person(s);

"Creditors' Meeting" means the meeting of Affected Creditors to be convened for the purposes of voting on the Plan, or any adjournment of such meeting;

"Crown Priority Claims" means any Claims of Her Majesty the Queen in right of Canada or in right of any province as described in Section 6(3) or Section 38(2) of the CCAA. For greater certainty, any Claim of Her Majesty the Queen in right of Canada or in right of any Province other than Crown Priority Claims shall be an Affected Claim hereunder;

"Determination Date" means July 31, 2015, being the date on which the Company commenced proceedings under the *BIA* by filing a notice of intention to make a proposal thereunder, which proceedings were subsequently taken up and continued by the *CCAA* Proceedings under the *CCAA*;

"DIP Claim" means any and all obligations of the Company to the DIP Lender pursuant to the DIP Facility Documents or the Order for New Financing as of or after the Plan Implementation Date;

"DIP Facility Documents" means the "committed term sheet" dated October 22, 2015 by and among, *inter alia*, the Company and the DIP Lender (as amended, modified, restated and/or supplemented from time to time) together with any related collateral, loan or security documents executed in connection therewith or which relate thereto;

"DIP Lender" means collectively, Third Eye Capital Corporation, acting as agent and hypothecary representative for and on behalf of certain affiliates and funds it manages as well as all of the "New Financing Parties" as defined in the Order for New Financing;

"DIP Lender Charge" means the "New Lender Charge" as set forth in the Order for New Financing;

"DIP Loan" means the loan(s) made and/or to be made by the DIP Lender as envisaged by the Order for New Financing;

"Directors" means all of the Company's past and present directors as well as any Persons who were or are deemed to be directors of the Company pursuant to any applicable Laws;

"Directors' Claims" means all Claims, of any nature or source whatsoever, of any of the Creditors against any or all of the Directors which arose before the Determination Date and which relate to obligations of the Company where such Directors are by law liable in their capacity as directors for payment of such obligations. Director's Liability shall, however, exclude all Claims which (i) relate to contractual rights of one or more Creditors arising from contracts with one or more Directors, including any guarantees given by any Directors relating to the Company's obligations, (ii) seek injunctive relieve against any of the Directors in relation to the Company, or (iii) are based on allegations of misrepresentation made by any Directors to Creditors or of wrongful or oppressive conduct by Directors;

"Disallowed Claim" means any Claim, or that portion thereof which has been revised or disallowed by the Monitor pursuant to the Claims Procedure Order or any other Order in respect of which an appeal by the Creditor has been dismissed or all appeal periods, if any, have expired;

"Disclaimed Leases" means each and every one of the Leases which (i) were disclaimed or resiliated by the Company pursuant to Section 65.2 of the BIA (as recognized by the Initial Order) or Section 32 of the CCAA, as exhaustively enumerated in the annex forming part of this Plan, and (ii) may be disclaimed or resiliated by the Company at any time between the date of this Amended Plan of Compromise and ArrangementPlan and the Sanction Date, and "Disclaimed Lease" means each of them;

"Disclaimed Landlord" means the landlord under any Disclaimed Lease and "Disclaimed Landlords" means all of them;

"Disclaimed Landlord Claims" means all Proven Claims of the Disclaimed Landlords for losses incurred by Disclaimed Landlords in relation to Disclaimed Leases, as provided for in Section 2.3 hereof and "Disclaimed Landlord Claim" means any of them;

"Disputed Claim" means a Claim or that portion thereof that is the object of a Notice of Revision or Disallowance and, in either case, has not become a Proven Claim or a Disallowed Claim;

"Excluded Claim" means any right of any person against the Company in connection with any indebtedness, liability or obligation of any kind which arose with respect to transactions which occurred after the Determination Date and any interest thereon, including the Administration Claim, the DIP Claim, a KERP Claim, the Replacement Financing Claim and any obligation of the Company towards creditors who have supplied or shall supply services, utilities, goods or materials or who have or shall have advanced funds to the Company after the Determination Date, but only to the extent of their claims in respect of the supply of such services, utilities, goods, materials or funds after the Determination Date and to the extent that such claims are not otherwise affected by the Plan. For greater certainty, a Restructuring Claim is not an Excluded Claim;

"Excluded Creditor" means a Person having a claim in respect of an Excluded Claim but only in respect of such Excluded Claim and to the extent that the Plan does not otherwise affect such claim;

"First Released Parties" has the meaning as set forth in Section 6.2 hereof;

"Fisher Group" means Kalman Fisher, 3482731 Canada Inc., 9318-5494 Quebec Inc. (formerly 7735235 Canada Inc.) and any Person who/which is, or was at any time within the 12 consecutive months immediately preceding the Determination Date, a Related Person to Kalman Fisher;

"Fisher Group Claims" means any Claims for payment of principal on any loans or advances made to the Company (or its predecessor) at any time prior to the

Determination Date by any member of the Fisher Group, which includes any Secured Claim by any member of the Fisher Group, but excludes any Claims or future claims by any member of the Fisher Group for (i) salary (or other remuneration), (ii) reimbursement of business expenses, or (iii) payment of principal on any loans or advances in lieu of salary (or other remuneration) or in lieu of interest;

"Gift Card Claim" has the meaning set forth in Section 2.4(a)(v) hereof;

"Governmental Authority" means any (i) multinational, national, provincial, state, regional, municipal, local or other government, governmental or public department, ministry, central bank, court, tribunal, arbitral body, commission, board, official, minister, bureau or agency, domestic or foreign, (ii) subdivision, agent, commission, board or authority of any of the foregoing; or (iii) quasi-governmental or private body, including any tribunal, commission, regulatory agency or self-regulatory organization, exercising any regulatory, expropriation or taxing authority under, or for the account of, any of the foregoing;

"Initial Order" means the "Initial Order" issued by the Court on August 12, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"KERP Charge" has the meaning ascribed to such term in the Initial Order;

"KERP Claim" means a claim or any other indebtedness or obligation secured by the KERP Charge and "KERP Claims" means all of them:

"Laws" means all laws, statutes, codes, ordinances, decrees, rules, regulations, by-laws, judicial or arbitral or administrative or ministerial or departmental or regulatory judgments, injunctions, orders or decisions of any Governmental Authority, statutory body or self-regulatory authority, including general principles of law having the force of law and the term "applicable" with respect to such Laws and, in the context that refers to any Person, means such Laws as are applicable to such Person or its business, undertaking, property or securities and emanate from a Governmental Authority or self-regulatory authority having jurisdiction over the Person or its business, undertaking, property or securities;

"Lease" means any lease and/or other agreement for the occupancy of real and immovable property as well as all amendments thereto and renewals thereof and "Leases" means all of them;

"Meeting Date" means the date fixed for the Creditors' Meeting in accordance with the Meeting Procedure Order, or any subsequent Order, or any subsequent date following any adjournment of that meeting, as the case may be;

"Meeting Procedure Order" means the "Meeting Procedure Order" issued by the Court on October 23, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"Monitor" means KPMG Inc., in its capacity as Monitor, as appointed by the Court pursuant to the Initial Order;

"Notice of Revision or Disallowance" has the meaning as set forth in the Claims Procedure Order;

"Order" means any order of the Court in the CCAA Proceedings;

"Order for New Financing" means the "Order for New Financing" issued by the Court on October 30, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"Original Currency" has the meaning set forth in Section 7.7 hereof;

"Payment Date" and "Payment Dates" have the meaning set forth in Section 2.5 hereof;

"Person" means any individual, corporation, limited or unlimited liability company, general or limited partnership, association, trust, trustee, executor, administrator, legal personal representative, estate, unincorporated organization, joint venture, governmental body or agency, or any other entity;

"Plan" means the present "Re-amended Plan of Compromise and Arrangement" of the Company pursuant to the provisions of the CCAA, as may be amended, varied or supplemented by the Company from time to time in accordance with its terms;

"Plan Implementation Conditions" has the meaning set forth in Section 8.1 hereof;

"Plan Implementation Date" means the date, on or prior to the Plan Implementation Deadline, upon which all of the Plan Implementation Conditions have occurred or have been satisfied or waived;

"Plan Implementation Deadline" has the meaning set forth in Section 8.1 hereof;

"Proof of Claim" has the meaning set forth in the Claims Procedure Order:

"Proven Claim" means, in respect of a Creditor, the amount of the Claim of such Creditor as finally determined for distribution purposes in accordance with the provisions of this Plan, the CCAA and the Claims Procedure Order and "Proven Claims" means all of them;

"Related Persons" has the meaning set forth in Section 4 of the BIA;

"Replacement Financing" means the new senior credit facilities to be provided to the Company by the DIP Lender, including a term loan facility and a revolving asset-based loan facility on terms and conditions acceptable to the DIP Lender and the Company, which shall completely repay and/or replace the DIP Loan;

"Replacement Financing Claim" means any and all indebtedness and obligations of the Company and the DIP Lender under the Replacement Financing;

"Required Majorities" means the affirmative vote of a majority in number of the Affected Creditors voting in the Affected Creditors Class, having Voting Claims and voting on the Plan (in person or by proxy) at the Creditors' Meeting and representing not less than 66-2/3% in value of the Voting Claims of the Affected Creditors voting (in person or by proxy) at the Creditors' Meeting;

"Reserve(s)" means the reserve(s) to be established and maintained under this Plan by the Monitor by holding, on account of Disputed Claims, an amount equal to the aggregate amount that the holders of Disputed Claims would be entitled to receive if all such Disputed Claims had been Proven Claims as of any Payment Date;

"Resolution" means, collectively, one or any of, the resolutions providing for the approval of the Plan by the Affected Creditors;

"Restructuring Claim" means any right or claim of any Person against the Company in connection with any indebtedness or obligation of any kind owed to such Person arising out of the Company's disclaimer—er, resiliation or renegotiation (at any time after the Determination Date) of any Leases (excluding the Disclaimed Landlord Claims, which shall be dealt with pursuant to Section 2.3 hereof), agreements or other contracts or the Company's restructuring under the CCAA, and includes any right or claim resulting, directly or indirectly, from the consequences and effects of the Plan's acceptance by the Affected Creditors, the Plan's sanction by the Sanction Order, the Plan's implementation and any debt forgiveness resulting from any of the foregoing; provided, however, that a Restructuring Claim shall not include an Excluded Claim. For greater certainty, a Restructuring Claim is an Affected Claim;

"Sanction Date" means the date on which the Sanction Order is issued:

"Sanction Order" means the Order to be made under the CCAA sanctioning the Plan, as such Order may be affirmed, amended or modified by the Court at any time prior to the Plan Implementation Date or, if appealed, then, unless such appeal is withdrawn, abandoned or denied, as affirmed or amended on appeal, in form and content which is satisfactory to the Petitioner acting reasonably;

"Second Released Party" and "Second Released Parties" have the meanings as set forth in Section 6.3 hereof;

"Section 6(5) Claim" means any claim of an employee or former employee described in Section 6(5) of the CCAA but only to the extent of such amounts as required to be paid under the CCAA;

"Section 19(2) Claims" means any claim described in Section 19(2) of the CCAA;

"Secured Claim" means the Claim of a Secured Creditor, to the extent of the value of such Secured Creditor's security;

"Secured Creditor" has the meaning set forth in the CCAA, but only to the extent that such Creditor's mortgage, hypothec, pledge, charge, lien, privilege, security interest or other rights over the property of the Company was valid, opposable, perfected, and could be set up against third parties, including a trustee to the Company's bankruptcy, on both the Determination Date and the Claims Bar Date, failing which that Secured Creditor will be deemed to be an "unsecured creditor", as defined in the CCAA;

"Stayed Parties" means Boutique Laura Canada Ltée/Laura's Shoppe Canada Ltd., 3482731 Canada Inc., 9318-5494 Quebec Inc. (formerly 7735235 Canada Inc.) and Kalman Fisher;

"Taxes" means all federal, state, provincial, territorial, county, municipal, local or foreign taxes, duties, imposts, levies, assessments, tariffs and other charges imposed, assessed or collected by a Governmental Authority, including (i) any gross income, net income, gross receipts, business, royalty, capital, capital gains, goods and services, value added, severance, stamp, franchise, occupation, premium, capital stock, sales and use, real property, land transfer, personal property, ad valorem, transfer, licence, profits, windfall profits, environmental, payroll, employment, employer health, pension plan, anti-dumping, countervail, excise, severance, stamp, occupation, or premium tax, (ii) all withholdings on amounts paid to or by the relevant Person, (iii) all employment insurance premiums, Canada, Quebec and any other pension plan contributions or premiums, (iv) any fine, penalty, interest, or addition to tax, (v) any tax imposed, assessed, or collected or payable pursuant to any tax-sharing agreement or any other contract relating to the sharing or payment of any such tax, levy, assessment, tariff, duty, deficiency, or fee, and (vi) any liability for any of the foregoing as a transferee, successor, guarantor, or by contract or by operation of law;

"Unaffected Claim" means any right of any Person in connection with any indebtedness, liability or obligation of any kind of the type described in Section 2.4(a) hereof. For greater certainty, the Unaffected Claims include the Excluded Claims;

"Unaffected Creditor" means a Person having a Claim in respect of an Unaffected Claim, but only in respect of such Unaffected Claim, and for greater certainty, includes an Excluded Creditor;

"Voting Claim" means, in respect of an Affected Creditor, the amount of such Affected Creditor's claim which has been accepted for voting purposes in accordance with the provisions of this Plan, the Claims Procedure Order and the CCAA and "Voting Claims" means all of them;

1.2. Interpretation

For purposes of this Plan:

- (a) any reference in this Plan to a contract, instrument, release, indenture, agreement or other document being in a particular form or on particular terms and conditions means that such document shall be substantially in such form or substantially on such terms and conditions;
- (b) any reference in this Plan to an existing document or exhibit filed or to be filed means such document or exhibit as it may have been or may be amended, modified or supplemented;
- (c) all references to currency and to "\$", "C\$" or "Dollars" are to Canadian dollars except as otherwise indicated;
- (d) all references in this Plan to Sections are references to Sections of this Plan;
- (e) unless otherwise specified, the words "hereof", "herein" and "hereto" refer to this Plan in its entirety rather than to any particular portion of this Plan;
- (f) the division of this Plan into Sections and paragraphs and the insertion of captions and headings to Sections and paragraphs are for convenience of reference only and are not intended to affect the interpretation of, or to be part of this Plan;
- (g) where the context requires, a word or words importing the singular shall include the plural and vice versa and a word or words importing one gender shall include all genders;
- (h) the words "includes" and "including" are not limiting; and
- (i) the word "or" is not exclusive.

1.3. Date For Any Action

In the event that any date (including any Payment Date) on which any action (including any payment) is required to be taken under this Plan by any of the parties is not a Business Day, that action (including any payment) shall be required to be taken on the next succeeding day which is a Business Day.

1.4. Statutory References

Any reference in this Plan to a statute includes all regulations made thereunder and all amendments to such statute or regulations in force, from time to time, or any statute or regulations that supplement or supersede such statute or regulations.

2. <u>COMPROMISE AND ARRANGEMENT</u>

2.1. Persons Affected

The purpose of this Plan is to provide for the full and final settlement of all Affected Claims. Except as specifically provided for in this Plan, this Plan will become effective on the Plan Implementation Date in accordance with its terms, and all Affected Claims against the Company will be fully and finally, settled, compromised and released in this Plan upon the Aggregate Distribution being fully paid to the Monitor, it being understood that any Affected Claim that is paid in full pursuant to this Plan before such date shall be settled, compromised and released on the date of such payment. This Plan shall be binding on and enure to the benefit of the Company, the Affected Creditors, the First Released Parties, the Second Released Parties, any trustee, agent or other Person acting on behalf of any Affected Creditor and such other Persons who have received the benefit of, or are bound by any waivers, releases or indemnities hereunder.

2.2. Classes of Affected Claims

There shall only be one class of Affected Creditors for the purpose of voting on, and receiving distributions pursuant to this Plan, being the "Affected Creditors Class".

2.3. <u>Disclaimed Landlord Claims</u>

Disclaimed Landlord Claims shall be Affected Claims under this Plan and shall be limited to an amount in respect of each Disclaimed Lease equal to the lesser of determined in accordance with Section 32(7) CCAA.

(a) the aggregate of:

- (i) the rent provided for in the Disclaimed Lease for the first year of such Disclaimed Lease following the date on which the disclaimer or resiliation becomes effective; and
- (ii) 15% of the rent for the remainder of the term of the Disclaimed-Lease after that year.

or;

(b) three years' rent provided for in the Disclaimed Lease.

2.4. <u>Unaffected Claims</u>

- (a) This Plan does not affect the following claims (each, an "Unaffected Claim" and, collectively, the "Unaffected Claims"), the holders of which will not be entitled to vote at the Creditors' Meeting or receive any distributions under this Plan, namely:
 - (i) the Excluded Claims;
 - (ii) any Administration Claim;
 - (iii) any DIP Claim;
 - (iv) any KERP Claims;
 - (v) any claim with respect to gift-cards, gift certificates, lay-away deposits and other customer certificates (collectively, "Gift Card Claims");
 - (vi) Crown Priority Claims;
 - (vii) Section 6(5) Claims;
 - (viii) Section 19(2) Claims;
 - (ix) Secured Claims;
 - (x) the Replacement Financing Claim; and
 - (xi) Fisher Group Claims,

all of which will be dealt with in Section 3 hereof;

(b) Nothing in this Plan shall affect the Company's rights and defences, both legal and equitable, with respect to any Unaffected Claim including any rights arising under or pursuant to the Claims Procedure Order or this Plan or any rights with respect to legal and equitable defences or entitlements to set-offs or recoupments against such Unaffected Claims.

2.5. <u>Treatment of Affected Claims</u>

By no later than each of the payment dates hereafter set forth (collectively the "Payment Dates" or individually a "Payment Date"), the aggregate sum of \$4 Million (the "Aggregate Distribution"), without any interest whatsoever thereon, shall be paid by the Company to the Monitor and shall thereafter be distributed by the Monitor to the Affected Creditors, according to the amount of their respective Proven Claims as hereafter set forth, namely:

- (a) by no later than April 30, 2016, \$1 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor as follows:
 - (i) an amount equal to the lesser of (A) the Proven Claim of each Affected Creditor, or (B) \$1,250, shall be distributed by the Monitor to the Affected Creditors, according to the amount of their Proven Claims; and
 - (ii) an amount equal to the difference between (A) \$1 Million, and (B) the aggregate amount to be distributed by the Monitor pursuant to Section 2.5(a)(i) above, shall be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the amounts set forth in Section 2.5(a)(i) above;
- (b) by no later than July 31, 2016, \$1 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distribution under Section 2.5(a) hereof;
- (c) by no later than December 15, 2016, \$1.8 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distributions under Sections 2.5(a) and 2.5(b) hereof; and
- (d) by no later than February 15, 2017, \$200,000 shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distributions under Sections 2.5(a), 2.5(b) and 2.5(c) hereof.

2.6. Settlement of Affected Claims

The payment of the entire Aggregate Distribution to the Monitor as set forth in Section 2 hereof shall constitute the full and final payment, reduction, settlement and transaction (as envisaged by the relevant provisions of the Civil Code of Quebec) of all of the Affected Claims of all of the Affected Creditors and engage the releases and other provisions set forth in this Plan.

2.7. Extension of Payment Dates

Any or all of the Payment Dates may be extended by the Company with the consent of the Monitor and the DIP Lender (if any portion of the DIP Loan or the Replacement Financing then remains outstanding), by additional periods not exceeding, in each case, 60 days. On application by the Company or the Monitor (in either case, with the consent of the DIP Lender, if any portion of the DIP Loan or the Replacement Financing remains outstanding) made prior to a Payment Date or, if applicable, prior to the expiry of any extension thereof as described above, the Court may issue Orders extending (or further extending) any or all of the Payment Dates for any additional period(s) deemed appropriate by the Court in the then existing circumstances.

2.8. Prepayment Rights

The Company (with the consent of the DIP Lender, if any portion of the DIP Loan or the Replacement Financing then remains outstanding) shall, at all times prior to the Final Distribution Date, be entitled to pre-pay the whole or any amount of the Aggregate Distribution prior to any Payment Date, whereupon the amount(s) of any such pre-payment(s) shall be distributed by the Monitor to the Affected Creditors in accordance with the provisions of **Section 2.5** hereof, *mutatis mutandis*.

2.9. No Acknowledgment or Admission

Nothing contained in this Plan shall constitute, under any circumstances whatsoever, an acknowledgment or admission by the Company as to the existence or validity of any Claims or Restructuring Claims, in general, and as to the existence or validity of any Restructuring Claims in respect of disclaimed or resiliated Leases (other than the Disclaimed Leases), in particular.

3. TREATMENT OF OTHER CLAIMS AND MATTERS

3.1. Treatment of the DIP Claim

On or before the earlier of (i) the Plan Implementation Date, or (ii) the date on which the DIP Claim becomes payable pursuant to the provisions of the DIP Facility Documents, the DIP Lender shall receive full payment of any remaining DIP Claim.

3.2. <u>Treatment of Excluded Claims</u>

Subject to Section 3.1 hereof, the Excluded Claims (other than the Administration Claim, the DIP Claim and any KERP Claim) will remain in full force and effect in accordance with their terms after the Plan Implementation Date, and will be paid in full by the Company in the normal course of its business as and when they become due.

3.3. <u>Treatment of Administration Claims</u>

All Administration Claims, if any, will be paid in full by the Company as and when they become due, and any remaining balance will be settled as soon as practicable after the Plan Implementation Date.

3.4. Treatment of Gift Card Claims

Gift Card Claims will be honored in accordance with the terms of the relevant gift card, gift certificate, lay-away deposit or other customer certificates upon presentation of such gift card, gift certificate, lay-away deposit or other customer certificates by the holder at any of the Petitioner's retail stores.

3.5. <u>Treatment of Certain Crown Priority Claims</u>

All Crown Priority Claims described in Section 6(3) of the CCAA, if any, will be paid in full by the Petitioner within 6 months immediately following the Sanction Date.

3.6. <u>Treatment of Section 6(5) Claims</u>

All Section 6(5) Claims, if any, will be paid in such amounts as required under the CCAA immediately after the Sanction Date.

3.7. Treatment of Section 19(2) Claims

All Section 19(2) Claims, if any, will be paid by the Company as and when they become due, unaffected by this Plan.

3.8. Treatment of KERP Claims

Each KERP Claim will be paid by the Company as and when it becomes due.

3.9. <u>Treatment of Secured Claims</u>

Secured Claims (other than the Administration Claim, the DIP Claim, the Replacement Financing Claim and KERP Claims) will be dealt with as provided for under agreements

entered into or as may be hereafter entered into between the Company and the relevant Secured Creditor, or as may be ordered by the Court. For greater certainty, Secured Claims will be unaffected by this Plan.

3.10. Treatment of Replacement Financing Claim

The Replacement Financing Claim shall be paid by the Company to the DIP Lender in accordance with the provisions, terms and conditions of the Replacement Financing.

3.11. Treatment of Fisher Group Claims

Conditional upon the enactment of a Resolution approving the Plan, issuance of the Sanction Order and implementation of the Plan as set forth in **Section 8** hereof, each member of the Fisher Group:

- (a) subordinates and postpones all of the Fisher Group Claims to and in favour of full payment of the Aggregate Distribution to the Affected Creditors, such that no member of the Fisher Group shall be entitled to receive any payment from the Company of any principal on any of the Fisher Group Claims unless and until the full Aggregate Distribution shall have been paid to the Monitor as set forth in the Plan; and
- (b) waives and renounces to any right to be an Affected Creditor.

4. VALUATION OF CLAIMS, CREDITORS' MEETING AND RELATED MATTERS

4.1. Conversion of Affected Claims into Canadian Currency

For the purposes of determining the value of Affected Claims denominated in currencies other than Canadian dollars for voting purposes, such Affected Claims shall be converted by the Monitor to Canadian dollars at the Bank of Canada noon spot rate of exchange for exchanging currency to Canadian dollars on the Determination Date (which in the case of US Dollars was C\$1.3080 for US\$1.00).

4.2. Affected Claims

Affected Creditors shall be entitled to prove their respective Affected Claims, vote their Voting Claims in respect of the Plan, and, if their Affected Claims become Proven Claims, receive the distributions, all as set forth in the Claims Procedure Order and this Plan.

All amounts recognized as Voting Claims or Proven Claims shall be net of any amount that the Company is entitled to offset, recoup, compensate or otherwise apply in reduction of such amounts.

4.3. Creditors' Meeting

The Creditors' Meeting shall be held in accordance with this Plan, the Meeting Procedure Order, and any further Order which may be made from time to time for the purposes of considering and voting on the Resolution or other matters to be considered at the Creditors' Meeting.

4.4. Approval by Affected Creditors

The Company will seek approval of the Plan by the affirmative vote of the Required Majorities. The Resolution to be voted on at the Creditors' Meeting will be decided by the Required Majorities on a vote as set forth in the Meeting Procedure Order. The result of any vote will be binding on all Affected Creditors, whether or not any such Affected Creditor is present and voting (in person or by proxy) at the Creditors' Meeting.

4.5. Amendment of Proofs of Claim

Any Affected Creditor who filed its Proof of Claim prior to the Claims Bar Date or who has been permitted to file a late claim pursuant to an Order, shall be entitled to amend its Proof of Claim in order for its Proof of Claim to conform to the provisions of this Plan.

4.6. <u>Late Filing of Claims for Disclaimed Leases</u>

Notwithstanding the Claims Bar Date and its effect or anything contained in this Plan or the Claims Procedure Order, any Disclaimed Landlord in respect of a Disclaimed Lease which may be disclaimed or resiliated between the date of this Amended Plan of Gempromise and Arrangement Plan and the Sanction Date shall be entitled, during (but not after) a period of 30 days following the date of the disclaimer or resiliation of such Lease, to file a Disclaimed Landlord Claim with the Monitor in conformity with Section 2.3 hereof. If such Disclaimed Landlord fails to file such Disclaimed Landlord Claim within such 30 day period, such Disclaimed Landlord shall have no right to receive any distribution of the Aggregate Distribution and the Company shall be released from such Disclaimed Landlord Claim and Sections 6.2 and 6.3 of the Plan shall apply to such Disclaimed Landlord Claim.

4.7. Claims Bar Date

Subject to the provisions of **Sections 4.5** and **4.6** hereof, if an Affected Creditor holding an Affected Claim has failed to file its Proof of Claim prior to the Claims Bar Date and has not been permitted to file a late claim pursuant to an Order, that Affected Creditor shall be barred from voting at the Creditors' Meeting, that Affected Creditor shall have no right to receive any distributions of the Aggregate Distribution hereunder, the Company shall be released from the Affected Claims of such Creditor and **Sections 6.2** and **6.3** of this Plan shall apply to all such Affected Claims.

5. PROCEDURE FOR RESOLVING DISTRIBUTIONS IN RESPECT OF DISPUTED CLAIMS

5.1. No Distributions Pending Allowance

Notwithstanding any other provision of this Plan, no distributions of the Aggregate Distribution hereunder shall be made by the Monitor with respect to a Disputed Claim unless and until it has become a Proven Claim. Disputed Claims shall be dealt with in accordance with the Claims Procedure Order and this Plan.

5.2. <u>Distribution From Reserve Once Disputed Claims Resolved</u>

The Monitor shall make allocations from the Reserve to holders of Disputed Claims following any of the Payment Dates in accordance with this Plan. To the extent that Disputed Claims become Proven Claims after any relevant Payment Date, the Monitor shall, from time to time at its sole discretion, distribute from the Reserve to the holders of such Proven Claims, the amount which they would have been entitled to receive in respect of such Proven Claims had such Claims been Proven Claims on any relevant Payment Date. To the extent that any Disputed Claim or a portion thereof has become a Disallowed Claim, then the Monitor shall, at any time which it deems appropriate in the circumstances, distribute, to the holders of Proven Claims, their *pro rata* share from the Reserve of such additional amount kept in the Reserve on account of any such Disallowed Claim.

6. <u>EFFECT OF THE PLAN AND RELEASES</u>

6.1. Effect of the Plan

Upon the Aggregate Distribution being fully remitted to the Monitor, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Performance"), whereupon all Affected Claims shall be deemed to be fully and finally settled, compromised, released and transacted (as envisaged by the relevant provisions of the Civil Code of Quebec), as of the date of issuance of the Certificate of Performance, subject only to an Affected Creditor's right to recover the distributions under this Plan, except for Claims as may have been fully paid prior to such date, in which case those Affected Claims will be deemed to be settled as of such payment.

6.2. Releases upon Implementation

Upon the Implementation of this Plan on the Plan Implementation Date, each of:

(a) the Monitor and its legal counsel in the CCAA proceedings;

- (b) the Company's legal counsel, financial advisors, consultants and agents of the Company (and their respective directors, officers and employees); and
- (c) the DIP Lender and its legal counsel,

(collectively the "First Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Person may be entitled to assert whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with the business and affairs of the Company, this Plan and the CCAA Proceedings, and all claims arising out of such actions or omissions shall be forever waived and released, all to the full extent permitted by law.

6.3. Releases after Performance

Upon the Aggregate Distribution being fully remitted to the Monitor and the Monitor's issuance and filing with the Court of the Certificate of Performance, each of:

- (a) the Company; and
- (b) the Directors and all present and future officers and employees of the Company,

(each a "Second Released Party" and collectively the "Second Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability. obligation, demand or cause of action of whatever nature which any Person (including any Secured Creditor in respect of each Second Released Party, except the Company and solely in respect to its Secured Claim), may be entitled to assert (including any and all Claims in respect of statutory liabilities and any Directors' Claims of all Directors and all present and former officers and employees of the Company and any alleged fiduciary or other duty), whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with Claims, the business and affairs of the Company, this Plan and the CCAA Proceedings, and all claims arising out of such actions or omissions shall be forever waived and released (other than the right to

enforce the Company's obligations under the Plan or any related document), all to the full extent permitted by law.

6.4. Exception to Releases

Nothing set forth in Sections 6.2 and 6.3 hereof shall:

- (a) release or discharge a Second Released Party from an Unaffected Claim:
- release or discharge the Company from or in respect of its obligations under this Plan;
- (c) affect the right of any Person:
 - (i) to recover an indemnity from any insurance coverage under which that Person is an insured; or
 - (ii) to obtain recovery on a claim or liability against a Second Released Party from any insurance coverage pursuant to which that Second Released Party is an insured, but, for certainty, any claim or liability to which an insurer is or would otherwise be subrogated as against the Company is released hereunder and the recovery to which such Person shall be entitled under such insurance coverage shall be limited to the proceeds of insurance actually paid by the insurer with respect to such claim or liability;
- (d) release or discharge the Directors with respect to matters set out in Section 5.1
 (2) of the CCAA; or
- (e) release or discharge any Person (whether or not a Director, officer or employee of the Company) other than the Company from any obligations arising from any guarantee furnished by such Person to and in favour of any Affected Creditor.

6.5. <u>Set-Off Rights</u>

Notwithstanding the releases set forth under the Plan, in general, and Sections 6.1, 6.2 and 6.3 hereof, in particular, all Claims shall remain subject to any rights of set-off that otherwise would be available to the Person against whom a Claim is asserted.

6.6. <u>Injunction Related to Releases</u>

The Sanction Order will enjoin the prosecution by or on behalf of any Person, whether directly, derivatively or otherwise, of any claim, obligation, suit, judgment, damage,

demand, debt, right, cause of action, liability or interest released, discharged or terminated pursuant to this Plan.

6.7. Waiver of Defaults

From and after the Plan Implementation Date, all Persons shall be deemed to have waived any and all defaults or occurrences of the Company then existing or previously committed by the Company, caused by the Company or arising, directly or indirectly, or non-compliance with any covenant, positive or negative pledge, warranty, representation, term, provision, condition or obligation, express or implied, in any contract, credit document, agreement for sale, lease or other agreement, written or oral, and any and all amendments or supplements thereto, existing between such Person and the Company arising from the Company's insolvency, the Company's filing a notice of intention to make a proposal under the *BIA*, the filing by the Company under the *CCAA* or the transactions contemplated by this Plan or otherwise, and any and all notices of default and demands for payment under any instrument, including any guarantee arising from such default, shall be deemed to have been rescinded.

6.8. Stay of Proceedings

The stay of proceedings set forth and ordered in the Initial Order (except to the extent modified by any other Order) in respect of the Company, the Directors and all other Stayed Parties shall be and remain in full force and effect up until and including the Plan Implementation Date and the Affected Creditors hereby consent to the Court's issuance of an Order to such effect.

7. PROVISIONS GOVERNING DISTRIBUTIONS

7.1. Partial Distributions for Claims Allowed

Except as otherwise provided herein or as ordered by the Court, distributions of the Aggregate Distribution shall be made by the Monitor at the time and in the manner deemed reasonable by the Monitor, and partial distributions may be made prior to receipt of the full amount of the Aggregate Distribution. In such a case, all such partial payments shall represent the *pro rata* amount of the distribution to which the holders of a Proven Claim would otherwise be entitled to receive.

7.2. Currency to be used for the Distribution

For the purposes of determining any Claims denominated in currencies other than Canadian Dollars for distribution purposes, such Claims shall be converted by the Monitor to Canadian dollars at the Bank of Canada noon spot rate of exchange for exchanging currency to Canadian dollars on the Determination Date (which in the case of US Dollars was C\$1.3080 for US\$1.00).

7.3. Assignment of Claims

For purposes of determining entitlement to receive a distribution pursuant to this Plan, the Company and the Monitor and each of their respective agents, successors and assigns shall have no obligation to recognize any transfer of Claims except as provided for under the Claims Procedure Order or the Meeting Procedure Order.

7.4. Interest on Affected Claims

Except as specifically provided in the Plan or the Sanction Order, interest shall not be treated as accruing on account of any Affected Claims for purposes of determining the allowance and distribution of such Affected Claim. To the extent that any Proven Claim to which a distribution under this Plan relates is comprised of indebtedness and accrued but unpaid interest thereon, such distribution shall, to the extent permitted by applicable Laws, be allocated for tax purposes to the principal amount of such Proven Claim (including any secured and unsecured portion(s) of the principal amount of such Proven Claim) first and then, to the extent that the consideration exceeds the principal amount of the Proven Claim, to the portion of such Proven Claim representing accrued but unpaid interest (including interest in respect of any secured portion of such Proven Claim).

7.5. <u>Distributions by Monitor</u>

The Monitor shall make all distributions required under this Plan in accordance with the provisions of this Plan, in general, and the provisions of **Section 5** and **Section 7** hereof, in particular. The Monitor shall receive, without further Court approval, reasonable compensation from the Company for distribution services rendered pursuant to the Plan.

7.6. Delivery of Distributions

- (a) Subject to Section 7.3 hereof, distributions shall be made by the Monitor (i) at the addresses set forth on the Proof of Claim form filed by the Affected Creditors (or at the last known addresses of such Affected Creditors if no Proof of Claim form was filed or if the Company or the Monitor has been notified in writing of a change of address), (ii) at the addresses set forth in any written notice of address change delivered to the Monitor after the date of any related Proof of Claim, or (iii) in a registered retirement savings plan account designated by any employee of the Company;
- (b) If any distribution to a Creditor is returned as undeliverable, no further distributions to such Creditor shall be made unless and until the Monitor is notified of the then-current address of such Creditor, at which time all missed distributions shall be made to such Creditor without interest. The Company shall make reasonable efforts to locate Affected Creditors for which distributions were undeliverable. Any claim for undeliverable distributions must be made on or

before the later to occur of (i) 3 months after the Payment Date, or (ii) 3 months after such Creditor's Claim becomes a Proven Claim, after which date all unclaimed property shall revert to the Company free of any restrictions or claims thereon and the claim of any Creditor with respect to such property shall be discharged and forever barred, notwithstanding any applicable Laws to the contrary.

7.7. No Double Recovery

The aggregate recovery on account of any Proven Claim from all sources, regardless of whether on account of a theory of primary or secondary liability, by reason of guarantee, surety, indemnity, joint and several obligation or otherwise, shall not exceed (i) 100% of the underlying indebtedness, liability or obligation giving rise to such Claim or, (ii) where the underlying indebtedness, liability or obligation giving rise to such Claim is denominated in a currency (the "Original Currency") other than Canadian dollars, 100% of such underlying indebtedness, liability or obligation after conversion of the value of the distributions received in Canadian dollars back to the Original Currency at the Bank of Canada noon spot rate of exchange for exchanging Canadian dollars to the Original Currency on the Determination Date.

7.8. Withholding Requirements

In connection with the Plan, any distribution made hereunder by the Monitor shall be made net of all applicable Taxes. Notwithstanding any other provision of the Plan, each Creditor that is to receive a distribution pursuant to the Plan shall have sole and exclusive responsibility for the satisfaction and payment of any Taxes imposed by any Governmental Authority (including income, withholding and other Taxes on account of such distribution). The Monitor, as necessary, shall be authorized to take any and all actions as may be necessary or appropriate to comply with such withholding and reporting requirements. All amounts withheld on account of Taxes shall be treated for all purposes as having been paid to the Creditor in respect of which such withholding was made, provided such withheld amounts are remitted to the appropriate Governmental Authority.

8. <u>IMPLEMENTATION OF THE PLAN</u>

8.1. Conditions Precedent to Implementation of Plan

The implementation of this Plan by the Company is subject to the occurrence and/or fulfillment of each of the following conditions precedent (the "Plan Implementation Conditions") on or prior to December 18, 2015 or such later date as may be ordered by the Court (the "Plan Implementation Deadline"), namely:

- (a) this Plan shall have been approved by Resolution enacted by the Required Majorities;
- (b) the Sanction Order sanctioning this Plan shall have been issued, and shall either (A) order its provisional execution notwithstanding appeal, or (B) not then be subject to any appeal therefrom, and the operation and effect of the Sanction Order shall not have been stayed, reversed or amended, and shall, among other things, declare and order that:
 - (i) this Plan, including the compromises and arrangements set out herein, is sanctioned and approved pursuant to Section 6 of the CCAA and, as at the Plan Implementation Date, will be effective and will enure to the benefit of and be binding upon the Company, the Affected Creditors and all other Persons referred to in this Plan or in the Sanction Order:
 - (ii) (A) this Plan has been approved by the Required Majorities of Affected Creditors of the Company in conformity with the CCAA; (B) the Company has complied with the provisions of the CCAA and the Orders of the Court made in the CCAA Proceedings in all respects; (C) the Court is satisfied that the Company has neither done nor purported to do anything that is not authorized by the CCAA; and (D) this Plan and the transactions contemplated thereby are fair and reasonable;
 - (iii) the full and final release and discharge of the Claims will be effective and will enure to the benefit of and be binding upon the Company, the Affected Creditors and all other Persons referred to in this Plan upon the issuance of the Certificate of Performance;
 - (iv) the Company and the Monitor are authorized to take all steps and actions necessary to implement this Plan;
 - all Proven Claims determined in accordance with the Claims Procedure Order and this Plan are final and binding on the Company and all Affected Creditors;
 - (vi) subject to Sections 4.5 and 4.6 hereof, any Claims for which a Proof of Claim has not been filed by the Claims Bar Date shall be forever barred and extinguished;
 - (vii) all distributions and payments by or at the direction of the Monitor, in each case on behalf of the Company, under the Plan are for the account of the Company and the fulfillment of its obligations under the Plan;

- (viii) the Company and the Monitor may apply to the Court for advice and direction in respect of any matters arising from or under the Plan;
- subject to the performance by the Company of its obligations under this Plan, all contracts, leases, agreements and arrangements to which the Company are a party and that have not been disclaimed, terminated or repudiated pursuant to the Initial Order and/or the CCAA will be and remain in full force and effect, unamended, as at the Plan Implementation Date, and no Person who is a party to any such contract, lease, agreement or other arrangement may accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of dilution or other remedy) or make any demand under or in respect of any such contract, lease, agreement or other arrangement and no automatic termination will have any validity or effect, by reason of:
 - (A) any event that occurred on or prior to the Plan Implementation Date and is not continuing that would have entitled such Person to enforce those rights or remedies (including defaults, events of default, or termination events arising as a result of the insolvency of the Company);
 - (B) the insolvency of the Company, the fact that the Company filed a notice of intention to make a proposal under the BIA or the fact that the Company sought or obtained relief under the CCAA; or
 - (C) any compromises or arrangements effected pursuant to this Plan or any action taken or transaction effected pursuant to this Plan;
- (x) the releases set forth in Section 6.2 hereof and the injunction set forth in Section 6.6 hereof are implemented and in effect;
- (xi) the commencement or prosecution, whether directly, derivatively or otherwise, or any demands, claims, actions, causes of action, counterclaims, suits or any indebtedness, liability, obligation or cause of action released and discharged pursuant to this Plan are enjoined; and
- (xii) the stay of proceedings as set forth in Section 6.8 hereof is and remains in full force and effect up until and including the Plan Implementation Date; and
- (c) the Replacement Financing shall have occurred.

8.2. Waiver of Conditions

Except for the condition set forth at **Section 8.1(c)** hereof, each of the Plan Implementation Conditions enure solely to the Company's benefit and, except for each of the conditions set forth in **Sections 8.1(a)** and **8.1(b)(i)** hereof, may be waived in whole or in part either (i) solely by the Company (and no other Person and without the consent of any other Person), without any other notice to parties in interest or the Court and without a hearing, or (ii) by the Court. The failure to satisfy or waive any of the Plan Implementation Conditions prior to the Plan Implementation Date may be asserted by the Company regardless of the circumstances giving rise to the failure of such Plan Implementation Conditions to be satisfied (including any action or inaction by the Company). The failure of the Company to exercise any of the foregoing rights shall not be deemed a waiver of any other rights, and each such right shall be deemed an ongoing right that may be asserted at any time. Notwithstanding the foregoing, the condition set forth at **Section 8.1(c)** hereof may only be waived in writing by the Company and the DIP Lender.

8.3. <u>Certificate of Implementation</u>

Upon the occurrence and/or fulfillment of all of the Plan Implementation Conditions (or waiver thereof, as the case may be) on or prior to the Plan Implementation Deadline, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Implementation").

8.4. Effect of Plan

Upon the issuance of the Certificate of Performance, the settlement of the Affected Claims in accordance with the Plan shall become final and binding on the Company, and all the Affected Creditors and their respective successors and assigns, and this Plan shall, upon the Aggregate Distribution being fully remitted to the Monitor and the Monitor's issuance and filing with this Court of the Certificate of Performance, result in the full and final settlement of all Affected Claims, including for greater certainty, any Restructuring Claim and any Claims resulting, directly or indirectly, from the consequences and effects relating to the acceptance of the Plan by the Affected Creditors, its sanction by the Court, or its implementation, and any indebtedness. obligations or undertakings which the Company may subsequently become subject to, directly or indirectly, as a result of an obligation, transaction or an event that occurred before the Determination Date, as well as any indebtedness, obligations or any undertakings that the Company may be subject to on any date whatsoever in connection with the Plan, the approval thereof by the Court or the implementation thereof. For greater certainty, this Plan shall not affect or impair any rights, remedies and recourses which the Company had, has or may have after the issuance of the Certificate of Performance in connection with transactions, facts or obligations existing prior to the Determination Date.

8.5. Nullity of Plan

In the event that all of the Plan Implementation Conditions have not occurred and/or been fulfilled (or waived, as the case may be) on or prior to the Plan Implementation Deadline, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Non-Implementation"). Upon the issuance of such Certificate of Non-Implementation, any settlement of the Affected Claims shall automatically become null, void and of no effect whatsoever and shall remain owing by the Company and neither the Company, the Affected Creditors, the Monitor nor any other Person affected by this Plan shall be bound, obliged or affected by any of the provisions of this Plan.

9. MISCELLANEOUS

9.1. Confirmation of Plan

Upon occurrence and/or fulfillment of all of the Plan Implementation Conditions (or waiver thereof, as the case may be) on or prior to the Plan Implementation Deadline, this Plan shall be binding upon the Company, the Affected Creditors and any and all other Persons affected by the provisions of this Plan as well as each of their respective successors and assigns.

9.2. Paramountcy

From and after the Plan Implementation Date, any conflict between this Plan and/or the covenants, warranties, representations, terms, conditions, provisions or obligations, express or implied, of any contract, mortgage, security agreement, and/or indenture, trust indenture, loan agreement, commitment letter, agreement for sale, the by-laws of the Company, lease or other agreement, undertaking or any other source of obligations, written or oral and any and all amendments or supplements thereto existing between one or more of the Affected Creditors and the Company as at the Plan Implementation Date will be deemed to be governed by the terms, conditions and provisions of this Plan and the Sanction Order, which shall take precedence and priority. For greater certainty, all Affected Creditors shall be deemed irrevocably for all purposes to consent to all transactions contemplated in and by this Plan.

9.3. Modification of Plan

The Company:

(a) in consultation with the Monitor and the DIP Lender (if any DIP Claim remains outstanding), reserves the right to file any modification of, or amendment or supplement to, this Plan by way of supplementary plan or plans of reorganization, compromise or arrangement (or any one or more thereof) at or before the Creditors' Meeting, in which case any such supplementary plan or plans of reorganization, compromise or arrangement (or an one or more thereof), shall, for all purposes, be and be deemed to form part of and be incorporated into this Plan. The Company shall file any supplementary plans with the Court as soon as practicable. The Company shall give notice to Affected Creditors of the details of any such modification, amendment or supplement at the Creditors' Meeting prior to the vote being taken to approve this Plan. The Company may give notice of a proposed modification, amendment or supplement to this Plan at or before the Creditors' Meeting by notice which shall be sufficient if given to those Affected Creditors present at such meeting in person or by proxy; and/or

(b) after the Creditors' Meeting (and both prior to and subsequent to the obtaining of the Sanction Order), the Company, in consultation with the Monitor and, if prior to the Plan Implementation Date, the DIP Lender (if any DIP Claim then remains), at any time and from time to time vary, amend, modify or supplement this Plan, (except to reduce the amount of the Aggregate Distribution or extend any Payment Dates other than as expressly provided herein), without the need for obtaining an Order of the Court or providing notice to the Affected Creditors if the Monitor determines that such variation, amendment, modification or supplement would not be materially prejudicial to the interests of the Affected Creditors under this Plan or the Sanction Order and is necessary in order to give effect to the substance of this Plan or the Sanction Order. All of the foregoing shall not require any further vote by or approval by the Affected Creditors or any approval by the Court.

9.4. Deeming Provisions

In this Plan, any deeming provisions are not rebuttable and are conclusive and irrevocable.

9.5. Sections 38 and 95 to 101 BIA

Notwithstanding Section 36.1 of the *CCAA*, Section 38 and Sections 95 through 101 of the *BIA* shall not apply to this Plan. Accordingly, neither the Monitor, any Creditor nor any other Person shall be entitled to exercise any right, remedy or recourse, or to commence any action, motion, application or any other proceeding against any member of the Fisher Group, any Creditor or any other Person in relation to the Company, based on Section 38 and Sections 95 through 101 of the *BIA*.

9.6. Responsibilities of the Monitor

The Monitor is acting in its capacity as Monitor in the CCAA Proceedings with respect to the Company and not in its personal or corporate capacity and will not be responsible or liable for any responsibilities or obligations of the Company under this Plan or otherwise, including with respect to the making of distributions or the receipt of any distribution by any Creditor or any other Person pursuant to the Plan. The Monitor will have the powers and protections granted to it by this Plan, the *CCAA*, the Initial Order, the Claims Procedure Order, the Meeting Procedure Order, the Sanction Order and any other Orders.

9.7. <u>Liability Limitations</u>

The Monitor, the Company and their respective legal counsel and other professional advisors, shall not be liable to any Person for any act or omission in connection with, or arising out of, the CCAA Proceedings, the Plan, the pursuit of sanctioning of the Plan, the consummation and implementation of the Plan or the administration of the Plan or the funds to be distributed under the Plan, except for their own wilful misconduct or gross negligence.

9.8. Notices

- (a) Any notices or communication to be made or given hereunder to the Company or the Monitor shall be in writing and shall refer to this Plan and may, subject as hereinafter provided, be made or given by personal delivery, by courier, by prepaid mail, by fax or by e-mail, addressed to the respective parties as follows:
 - (i) if to the Company:

LAURA'S SHOPPE (P.V.) INC.

2955 Jules-Brillant Laval, Quebec H7P 6B2

Attention:

Kalman Fisher and

Josh Fisher

E-Mails:

kfisher@laura.ca

josh.fisher@laura.ca

Fax:

(450) 973-6099

with copy to:

KUGLER KANDESTIN LLP

1 Place Ville Marie Suite 2101 · Montreal, Quebec H3B 2C6

Attention:

Gerald F. Kandestin,

David Stolow and

Jeremy Cuttler

E-Mail:

gkandestin@kklex.com

dstolow@kklex.com

jcuttler@kklex.com

Fax:

(514) 875-8424

(ii) if to the Monitor:

KPMG LLP

KPMG Tower 600 De Maisonneuve Blvd., West Suite 1500 Montreal, Quebec H3A 0A3

Attention:

Dev Coossa

E-Mail:

dcoossa@kpmg.ca

Fax:

(514) 840-2121

with copy to:

FISHMAN FLANZ MELAND PAQUIN, L.L.P.

1250 Rene-Levesque Blvd., West Suite 4100 Montreal, Quebec H3B 4W8

Attention:

Avram Fishman and

Mark Meland

E-Mails:

afishman@ffmp.ca

mmeland@ffmp.ca

Fax:

(514) 932-4170

or to such other address as any party may from time to time notify the others in accordance with this **Section 9.8**. In the event of any strike, lock-out or other event which interrupts postal service in any part of Canada, all notices and communications during such interruption may only be given or made by personal delivery, fax or e-mail and any notice or other communication given or made by prepaid mail within the 5 Business Day period immediately preceding the commencement of such interruption, unless actually received, shall be deemed not to have been given or made. All such notices and communications shall be deemed to have been received, in the case of notice by delivery, fax or e-mail prior to 5:00 p.m. (local time) on a Business Day, when received or if received after 5:00 p.m. (local time) on a Business Day or at any time on a non-Business

Day, on the next following Business Day and, in the case of notice mailed as aforesaid, on the fourth Business Day following the date on which such notice or other communication is mailed. The unintentional failure by the Company or the Monitor to give any notice contemplated hereunder to any particular Creditor shall not invalidate this Plan or any action taken by any Person pursuant to this Plan.

(b) Any notices or communication to be made or given hereunder by the Monitor or the Company to a Creditor may be sent by fax, e-mail, ordinary mail, registered mail, courier or facsimile transmission to the e-mail address, address or fax number specified by such Creditor in its Proof of Claim, or in any subsequent written notice of change of address given to the Monitor. A Creditor shall be deemed to have received any document sent pursuant to this Plan 4 Business Days after the document is sent by ordinary or registered mail and on the Business Day immediately following the day on which the document is sent by courier, fax or e-mail.

9.9. Severability of Plan Provisions

If, prior to the Sanction Date, any term or provision of this Plan is held by the Court to be invalid, void or unenforceable, the Court, at the request of the Company which request shall be made in consultation with the Monitor and the DIP Lender (if any DIP Claim then remains), shall have the power to either:

- (a) sever such term or provision from the balance of this Plan and provide the Company with the option to proceed with the implementation of the balance of this Plan as of and with effect from the Plan Implementation Date; or
- (b) alter and interpret such term or provision to make it valid and enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpreted.

Notwithstanding any such severing, holding, alteration or interpretation, and provided the Company proceeds with the implementation of this Plan, the remainder of the terms and provisions of this Plan shall remain in full force and effect and shall in no way be affected, impaired or invalidated by such severing, holding, alteration or interpretation.

9.10. Revocation, Withdrawal or Non-Consummation

The Company, upon consultation with the Monitor and the DIP Lender, reserves the right to revoke or withdraw this Plan at any time prior to the Sanction Date and to file subsequent plans of arrangement and compromise. If the Company revokes or withdraws this Plan, or if the Sanction Order is not issued:

- (a) this Plan shall be null, void and inoperative in all respects;
- (b) any Claim, any settlement or compromise embodied in this Plan (including the fixing or limiting of any Claim to an amount certain), assumption or termination by this Plan, and any document or agreement executed pursuant to this Plan shall be deemed null, void and inoperative; and
- (c) nothing contained in this Plan, and no act taken in preparation for consummation of this Plan, shall:
 - constitute or be deemed to constitute a waiver or release of any Claims by or against the Company or any other Person;
 - (ii) prejudice in any manner the rights of the Company or any Person in any further proceedings involving the Company; or
 - (iii) constitute an admission of any sort by the Company or any other Person.

9.11. Further Assurance

Notwithstanding that the transactions and events set out in this Plan shall occur and be deemed to occur in the order set out herein without any additional act or formality, each of the Persons affected hereby shall make, do and execute, or cause to be made, done and executed at the cost of the requesting party, all such further acts, deeds, agreements, transfers, assurances, instruments or documents as may reasonably be required by the Company in order to better implement this Plan.

9.12. Governing Law

This Plan shall be governed by and construed in accordance with the law of the Province of Quebec and the federal laws of Canada applicable therein. Any questions as to the interpretation or application of this Plan and all proceedings taken in connection with this Plan and its provisions shall be subject to the exclusive jurisdiction of the Court.

9.13. Successors and Assigns

This Plan shall be binding upon and shall enure to the benefit of the heirs, administrators, executors, legal personal representatives, successors and permitted assigns of the Company, the Directors, the Affected Creditors or any other Persons affected by or benefiting from the provisions of this Plan.

9.14. French Language Version of Plan

In the event of any discrepancy between any of the provisions of the English language version of this Plan and any French language version thereof, the provisions of the English version of this Plan shall, under all circumstances, prevail and govern.

9.15. Choice of Language

The Company and each of the hereafter enumerated members of the Fisher Group acknowledges that it (or he) has required that this Plan and all related documents be prepared, in English. La Compagnie et chacun des membres du "Fisher Group" ci-après énumerés reconnaît avoir exigé que ce Plan et tous les documents connexes soient rédigés en anglais.

(signature page to follow)

MONTREAL, Province of Quebec, this	day of	, 2015.
	LAURA'S SHOPPE (P.V.) INC. Per:	·
	Kalman Fisher, President	**************************************
EACH of the following members of the cognizance of the contents of this Plan a hereof.	e Fisher Group acknowledges nd consents to the provisions of	haven taken Section 3.11
	KALMAN FISHER	A design of the second of the
	3482731 CANADA INC. Per:	
	Kalman Fisher	***************************************
	9318-5494 QUEBEC INC. Per:	
	Kalman Fisher	

•

A-N-N-E-X (TO-PLAN-OF-COMPROMISE AND ARRANGEMENT OF LAURA'S SHOPPE (P.V.) INC.)

DISCLAIMED LEASES

	LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
4	Calloway REIT (Halifax) Ltd. (Smartcentres)	Bayers Lake Center 208 Chain Lake Drive Halifax, Nova Scotia B3S 1C5	Laura
2.	The Boulevard Shopping Centre (Montreal) Limited Partnership	The Boulevard Shopping Centre 4270 Jean Talon Street East Montreal, Quebec H1S-1J7	Laura
3	Calloway Reit (Cambridge) Inc. (Smartcentres)	Cambridge Centre 28 Pinebush Road Cambridge, Ontario	Laura
4	Riocan Holdings (Quebec) Inc.	Mega Centre Notre-Dame Notre-Dame and Autoroute 13 Laval, Quebec H7X-3V5	Laura
5	3945333 Manitoba Limited (Smartcentres)	SmartCentres Winnipeg-Southwest 1665 Kenaston Blvd. Unit P3 Winnipeg, Manitoba R3P 2M4	Laura
6	Riocan Property Services Trust as agent for 2113362 Ontario Limited, as to an undivided one half interest, and Trinity Properties Alberta Limited as to an undivided one- half interest	Beacon Hill Centre 11834 Sarcee Trail NW Calgary, Alberta T3R-0A1	Laura
7	Beauward Shopping Centres Ltd.	Galeries Joliette 1075 Firestone Blvd. Joliette, Quebec J6E-6X6	Laura `

	LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
8	Montez Core Income Fund Limited Partnership—and- Hillcrest Property Holdings Inc.	Hillerest Mall 9350 Yonge-Street Richmond Hill, Ontario L4C 5G2	Laura
9,	Carrefour-Richelieu-Realties Ltd. (Westeliff)	Carrefour Angrignon 7077 Newman Blvd. LaSalle, Quebec H8N 1X1	Laura Plus
10.	Omers Realty Management Corporation, by its manager OPGI Management GP Inc. (as general partner of the OPGI Management Limited Partnership)	Square One Shopping Centre 100 City Centre Drive Mississauga, Ontario L5B-2C9	Laura Plus
11	Park Royal Shopping Centre Holdings Ltd.	Park-Royal Shopping Centre 2002 Park-Royal South West Vancouver, BC V7T-2W4	Melanie Lyne
12.	Oxford Properties Group Incand-CPP Investment Board Real Estate Holdings Inc.	Edmonton City Centre West 10020-102 Avenue Edmonton, Alberta T5J 4B7	Melanie-Lyne
13.—	Dartmouth-Grossing-Limited	Dartmouth Crossing 34 Logicalmond Close Dartmouth, Nova Scotia B3B-0C8	Melanie Lyne
14.	RRL Burloak Inc. and 2121049 Ontario Limited, by its agent RioCan Property Services Inc. (Trustee for RioCan Real Estate Investment Trust)	RioCan-Centre-Burloak 3543-Wyeeroft-Road Oakville, Ontario	Melanie Lyne
15.—	Riotrin Properties (Orleans) Inc.	Trinity Crossing 4210 Innes Road Orleans, Ontario K4A 5E6	Melanie-Lyne

	LANDLORD	STORE/OFFICE LOCATION	STORE BANNER (if applicable)
16. –	9191-3723 Quebec Inc.	2955 rue Jules-Brillant Laval, Quebec H7P-6B2	N/A
17. -	Desjardins Financial Security Life Assurance Company	151-City-Centre Drive Suites-603/701 Mississauga, Ontario L5B-1M7	N/A
 48	3608727 Canada Inc.	2519-2525 boul. Le Corbusier Laval, Quebec H7S-1Z4	A\\A

EXHIBIT P-6

(Draft Sanction Order)

CANADA

SUPERIOR COURT Commercial Division

PROVINCE OF QUÉBEC DISTRICT OF MONTRÉAL

(Sitting as a court designated pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. c. C-36)

N°: 500-11-049256-155

IN THE MATTER OF THE PLAN OF COMPROMISE AND ARRANGEMENT OF:

MAGASIN LAURA (P.V.) INC. / LAURA'S SHOPPE (P.V.) INC.

Petitioner

-and-

BOUTIQUE LAURA CANADA LTÉE / LAURA'S SHOPPE CANADA LTD.

-and-

3482731 CANADA INC.

-and-

9318-5494 QUEBEC INC.

-and-

KALMAN FISHER

Stayed Parties

-and-

KPMG INC.

Monitor

SANCTION ORDER

(Sections 6, 9, 11 and 11.02 of the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36)

THE COURT is seized with the "Motion for Sanction of the Petitioner's Re-Amended Plan of Compromise and Arrangement and for Other Relief" (the "**Motion**") filed by Magasin Laura (P.V.) Inc. / Laura's Shoppe (P.V.) Inc. (the "**Petitioner**");

SEEING the Initial Order issued by this Honourable Court on August 12, 2015, as amended, pursuant to the *Companies' Creditors Arrangement Act* (the "*CCAA*");

SEEING the Claims Procedure Order issued by this Honourable Court on September 11, 2015 pursuant to the *CCAA*;

SEEING the Order for the Filing of the Plan of Compromise and Arrangement and the Calling of a Creditors' Meeting issued by this Honourable Court on October 23, 2015;

CONSIDERING the allegations contained in the Motion, the exhibits and the affidavit in support thereof;

CONSIDERING the representations of counsel;

CONSIDERING the Monitor's Fourth Report (the "Monitor's Report");

CONSIDERING Sections 6, 9, 11 and 11.02 of the *CCAA*;

CONSIDERING that the Motion was duly served to the service list in the *CCAA* Proceedings;

FOR THE FOREGOING REASONS, THE COURT:

- [1] **GRANTS** the Motion;
- [2] **DECLARES** that the service of the Motion constitutes good and sufficient service on all persons;
- [3] **DECLARES** that, unless otherwise indicated, capitalized terms found herein shall have the same meaning ascribed thereto in the Petitioner's Re-Amended Plan of Compromise and Arrangement dated November 18, 2015 and filed into the Court record on November 19, 2015, a copy of which is annexed hereto (the "**Plan**");

Notification and Meeting

[4] **DECLARES** that the notification procedures set forth in the Meeting Procedure Order have been duly followed and that there has been valid and sufficient notice of the Creditors' Meeting and service, delivery and notice of the "Meeting Materials" contemplated therein and that no further notice is or shall be required;

Plan Sanction, Plan Implementation, Releases and Discharges

[5] **ORDERS** and **DECLARES** that the Plan, including the compromises and arrangements set forth therein, is hereby sanctioned and approved pursuant to Section 6 of the *CCAA* and, as at the Plan Implementation Date, will be effective and will enure to the benefit of and be binding upon the Petitioner, the Affected Creditors and all other Persons referred to in the Plan or this Order;

[6] **DECLARES** that:

- (a) the Plan has been approved by the Required Majorities of Affected Creditors of the Petitioner in conformity with the CCAA;
- (b) the Petitioner has complied with the provisions of the CCAA and the Orders of the Court made in the CCAA Proceedings in all respects;
- (c) the Court is satisfied that the Petitioner has neither done nor purported to do anything that is not authorized by the CCAA; and
- (d) the Plan and the transactions contemplated thereby are fair and reasonable.
- [7] **DECLARES** that the full and final release and discharge of the Claims will be effective and will enure to the benefit of and be binding upon the Petitioner, the Affected Creditors and all other Persons referred to in the Plan upon the Monitor's issuance and filing with the Court of the Certificate of Performance;
- [8] **ORDERS** that the Petitioner and the Monitor are authorized to take all steps and actions necessary to implement the Plan;
- [9] ORDERS and DECLARES that all Proven Claims determined in accordance with the Claims Procedure Order and the Plan are final and binding on the Petitioner and all Affected Creditors;
- [10] **ORDERS** and **DECLARES** that, subject to Sections 4.5 and 4.6 of the Plan, any Claims for which a Proof of Claim has not been filed by the Claims Bar Date are and shall be forever barred and extinguished;
- [11] **DECLARES** that all distributions and payments by or at the direction of the Monitor, in each case on behalf of the Petitioner, under the Plan are for the account of the Petitioner and the fulfillment of its obligations under the Plan;
- [12] **DECLARES** that the Petitioner and the Monitor may apply to the Court for advice and direction in respect of any matters arising from or under the Plan;
- [13] **ORDERS** and **DECLARES** that, subject to the performance by the Petitioner of its obligations under the Plan, all contracts, leases, agreements and arrangements to which the Petitioner is a party and that have not been disclaimed, terminated or repudiated pursuant to the Initial Order and/or the

CCAA will be and remain in full force and effect, unamended, as at the Plan Implementation Date, and no Person who is a party to any such contract, lease, agreement or other arrangement may accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of dilution or other remedy) or make any demand under or in respect of any such contract, lease, agreement or other arrangement and no automatic termination will have any validity or effect, by reason of:

- (a) any event that occurred on or prior to the Plan Implementation Date and is not continuing that would have entitled such Person to enforce those rights or remedies (including defaults, events of default, or termination events arising as a result of the insolvency of the Petitioner);
- (b) the insolvency of the Petitioner, the fact that the Petitioner filed a notice of intention to make a proposal under the *BIA* or the fact that the Petitioner sought or obtained relief under the *CCAA*; or
- (c) any compromises or arrangements effected pursuant to the Plan or any action taken or transaction effected pursuant to the Plan.
- ORDERS that, subject always to any exceptions set forth in Section 6.4 of the [14] Plan, upon the Monitor's issuance and filing with the Court of the Certificate of Implementation on the Plan Implementation Date, each of the First Released Parties shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Person may be entitled to assert whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with the business and affairs of the Petitioner, the Plan and the CCAA Proceedings, and all claims arising out of such actions or omissions shall be forever waived and released, all to the full extent permitted by law and ORDERS that, upon the implementation of the Plan on the Plan Implementation Date, the commencement or prosecution, whether directly, derivatively or otherwise, of any of the above claims or matters against the First Released Parties is hereby enjoined;
- [15] **ORDERS** that, subject always to any exceptions set forth in Section 6.4 of the Plan, upon the Monitor's issuance and filing with the Court of the Certificate of Performance, each of the Second Released Parties shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any

Person (including any Secured Creditor in respect of each Second Released Party, except the Petitioner and solely in respect to its Secured Claim), may be entitled to assert (including any and all Claims in respect of statutory liabilities and any Directors' Claims of all Directors and all present and former officers and employees of the Petitioner and any alleged fiduciary or other duty), whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with Claims, the business and affairs of the Petitioner, the Plan and the CCAA Proceedings, and all claims arising out of such actions or omissions shall be forever waived and released (other than the right to enforce the Petitioner's obligations under the Plan or any related document), all to the full extent permitted by law and ORDERS that, upon the Monitor's issuance and filing with the Court of the Certificate of Performance, the commencement or prosecution, whether directly, derivatively or otherwise, of any of the above claims or matters against the Second Released Parties is hereby enjoined:

General Provisions

- [16] **ORDERS** that all Orders in the *CCAA* Proceedings shall continue in full force and effect in accordance with their respective terms, except to the extent that such Orders are varied by, or inconsistent with, this Sanction Order or the Plan;
- [17] **DECLARES** that this Order shall have full force and effect in all provinces and territories in Canada;
- [18] **REQUESTS** the aid and recognition of any Court or administrative body in any Province of Canada and any Canadian federal court or administrative body and any federal or state court or administrative body in the United States of America and any court or administrative body elsewhere, to act in aid of and to be complementary to this Court in carrying out the terms of this Order;
- [19] **ORDERS** the provisional execution of this Order notwithstanding any appeal;
- [20] **THE WHOLE** without costs.

Montréal, November	, 2015
MARIE-ANNE PAQUETT	F JSC

ANNEX

COPY OF RE-AMENDED PLAN OF COMPROMISE AND ARRANGEMENT OF LAURA'S SHOPPE (P.V.) INC.

CANADA

PROVINCE OF QUEBEC DISTRICT OF MONTREAL

NO.: 500-11-049256-155

SUPERIOR COURT (Commercial Division)

(Sitting as a court designated pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. c. C-36)

IN THE MATTER OF THE PLAN OF COMPROMISE AND ARRANGEMENT OF:

MAGASIN LAURA (P.V.) INC. / LAURA'S SHOPPE (P.V.) INC.

Debtor/Petitioner

-and-

BOUTIQUE LAURA CANADA LTÉE / LAURA'S SHOPPE CANADA LTD.

-and-

3482731 CANADA INC.

-and-

9318-5494 QUEBEC INC.

-and-

KALMAN FISHER

Stayed Parties

-and-

KPMG INC.

Monitor

RE-AMENDED PLAN OF COMPROMISE AND ARRANGEMENT Under the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36

TABLE OF CONTENTS

1	INTERPRETATION	<u>1</u>
1.1.	Definitions	
1.2.	Interpretation	9
1.3.	Date For Any Action	.10
1.4.	Statutory References	.10
2.	COMPROMISE AND ARRANGEMENT	10
2.1.	Persons Affected	.10
2.2.	Classes of Affected Claims	. 10
2.3.	Disclaimed Landlord Claims	.10
2.4.	Unaffected Claims	.10
2.5.	Treatment of Affected Claims	.11
2.6.	Settlement of Affected Claims	.12
2.7.	Extension of Payment Dates	.12
2.8.	Prepayment Rights	. 13
2.9.	No Acknowledgment or Admission	. 13
3.	TREATMENT OF OTHER CLAIMS AND MATTERS	<u> 13</u>
3.1.	Treatment of the DIP Claim	. 13
3.2.	Treatment of Excluded Claims	. 13
3.3.	Treatment of Administration Claims	. 13
3.4.	Treatment of Gift Card Claims	.14
3.5.	Treatment of Certain Crown Priority Claims	14
3.6.	Treatment of Section 6(5) Claims	.14
3.7.	Treatment of Section 19(2) Claims	14
3.8.	Treatment of KERP Claims	14
3.9.	Treatment of Secured Claims	14
3.10.	Treatment of Replacement Financing Claim	14
3.11.	Treatment of Fisher Group Claims	14
4.	VALUATION OF CLAIMS, CREDITORS' MEETING AND RELATED MATTERS	<u>. 15</u>
4.1.	Conversion of Affected Claims into Canadian Currency	15
4.2.	Affected Claims	15
4.3	Creditors' Meeting	15

4.4.	Approval by Affected Creditors	15
4.5.	Amendment of Proofs of Claim	16
4.6.	Late Filing of Claims for Disclaimed Leases	16
4.7.	Claims Bar Date	16
5. PRO	CEDURE FOR RESOLVING DISTRIBUTIONS IN RESPECT OF DISPUTE	D CLAIMS 16
5.1.	No Distributions Pending Allowance	16
5.2.	Distribution From Reserve Once Disputed Claims Resolved	16
6. EFF	ECT OF THE PLAN AND RELEASES	17
6.1.	Effect of the Plan	17
6.2.	Releases upon Implementation	17
6.3.	Releases after Performance	18
6.4.	Exception to Releases	18
6.5.	Set-Off Rights	19
6.6.	Injunction Related to Releases	19
6.7.	Waiver of Defaults	19
6.8.	Stay of Proceedings	20
7. PRO	VISIONS GOVERNING DISTRIBUTIONS	20
7.1.	Partial Distributions for Claims Allowed	20
7.2.	Currency to be used for the Distribution	20
7.3.	Assignment of Claims	20
7.4.	Interest on Affected Claims	20
7.5.	Distributions by Monitor	21
7.6.	Delivery of Distributions	21
7.7.	No Double Recovery	21
7.8.	Withholding Requirements	22
8. IMPL	EMENTATION OF THE PLAN	22
8.1.	Conditions Precedent to Implementation of Plan	22
8.2.	Waiver of Conditions	24
8.3.	Certificate of Implementation	25
B.4.	Effect of Plan	25
8.5.	Nullity of Plan	25
9. MISC	ELLANEOUS	26
9.1.	Confirmation of Plan	26

9.2.	Paramountcy	26
9.3.	Modification of Plan	26
9.4.	Deeming Provisions	27
9.5.	Sections 38 and 95 to 101 BIA	27
9.6.	Responsibilities of the Monitor	27
9.7.	Liability Limitations	27
9.8.	Notices	28
9.9.	Severability of Plan Provisions	30
9.10.	Revocation, Withdrawal or Non-Consummation	30
9.11.	Further Assurance	31
9.12.	Governing Law	31
9.13.	Successors and Assigns	31
9.14.	French Language Version of Plan	31
9.15.	Choice of Language	31

1. INTERPRETATION

1.1. Definitions

In this Plan, unless otherwise stated or the context otherwise requires:

"Administration Charge" has the meaning ascribed to such term in the Initial Order;

"Administration Claim" means a claim or any other indebtedness or obligation secured by the Administration Charge;

"Affected Claim" means any Claim other than an Unaffected Claim. For greater certainty, Affected Claims include Disclaimed Landlord Claims, any Claim of Her Majesty the Queen in right of Canada or any province (other than Crown Priority Claims), Directors Claims and Restructuring Claims;

"Affected Creditor" means a Creditor holding an Affected Claim, but only to the extent of its Affected Claim;

"Affected Creditors Class" has the meaning set forth in Section 2.2 hereof;

"Aggregate Distribution" has the meaning set forth in Section 2.5 hereof;

"BIA" means the Bankruptcy and Insolvency Act, R.S.C., 1985, c. B-3, as amended;

"Business Day" means a day, other than a Saturday, a Sunday, or a non-juridical day (as defined in article 6 of the Quebec Code of Civil Procedure, R.S.Q., c. C-25, as amended);

"CCAA" means the Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended;

"CCAA Charge(s)" means the Administration Charge, the DIP Lender Charge and the KERP Charge;

"CCAA Proceedings" means proceedings in respect of the Company before the Court commenced, taken up and continued under the CCAA;

"Certificate of Implementation" has the meaning set forth in Section 8.3 hereof;

"Certificate of Performance" has the meaning set forth in Section 6.1 hereof;

"Certificate of Non-Implementation" has the meaning set forth in Section 8.5 hereof;

"Chair" means a representative of the Monitor, or any other individual designated by the Monitor to preside as chairperson at the Creditors' Meeting;

"Claim" means any right of any Person against the Company in connection with any indebtedness, liability or obligation of any kind of the Company owed to such Person and any interest, or penalties accrued thereon or costs payable in respect thereof, whether liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, unsecured, present, future, known or unknown, by guarantee, surety or otherwise, and whether or not such right is executory or anticipatory in nature, including the right or ability of any Person to advance a claim for contribution or indemnity or otherwise with respect to any matter, action, cause or chose in action, whether existing at present or commenced in the future, which indebtedness, liability or obligation is based in whole or in part on facts existing, or transactions which occurred, prior to the Determination Date, or which would have been claims provable in bankruptcy had the Company become bankrupt on the Determination Date, and, without limitation, including any Restructuring Claim, but excluding any Unaffected Claim and "Claims" means all of them;

"Claims Bar Date" has the meaning as set forth in the Claims Procedure Order;

"Claims Procedure Order" means the "Claims Procedure Order" issued by the Court on September 11, 2015, establishing, among other things, procedures for proving Claims, as amended or supplemented from time to time by further Order(s) of the Court;

"Company", "Debtor" or "Petitioner" means Magasin Laura (P.V.) Inc./Laura's Shoppe (P.V.) Inc.;

"Court" means the Commercial Division of the Quebec Superior Court for the District of Montreal:

"Creditor(s)" means any Person(s) having a Claim and may, where the context requires, include the assignee of a Claim or a trustee, interim receiver, receiver, receiver and manager, or other Person acting on behalf of such Person(s);

"Creditors' Meeting" means the meeting of Affected Creditors to be convened for the purposes of voting on the Plan, or any adjournment of such meeting;

"Crown Priority Claims" means any Claims of Her Majesty the Queen in right of Canada or in right of any province as described in Section 6(3) or Section 38(2) of the CCAA. For greater certainty, any Claim of Her Majesty the Queen in right of Canada or in right of any Province other than Crown Priority Claims shall be an Affected Claim hereunder;

"Determination Date" means July 31, 2015, being the date on which the Company commenced proceedings under the *BIA* by filing a notice of intention to make a proposal thereunder, which proceedings were subsequently taken up and continued by the *CCAA* Proceedings under the *CCAA*;

"DIP Claim" means any and all obligations of the Company to the DIP Lender pursuant to the DIP Facility Documents or the Order for New Financing as of or after the Plan Implementation Date:

"DIP Facility Documents" means the "committed term sheet" dated October 22, 2015 by and among, *inter alia*, the Company and the DIP Lender (as amended, modified, restated and/or supplemented from time to time) together with any related collateral, loan or security documents executed in connection therewith or which relate thereto;

"DIP Lender" means collectively, Third Eye Capital Corporation, acting as agent and hypothecary representative for and on behalf of certain affiliates and funds it manages as well as all of the "New Financing Parties" as defined in the Order for New Financing;

"DIP Lender Charge" means the "New Lender Charge" as set forth in the Order for New Financing;

"DIP Loan" means the loan(s) made and/or to be made by the DIP Lender as envisaged by the Order for New Financing;

"Directors" means all of the Company's past and present directors as well as any Persons who were or are deemed to be directors of the Company pursuant to any applicable Laws;

"Directors' Claims" means all Claims, of any nature or source whatsoever, of any of the Creditors against any or all of the Directors which arose before the Determination Date and which relate to obligations of the Company where such Directors are by law liable in their capacity as directors for payment of such obligations. Director's Liability shall, however, exclude all Claims which (i) relate to contractual rights of one or more Creditors arising from contracts with one or more Directors, including any guarantees given by any Directors relating to the Company's obligations, (ii) seek injunctive relieve against any of the Directors in relation to the Company, or (iii) are based on allegations of misrepresentation made by any Directors to Creditors or of wrongful or oppressive conduct by Directors;

"Disallowed Claim" means any Claim, or that portion thereof which has been revised or disallowed by the Monitor pursuant to the Claims Procedure Order or any other Order in respect of which an appeal by the Creditor has been dismissed or all appeal periods, if any, have expired;

"Disclaimed Leases" means each and every one of the Leases which (i) were disclaimed or resiliated by the Company pursuant to Section 65.2 of the *BIA* (as recognized by the Initial Order) or Section 32 of the *CCAA*, and (ii) may be disclaimed or resiliated by the Company at any time between the date of this Plan and the Sanction Date, and "Disclaimed Lease" means each of them;

"Disclaimed Landlord" means the landlord under any Disclaimed Lease and "Disclaimed Landlords" means all of them:

"Disclaimed Landlord Claims" means all Proven Claims of the Disclaimed Landlords for losses incurred by Disclaimed Landlords in relation to Disclaimed Leases, as provided for in Section 2.3 hereof and "Disclaimed Landlord Claim" means any of them:

"Disputed Claim" means a Claim or that portion thereof that is the object of a Notice of Revision or Disallowance and, in either case, has not become a Proven Claim or a Disallowed Claim:

"Excluded Claim" means any right of any person against the Company in connection with any indebtedness, liability or obligation of any kind which arose with respect to transactions which occurred after the Determination Date and any interest thereon, including the Administration Claim, the DIP Claim, a KERP Claim, the Replacement Financing Claim and any obligation of the Company towards creditors who have supplied or shall supply services, utilities, goods or materials or who have or shall have advanced funds to the Company after the Determination Date, but only to the extent of their claims in respect of the supply of such services, utilities, goods, materials or funds after the Determination Date and to the extent that such claims are not otherwise affected by the Plan. For greater certainty, a Restructuring Claim is not an Excluded Claim:

"Excluded Creditor" means a Person having a claim in respect of an Excluded Claim but only in respect of such Excluded Claim and to the extent that the Plan does not otherwise affect such claim;

"First Released Parties" has the meaning as set forth in Section 6.2 hereof;

"Fisher Group" means Kalman Fisher, 3482731 Canada Inc., 9318-5494 Quebec Inc. (formerly 7735235 Canada Inc.) and any Person who/which is, or was at any time within the 12 consecutive months immediately preceding the Determination Date, a Related Person to Kalman Fisher;

"Fisher Group Claims" means any Claims for payment of principal on any loans or advances made to the Company (or its predecessor) at any time prior to the Determination Date by any member of the Fisher Group, which includes any Secured Claim by any member of the Fisher Group, but excludes any Claims or future claims by any member of the Fisher Group for (i) salary (or other remuneration), (ii) reimbursement of business expenses, or (iii) payment of principal on any loans or advances in lieu of salary (or other remuneration) or in lieu of interest;

"Gift Card Claim" has the meaning set forth in Section 2.4(a)(v) hereof;

"Governmental Authority" means any (i) multinational, national, provincial, state, regional, municipal, local or other government, governmental or public department, ministry, central bank, court, tribunal, arbitral body, commission, board, official, minister, bureau or agency, domestic or foreign, (ii) subdivision, agent, commission, board or authority of any of the foregoing; or (iii) quasi-governmental or private body, including any tribunal, commission, regulatory agency or self-regulatory organization, exercising any regulatory, expropriation or taxing authority under, or for the account of, any of the foregoing;

"Initial Order" means the "Initial Order" issued by the Court on August 12, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"KERP Charge" has the meaning ascribed to such term in the Initial Order:

"KERP Claim" means a claim or any other indebtedness or obligation secured by the KERP Charge and "KERP Claims" means all of them;

"Laws" means all laws, statutes, codes, ordinances, decrees, rules, regulations, by-laws, judicial or arbitral or administrative or ministerial or departmental or regulatory judgments, injunctions, orders or decisions of any Governmental Authority, statutory body or self-regulatory authority, including general principles of law having the force of law and the term "applicable" with respect to such Laws and, in the context that refers to any Person, means such Laws as are applicable to such Person or its business, undertaking, property or securities and emanate from a Governmental Authority or self-regulatory authority having jurisdiction over the Person or its business, undertaking, property or securities;

"Lease" means any lease and/or other agreement for the occupancy of real and immovable property as well as all amendments thereto and renewals thereof and "Leases" means all of them;

"Meeting Date" means the date fixed for the Creditors' Meeting in accordance with the Meeting Procedure Order, or any subsequent Order, or any subsequent date following any adjournment of that meeting, as the case may be;

"Meeting Procedure Order" means the "Meeting Procedure Order" issued by the Court on October 23, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"Monitor" means KPMG Inc., in its capacity as Monitor, as appointed by the Court pursuant to the Initial Order;

"Notice of Revision or Disallowance" has the meaning as set forth in the Claims Procedure Order;

"Order" means any order of the Court in the CCAA Proceedings;

"Order for New Financing" means the "Order for New Financing" issued by the Court on October 30, 2015, as renewed and amended by the Court from time to time, under the CCAA;

"Original Currency" has the meaning set forth in Section 7.7 hereof;

"Payment Date" and "Payment Dates" have the meaning set forth in Section 2.5 hereof;

"Person" means any individual, corporation, limited or unlimited liability company, general or limited partnership, association, trust, trustee, executor, administrator, legal personal representative, estate, unincorporated organization, joint venture, governmental body or agency, or any other entity;

"Plan" means the present "Re-amended Plan of Compromise and Arrangement" of the Company pursuant to the provisions of the CCAA, as may be amended, varied or supplemented by the Company from time to time in accordance with its terms;

"Plan Implementation Conditions" has the meaning set forth in Section 8.1 hereof;

"Plan Implementation Date" means the date, on or prior to the Plan Implementation Deadline, upon which all of the Plan Implementation Conditions have occurred or have been satisfied or waived;

"Plan Implementation Deadline" has the meaning set forth in Section 8.1 hereof;

"Proof of Claim" has the meaning set forth in the Claims Procedure Order;

"Proven Claim" means, in respect of a Creditor, the amount of the Claim of such Creditor as finally determined for distribution purposes in accordance with the provisions of this Plan, the CCAA and the Claims Procedure Order and "Proven Claims" means all of them;

"Related Persons" has the meaning set forth in Section 4 of the BIA:

"Replacement Financing" means the new senior credit facilities to be provided to the Company by the DIP Lender, including a term loan facility and a revolving asset-based loan facility on terms and conditions acceptable to the DIP Lender and the Company, which shall completely repay and/or replace the DIP Loan:

"Replacement Financing Claim" means any and all indebtedness and obligations of the Company and the DIP Lender under the Replacement Financing:

"Required Majorities" means the affirmative vote of a majority in number of the Affected Creditors voting in the Affected Creditors Class, having Voting Claims and voting on the Plan (in person or by proxy) at the Creditors' Meeting and representing not less than 66-2/3% in value of the Voting Claims of the Affected Creditors voting (in person or by proxy) at the Creditors' Meeting;

"Reserve(s)" means the reserve(s) to be established and maintained under this Plan by the Monitor by holding, on account of Disputed Claims, an amount equal to the aggregate amount that the holders of Disputed Claims would be entitled to receive if all such Disputed Claims had been Proven Claims as of any Payment Date;

"Resolution" means, collectively, one or any of, the resolutions providing for the approval of the Plan by the Affected Creditors;

"Restructuring Claim" means any right or claim of any Person against the Company in connection with any indebtedness or obligation of any kind owed to such Person arising out of the Company's disclaimer, resiliation or renegotiation (at any time after the Determination Date) of any Leases (excluding the Disclaimed Landlord Claims, which shall be dealt with pursuant to Section 2.3 hereof), agreements or other contracts or the Company's restructuring under the CCAA, and includes any right or claim resulting, directly or indirectly, from the consequences and effects of the Plan's acceptance by the Affected Creditors, the Plan's sanction by the Sanction Order, the Plan's implementation and any debt forgiveness resulting from any of the foregoing; provided, however, that a Restructuring Claim shall not include an Excluded Claim. For greater certainty, a Restructuring Claim is an Affected Claim;

"Sanction Date" means the date on which the Sanction Order is issued:

"Sanction Order" means the Order to be made under the CCAA sanctioning the Plan, as such Order may be affirmed, amended or modified by the Court at any time prior to the Plan Implementation Date or, if appealed, then, unless such appeal is withdrawn, abandoned or denied, as affirmed or amended on appeal, in form and content which is satisfactory to the Petitioner acting reasonably;

"Second Released Party" and "Second Released Parties" have the meanings as set forth in Section 6.3 hereof;

"Section 6(5) Claim" means any claim of an employee or former employee described in Section 6(5) of the CCAA but only to the extent of such amounts as required to be paid under the CCAA:

"Section 19(2) Claims" means any claim described in Section 19(2) of the CCAA;

"Secured Claim" means the Claim of a Secured Creditor, to the extent of the value of such Secured Creditor's security;

"Secured Creditor" has the meaning set forth in the CCAA, but only to the extent that such Creditor's mortgage, hypothec, pledge, charge, lien, privilege, security interest or other rights over the property of the Company was valid, opposable, perfected, and could be set up against third parties, including a trustee to the Company's bankruptcy, on both the Determination Date and the Claims Bar Date, failing which that Secured Creditor will be deemed to be an "unsecured creditor", as defined in the CCAA;

"Stayed Parties" means Boutique Laura Canada Ltée/Laura's Shoppe Canada Ltd., 3482731 Canada Inc., 9318-5494 Quebec Inc. (formerly 7735235 Canada Inc.) and Kalman Fisher:

"Taxes" means all federal, state, provincial, territorial, county, municipal, local or foreign taxes, duties, imposts, levies, assessments, tariffs and other charges imposed, assessed or collected by a Governmental Authority, including (i) any gross income, net income, gross receipts, business, royalty, capital, capital gains, goods and services, value added, severance, stamp, franchise, occupation, premium, capital stock, sales and use, real property, land transfer, personal property, ad valorem, transfer, licence, profits, windfall profits, environmental, payroll, employment, employer health, pension plan, anti-dumping, countervail, excise, severance, stamp, occupation, or premium tax, (ii) all withholdings on amounts paid to or by the relevant Person, (iii) all employment insurance premiums, Canada, Quebec and any other pension plan contributions or premiums, (iv) any fine, penalty, interest, or addition to tax, (v) any tax imposed, assessed, or collected or payable pursuant to any tax-sharing agreement or any other contract relating to the sharing or payment of any such tax, levy, assessment, tariff, duty, deficiency, or fee, and (vi) any liability for any of the foregoing as a transferee, successor, guarantor, or by contract or by operation of law;

"Unaffected Claim" means any right of any Person in connection with any indebtedness, liability or obligation of any kind of the type described in **Section 2.4(a)** hereof. For greater certainty, the Unaffected Claims include the Excluded Claims;

"Unaffected Creditor" means a Person having a Claim in respect of an Unaffected Claim, but only in respect of such Unaffected Claim, and for greater certainty, includes an Excluded Creditor;

"Voting Claim" means, in respect of an Affected Creditor, the amount of such Affected Creditor's claim which has been accepted for voting purposes in accordance with the provisions of this Plan, the Claims Procedure Order and the CCAA and "Voting Claims" means all of them:

1.2. Interpretation

For purposes of this Plan:

- (a) any reference in this Plan to a contract, instrument, release, indenture, agreement or other document being in a particular form or on particular terms and conditions means that such document shall be substantially in such form or substantially on such terms and conditions;
- (b) any reference in this Plan to an existing document or exhibit filed or to be filed means such document or exhibit as it may have been or may be amended, modified or supplemented;
- (c) all references to currency and to "\$", "C\$" or "Dollars" are to Canadian dollars except as otherwise indicated;
- (d) all references in this Plan to Sections are references to Sections of this Plan;
- (e) unless otherwise specified, the words "hereof", "herein" and "hereto" refer to this Plan in its entirety rather than to any particular portion of this Plan;
- (f) the division of this Plan into Sections and paragraphs and the insertion of captions and headings to Sections and paragraphs are for convenience of reference only and are not intended to affect the interpretation of, or to be part of this Plan;
- (g) where the context requires, a word or words importing the singular shall include the plural and vice versa and a word or words importing one gender shall include all genders;
- (h) the words "includes" and "including" are not limiting; and
- (i) the word "or" is not exclusive.

1.3. Date For Any Action

In the event that any date (including any Payment Date) on which any action (including any payment) is required to be taken under this Plan by any of the parties is not a Business Day, that action (including any payment) shall be required to be taken on the next succeeding day which is a Business Day.

1.4. Statutory References

Any reference in this Plan to a statute includes all regulations made thereunder and all amendments to such statute or regulations in force, from time to time, or any statute or regulations that supplement or supersede such statute or regulations.

2. COMPROMISE AND ARRANGEMENT

2.1. Persons Affected

The purpose of this Plan is to provide for the full and final settlement of all Affected Claims. Except as specifically provided for in this Plan, this Plan will become effective on the Plan Implementation Date in accordance with its terms, and all Affected Claims against the Company will be fully and finally, settled, compromised and released in this Plan upon the Aggregate Distribution being fully paid to the Monitor, it being understood that any Affected Claim that is paid in full pursuant to this Plan before such date shall be settled, compromised and released on the date of such payment. This Plan shall be binding on and enure to the benefit of the Company, the Affected Creditors, the First Released Parties, the Second Released Parties, any trustee, agent or other Person acting on behalf of any Affected Creditor and such other Persons who have received the benefit of, or are bound by any waivers, releases or indemnities hereunder.

2.2. Classes of Affected Claims

There shall only be one class of Affected Creditors for the purpose of voting on, and receiving distributions pursuant to this Plan, being the "Affected Creditors Class".

2.3. <u>Disclaimed Landlord Claims</u>

Disclaimed Landlord Claims shall be Affected Claims under this Plan and shall be determined in accordance with Section 32(7) CCAA.

2.4. Unaffected Claims

(a) This Plan does not affect the following claims (each, an "Unaffected Claim" and, collectively, the "Unaffected Claims"), the holders of which will not be entitled to

vote at the Creditors' Meeting or receive any distributions under this Plan, namely:

- (i) the Excluded Claims:
- (ii) any Administration Claim;
- (iii) any DIP Claim;
- (iv) any KERP Claims;
- (v) any claim with respect to gift-cards, gift certificates, lay-away deposits and other customer certificates (collectively, "Gift Card Claims"):
- (vi) Crown Priority Claims;
- (vii) Section 6(5) Claims;
- (viii) Section 19(2) Claims;
- (ix) Secured Claims;
- (x) the Replacement Financing Claim; and
- (xi) Fisher Group Claims,

all of which will be dealt with in Section 3 hereof;

(b) Nothing in this Plan shall affect the Company's rights and defences, both legal and equitable, with respect to any Unaffected Claim including any rights arising under or pursuant to the Claims Procedure Order or this Plan or any rights with respect to legal and equitable defences or entitlements to set-offs or recoupments against such Unaffected Claims.

2.5. Treatment of Affected Claims

By no later than each of the payment dates hereafter set forth (collectively the "Payment Dates" or individually a "Payment Date"), the aggregate sum of \$4 Million (the "Aggregate Distribution"), without any interest whatsoever thereon, shall be paid by the Company to the Monitor and shall thereafter be distributed by the Monitor to the Affected Creditors, according to the amount of their respective Proven Claims as hereafter set forth, namely:

- (a) by no later than April 30, 2016, \$1 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor as follows:
 - (i) an amount equal to the lesser of (A) the Proven Claim of each Affected Creditor, or (B) \$1,250, shall be distributed by the Monitor to the Affected Creditors, according to the amount of their Proven Claims; and
 - (ii) an amount equal to the difference between (A) \$1 Million, and (B) the aggregate amount to be distributed by the Monitor pursuant to Section 2.5(a)(i) above, shall be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the amounts set forth in Section 2.5(a)(i) above;
- (b) by no later than July 31, 2016, \$1 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distribution under Section 2.5(a) hereof;
- by no later than December 15, 2016, \$1.8 Million shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distributions under Sections 2.5(a) and 2.5(b) hereof; and
- (d) by no later than February 15, 2017, \$200,000 shall be paid by the Company to the Monitor, which shall thereafter be distributed by the Monitor to the Affected Creditors on a pro rata basis, according to the amounts of their respective Proven Claims, less any amounts received in respect of the distributions under Sections 2.5(a), 2.5(b) and 2.5(c) hereof.

2.6. Settlement of Affected Claims

The payment of the entire Aggregate Distribution to the Monitor as set forth in **Section 2** hereof shall constitute the full and final payment, reduction, settlement and transaction (as envisaged by the relevant provisions of the *Civil Code of Quebec*) of all of the Affected Claims of all of the Affected Creditors and engage the releases and other provisions set forth in this Plan.

2.7. Extension of Payment Dates

Any or all of the Payment Dates may be extended by the Company with the consent of the Monitor and the DIP Lender (if any portion of the DIP Loan or the Replacement Financing then remains outstanding), by additional periods not exceeding, in each case, 60 days. On application by the Company or the Monitor (in either case, with the consent of the DIP Lender, if any portion of the DIP Loan or the Replacement Financing remains outstanding) made prior to a Payment Date or, if applicable, prior to the expiry of any extension thereof as described above, the Court may issue Orders extending (or further extending) any or all of the Payment Dates for any additional period(s) deemed appropriate by the Court in the then existing circumstances.

2.8. Prepayment Rights

The Company (with the consent of the DIP Lender, if any portion of the DIP Loan or the Replacement Financing then remains outstanding) shall, at all times prior to the Final Distribution Date, be entitled to pre-pay the whole or any amount of the Aggregate Distribution prior to any Payment Date, whereupon the amount(s) of any such pre-payment(s) shall be distributed by the Monitor to the Affected Creditors in accordance with the provisions of **Section 2.5** hereof, *mutatis mutandis*.

2.9. No Acknowledgment or Admission

Nothing contained in this Plan shall constitute, under any circumstances whatsoever, an acknowledgment or admission by the Company as to the existence or validity of any Claims or Restructuring Claims.

3. TREATMENT OF OTHER CLAIMS AND MATTERS

3.1. Treatment of the DIP Claim

On or before the earlier of (i) the Plan Implementation Date, or (ii) the date on which the DIP Claim becomes payable pursuant to the provisions of the DIP Facility Documents, the DIP Lender shall receive full payment of any remaining DIP Claim.

3.2. Treatment of Excluded Claims

Subject to **Section 3.1** hereof, the Excluded Claims (other than the Administration Claim, the DIP Claim and any KERP Claim) will remain in full force and effect in accordance with their terms after the Plan Implementation Date, and will be paid in full by the Company in the normal course of its business as and when they become due.

3.3. Treatment of Administration Claims

All Administration Claims, if any, will be paid in full by the Company as and when they become due, and any remaining balance will be settled as soon as practicable after the Plan Implementation Date.

3.4. Treatment of Gift Card Claims

Gift Card Claims will be honored in accordance with the terms of the relevant gift card, gift certificate, lay-away deposit or other customer certificates upon presentation of such gift card, gift certificate, lay-away deposit or other customer certificates by the holder at any of the Petitioner's retail stores.

3.5. Treatment of Certain Crown Priority Claims

All Crown Priority Claims described in Section 6(3) of the *CCAA*, if any, will be paid in full by the Petitioner within 6 months immediately following the Sanction Date.

3.6. Treatment of Section 6(5) Claims

All Section 6(5) Claims, if any, will be paid in such amounts as required under the CCAA immediately after the Sanction Date.

3.7. Treatment of Section 19(2) Claims

All Section 19(2) Claims, if any, will be paid by the Company as and when they become due, unaffected by this Plan.

3.8. <u>Treatment of KERP Claims</u>

Each KERP Claim will be paid by the Company as and when it becomes due.

3.9. Treatment of Secured Claims

Secured Claims (other than the Administration Claim, the DIP Claim, the Replacement Financing Claim and KERP Claims) will be dealt with as provided for under agreements entered into or as may be hereafter entered into between the Company and the relevant Secured Creditor, or as may be ordered by the Court. For greater certainty, Secured Claims will be unaffected by this Plan.

3.10. Treatment of Replacement Financing Claim

The Replacement Financing Claim shall be paid by the Company to the DIP Lender in accordance with the provisions, terms and conditions of the Replacement Financing.

3.11. Treatment of Fisher Group Claims

Conditional upon the enactment of a Resolution approving the Plan, issuance of the Sanction Order and implementation of the Plan as set forth in **Section 8** hereof, each member of the Fisher Group:

- (a) subordinates and postpones all of the Fisher Group Claims to and in favour of full payment of the Aggregate Distribution to the Affected Creditors, such that no member of the Fisher Group shall be entitled to receive any payment from the Company of any principal on any of the Fisher Group Claims unless and until the full Aggregate Distribution shall have been paid to the Monitor as set forth in the Plan; and
- (b) waives and renounces to any right to be an Affected Creditor.

4. <u>VALUATION OF CLAIMS, CREDITORS' MEETING AND RELATED MATTERS</u>

4.1. Conversion of Affected Claims into Canadian Currency

For the purposes of determining the value of Affected Claims denominated in currencies other than Canadian dollars for voting purposes, such Affected Claims shall be converted by the Monitor to Canadian dollars at the Bank of Canada noon spot rate of exchange for exchanging currency to Canadian dollars on the Determination Date (which in the case of US Dollars was C\$1.3080 for US\$1.00).

4.2. Affected Claims

Affected Creditors shall be entitled to prove their respective Affected Claims, vote their Voting Claims in respect of the Plan, and, if their Affected Claims become Proven Claims, receive the distributions, all as set forth in the Claims Procedure Order and this Plan.

All amounts recognized as Voting Claims or Proven Claims shall be net of any amount that the Company is entitled to offset, recoup, compensate or otherwise apply in reduction of such amounts.

4.3. Creditors' Meeting

The Creditors' Meeting shall be held in accordance with this Plan, the Meeting Procedure Order, and any further Order which may be made from time to time for the purposes of considering and voting on the Resolution or other matters to be considered at the Creditors' Meeting.

4.4. Approval by Affected Creditors

The Company will seek approval of the Plan by the affirmative vote of the Required Majorities. The Resolution to be voted on at the Creditors' Meeting will be decided by the Required Majorities on a vote as set forth in the Meeting Procedure Order. The result of any vote will be binding on all Affected Creditors, whether or not any such Affected Creditor is present and voting (in person or by proxy) at the Creditors' Meeting.

4.5. Amendment of Proofs of Claim

Any Affected Creditor who filed its Proof of Claim prior to the Claims Bar Date or who has been permitted to file a late claim pursuant to an Order, shall be entitled to amend its Proof of Claim in order for its Proof of Claim to conform to the provisions of this Plan.

4.6. Late Filing of Claims for Disclaimed Leases

Notwithstanding the Claims Bar Date and its effect or anything contained in this Plan or the Claims Procedure Order, any Disclaimed Landlord in respect of a Disclaimed Lease which may be disclaimed or resiliated between the date of this Plan and the Sanction Date shall be entitled, during (but not after) a period of 30 days following the date of the disclaimer or resiliation of such Lease, to file a Disclaimed Landlord Claim with the Monitor in conformity with Section 2.3 hereof. If such Disclaimed Landlord fails to file such Disclaimed Landlord Claim within such 30 day period, such Disclaimed Landlord shall have no right to receive any distribution of the Aggregate Distribution and the Company shall be released from such Disclaimed Landlord Claim and Sections 6.2 and 6.3 of the Plan shall apply to such Disclaimed Landlord Claim.

4.7. Claims Bar Date

Subject to the provisions of **Sections 4.5** and **4.6** hereof, if an Affected Creditor holding an Affected Claim has failed to file its Proof of Claim prior to the Claims Bar Date and has not been permitted to file a late claim pursuant to an Order, that Affected Creditor shall be barred from voting at the Creditors' Meeting, that Affected Creditor shall have no right to receive any distributions of the Aggregate Distribution hereunder, the Company shall be released from the Affected Claims of such Creditor and **Sections 6.2** and **6.3** of this Plan shall apply to all such Affected Claims.

5. PROCEDURE FOR RESOLVING DISTRIBUTIONS IN RESPECT OF DISPUTED CLAIMS

5.1. No Distributions Pending Allowance

Notwithstanding any other provision of this Plan, no distributions of the Aggregate Distribution hereunder shall be made by the Monitor with respect to a Disputed Claim unless and until it has become a Proven Claim. Disputed Claims shall be dealt with in accordance with the Claims Procedure Order and this Plan.

5.2. Distribution From Reserve Once Disputed Claims Resolved

The Monitor shall make allocations from the Reserve to holders of Disputed Claims following any of the Payment Dates in accordance with this Plan. To the extent that Disputed Claims become Proven Claims after any relevant Payment Date, the Monitor

shall, from time to time at its sole discretion, distribute from the Reserve to the holders of such Proven Claims, the amount which they would have been entitled to receive in respect of such Proven Claims had such Claims been Proven Claims on any relevant Payment Date. To the extent that any Disputed Claim or a portion thereof has become a Disallowed Claim, then the Monitor shall, at any time which it deems appropriate in the circumstances, distribute, to the holders of Proven Claims, their *pro rata* share from the Reserve of such additional amount kept in the Reserve on account of any such Disallowed Claim.

6. <u>EFFECT OF THE PLAN AND RELEASES</u>

6.1. Effect of the Plan

Upon the Aggregate Distribution being fully remitted to the Monitor, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Performance"), whereupon all Affected Claims shall be deemed to be fully and finally settled, compromised, released and transacted (as envisaged by the relevant provisions of the Civil Code of Quebec), as of the date of issuance of the Certificate of Performance, subject only to an Affected Creditor's right to recover the distributions under this Plan, except for Claims as may have been fully paid prior to such date, in which case those Affected Claims will be deemed to be settled as of such payment.

6.2. Releases upon Implementation

Upon the Implementation of this Plan on the Plan Implementation Date, each of:

- (a) the Monitor and its legal counsel in the CCAA proceedings;
- (b) the Company's legal counsel, financial advisors, consultants and agents of the Company (and their respective directors, officers and employees); and
- (c) the DIP Lender and its legal counsel,

(collectively the "First Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Person may be entitled to assert whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with the business and affairs of the Company, this Plan and the CCAA Proceedings, and all

claims arising out of such actions or omissions shall be forever waived and released, all to the full extent permitted by law.

6.3. Releases after Performance

Upon the Aggregate Distribution being fully remitted to the Monitor and the Monitor's issuance and filing with the Court of the Certificate of Performance, each of:

- (a) the Company; and
- (b) the Directors and all present and future officers and employees of the Company,

(each a "Second Released Party" and collectively the "Second Released Parties") shall be released and discharged from any and all demands, claims, actions, causes of action, counterclaims, suits, debts, sums of money, accounts, covenants, damages, judgments, expenses, executions, liens and other recoveries on account of any liability, obligation, demand or cause of action of whatever nature which any Person (including any Secured Creditor in respect of each Second Released Party, except the Company and solely in respect to its Secured Claim), may be entitled to assert (including any and all Claims in respect of statutory liabilities and any Directors' Claims of all Directors and all present and former officers and employees of the Company and any alleged fiduciary or other duty), whether known or unknown, matured or unmatured, foreseen or unforeseen, existing or hereafter arising, based in whole or in part on any act or omission, transaction, duty, responsibility, indebtedness, liability, obligation, dealing or other occurrence existing or taking place on or prior to the Plan Implementation Date in any way relating to, arising out of or in connection with Claims, the business and affairs of the Company, this Plan and the CCAA Proceedings, and all claims arising out of such actions or omissions shall be forever waived and released (other than the right to enforce the Company's obligations under the Plan or any related document), all to the full extent permitted by law.

6.4. Exception to Releases

Nothing set forth in Sections 6.2 and 6.3 hereof shall:

- (a) release or discharge a Second Released Party from an Unaffected Claim;
- (b) release or discharge the Company from or in respect of its obligations under this Plan:
- (c) affect the right of any Person:
 - (i) to recover an indemnity from any insurance coverage under which that Person is an insured; or

- (ii) to obtain recovery on a claim or liability against a Second Released Party from any insurance coverage pursuant to which that Second Released Party is an insured, but, for certainty, any claim or liability to which an insurer is or would otherwise be subrogated as against the Company is released hereunder and the recovery to which such Person shall be entitled under such insurance coverage shall be limited to the proceeds of insurance actually paid by the insurer with respect to such claim or liability;
- release or discharge the Directors with respect to matters set out in Section 5.1
 (2) of the CCAA; or
- (e) release or discharge any Person (whether or not a Director, officer or employee of the Company) other than the Company from any obligations arising from any guarantee furnished by such Person to and in favour of any Affected Creditor.

6.5. Set-Off Rights

Notwithstanding the releases set forth under the Plan, in general, and **Sections 6.1**, **6.2** and **6.3** hereof, in particular, all Claims shall remain subject to any rights of set-off that otherwise would be available to the Person against whom a Claim is asserted.

6.6. <u>Injunction Related to Releases</u>

The Sanction Order will enjoin the prosecution by or on behalf of any Person, whether directly, derivatively or otherwise, of any claim, obligation, suit, judgment, damage, demand, debt, right, cause of action, liability or interest released, discharged or terminated pursuant to this Plan.

6.7. Waiver of Defaults

From and after the Plan Implementation Date, all Persons shall be deemed to have waived any and all defaults or occurrences of the Company then existing or previously committed by the Company, caused by the Company or arising, directly or indirectly, or non-compliance with any covenant, positive or negative pledge, warranty, representation, term, provision, condition or obligation, express or implied, in any contract, credit document, agreement for sale, lease or other agreement, written or oral, and any and all amendments or supplements thereto, existing between such Person and the Company arising from the Company's insolvency, the Company's filing a notice of intention to make a proposal under the BIA, the filing by the Company under the CCAA or the transactions contemplated by this Plan or otherwise, and any and all notices of default and demands for payment under any instrument, including any guarantee arising from such default, shall be deemed to have been rescinded.

6.8. Stay of Proceedings

The stay of proceedings set forth and ordered in the Initial Order (except to the extent modified by any other Order) in respect of the Company, the Directors and all other Stayed Parties shall be and remain in full force and effect up until and including the Plan Implementation Date and the Affected Creditors hereby consent to the Court's issuance of an Order to such effect.

7. PROVISIONS GOVERNING DISTRIBUTIONS

7.1. Partial Distributions for Claims Allowed

Except as otherwise provided herein or as ordered by the Court, distributions of the Aggregate Distribution shall be made by the Monitor at the time and in the manner deemed reasonable by the Monitor, and partial distributions may be made prior to receipt of the full amount of the Aggregate Distribution. In such a case, all such partial payments shall represent the *pro rata* amount of the distribution to which the holders of a Proven Claim would otherwise be entitled to receive.

7.2. Currency to be used for the Distribution

For the purposes of determining any Claims denominated in currencies other than Canadian Dollars for distribution purposes, such Claims shall be converted by the Monitor to Canadian dollars at the Bank of Canada noon spot rate of exchange for exchanging currency to Canadian dollars on the Determination Date (which in the case of US Dollars was C\$1.3080 for US\$1.00).

7.3. Assignment of Claims

For purposes of determining entitlement to receive a distribution pursuant to this Plan, the Company and the Monitor and each of their respective agents, successors and assigns shall have no obligation to recognize any transfer of Claims except as provided for under the Claims Procedure Order or the Meeting Procedure Order.

7.4. Interest on Affected Claims

Except as specifically provided in the Plan or the Sanction Order, interest shall not be treated as accruing on account of any Affected Claims for purposes of determining the allowance and distribution of such Affected Claim. To the extent that any Proven Claim to which a distribution under this Plan relates is comprised of indebtedness and accrued but unpaid interest thereon, such distribution shall, to the extent permitted by applicable Laws, be allocated for tax purposes to the principal amount of such Proven Claim (including any secured and unsecured portion(s) of the principal amount of such Proven Claim) first and then, to the extent that the consideration exceeds the principal amount of

the Proven Claim, to the portion of such Proven Claim representing accrued but unpaid interest (including interest in respect of any secured portion of such Proven Claim).

7.5. Distributions by Monitor

The Monitor shall make all distributions required under this Plan in accordance with the provisions of this Plan, in general, and the provisions of **Section 5** and **Section 7** hereof, in particular. The Monitor shall receive, without further Court approval, reasonable compensation from the Company for distribution services rendered pursuant to the Plan.

7.6. <u>Delivery of Distributions</u>

- (a) Subject to Section 7.3 hereof, distributions shall be made by the Monitor (i) at the addresses set forth on the Proof of Claim form filed by the Affected Creditors (or at the last known addresses of such Affected Creditors if no Proof of Claim form was filed or if the Company or the Monitor has been notified in writing of a change of address), (ii) at the addresses set forth in any written notice of address change delivered to the Monitor after the date of any related Proof of Claim, or (iii) in a registered retirement savings plan account designated by any employee of the Company;
- (b) If any distribution to a Creditor is returned as undeliverable, no further distributions to such Creditor shall be made unless and until the Monitor is notified of the then-current address of such Creditor, at which time all missed distributions shall be made to such Creditor without interest. The Company shall make reasonable efforts to locate Affected Creditors for which distributions were undeliverable. Any claim for undeliverable distributions must be made on or before the later to occur of (i) 3 months after the Payment Date, or (ii) 3 months after such Creditor's Claim becomes a Proven Claim, after which date all unclaimed property shall revert to the Company free of any restrictions or claims thereon and the claim of any Creditor with respect to such property shall be discharged and forever barred, notwithstanding any applicable Laws to the contrary.

7.7. No Double Recovery

The aggregate recovery on account of any Proven Claim from all sources, regardless of whether on account of a theory of primary or secondary liability, by reason of guarantee, surety, indemnity, joint and several obligation or otherwise, shall not exceed (i) 100% of the underlying indebtedness, liability or obligation giving rise to such Claim or, (ii) where the underlying indebtedness, liability or obligation giving rise to such Claim is denominated in a currency (the "Original Currency") other than Canadian dollars, 100% of such underlying indebtedness, liability or obligation after conversion of the value of the distributions received in Canadian dollars back to the Original Currency at the Bank of

Canada noon spot rate of exchange for exchanging Canadian dollars to the Original Currency on the Determination Date.

7.8. Withholding Requirements

In connection with the Plan, any distribution made hereunder by the Monitor shall be made net of all applicable Taxes. Notwithstanding any other provision of the Plan, each Creditor that is to receive a distribution pursuant to the Plan shall have sole and exclusive responsibility for the satisfaction and payment of any Taxes imposed by any Governmental Authority (including income, withholding and other Taxes on account of such distribution). The Monitor, as necessary, shall be authorized to take any and all actions as may be necessary or appropriate to comply with such withholding and reporting requirements. All amounts withheld on account of Taxes shall be treated for all purposes as having been paid to the Creditor in respect of which such withholding was made, provided such withheld amounts are remitted to the appropriate Governmental Authority.

8. IMPLEMENTATION OF THE PLAN

8.1. Conditions Precedent to Implementation of Plan

The implementation of this Plan by the Company is subject to the occurrence and/or fulfillment of each of the following conditions precedent (the "Plan Implementation Conditions") on or prior to December 18, 2015 or such later date as may be ordered by the Court (the "Plan Implementation Deadline"), namely:

- (a) this Plan shall have been approved by Resolution enacted by the Required Majorities;
- (b) the Sanction Order sanctioning this Plan shall have been issued, and shall either (A) order its provisional execution notwithstanding appeal, or (B) not then be subject to any appeal therefrom, and the operation and effect of the Sanction Order shall not have been stayed, reversed or amended, and shall, among other things, declare and order that:
 - (i) this Plan, including the compromises and arrangements set out herein, is sanctioned and approved pursuant to Section 6 of the CCAA and, as at the Plan Implementation Date, will be effective and will enure to the benefit of and be binding upon the Company, the Affected Creditors and all other Persons referred to in this Plan or in the Sanction Order;
 - (ii) (A) this Plan has been approved by the Required Majorities of Affected Creditors of the Company in conformity with the CCAA; (B) the Company has complied with the provisions of the CCAA and the Orders of the Court

made in the CCAA Proceedings in all respects; (C) the Court is satisfied that the Company has neither done nor purported to do anything that is not authorized by the CCAA; and (D) this Plan and the transactions contemplated thereby are fair and reasonable;

- (iii) the full and final release and discharge of the Claims will be effective and will enure to the benefit of and be binding upon the Company, the Affected Creditors and all other Persons referred to in this Plan upon the issuance of the Certificate of Performance;
- (iv) the Company and the Monitor are authorized to take all steps and actions necessary to implement this Plan;
- (v) all Proven Claims determined in accordance with the Claims Procedure Order and this Plan are final and binding on the Company and all Affected Creditors:
- (vi) subject to Sections 4.5 and 4.6 hereof, any Claims for which a Proof of Claim has not been filed by the Claims Bar Date shall be forever barred and extinguished;
- (vii) all distributions and payments by or at the direction of the Monitor, in each case on behalf of the Company, under the Plan are for the account of the Company and the fulfillment of its obligations under the Plan;
- (viii) the Company and the Monitor may apply to the Court for advice and direction in respect of any matters arising from or under the Plan;
- subject to the performance by the Company of its obligations under this Plan, all contracts, leases, agreements and arrangements to which the Company are a party and that have not been disclaimed, terminated or repudiated pursuant to the Initial Order and/or the CCAA will be and remain in full force and effect, unamended, as at the Plan Implementation Date, and no Person who is a party to any such contract, lease, agreement or other arrangement may accelerate, terminate, rescind, refuse to perform or otherwise repudiate its obligations thereunder, or enforce or exercise any right (including any right of dilution or other remedy) or make any demand under or in respect of any such contract, lease, agreement or other arrangement and no automatic termination will have any validity or effect, by reason of:
 - (A) any event that occurred on or prior to the Plan Implementation Date and is not continuing that would have entitled such Person to enforce those rights or remedies (including defaults, events of

default, or termination events arising as a result of the insolvency of the Company);

- (B) the insolvency of the Company, the fact that the Company filed a notice of intention to make a proposal under the *BIA* or the fact that the Company sought or obtained relief under the *CCAA*; or
- (C) any compromises or arrangements effected pursuant to this Plan or any action taken or transaction effected pursuant to this Plan;
- the releases set forth in **Section 6.2** hereof and the injunction set forth in **Section 6.6** hereof are implemented and in effect;
- (xi) the commencement or prosecution, whether directly, derivatively or otherwise, or any demands, claims, actions, causes of action, counterclaims, suits or any indebtedness, liability, obligation or cause of action released and discharged pursuant to this Plan are enjoined; and
- (xii) the stay of proceedings as set forth in **Section 6.8** hereof is and remains in full force and effect up until and including the Plan Implementation Date; and
- (c) the Replacement Financing shall have occurred.

8.2. Waiver of Conditions

Except for the condition set forth at Section 8.1(c) hereof, each of the Plan Implementation Conditions enure solely to the Company's benefit and, except for each of the conditions set forth in Sections 8.1(a) and 8.1(b)(i) hereof, may be waived in whole or in part either (i) solely by the Company (and no other Person and without the consent of any other Person), without any other notice to parties in interest or the Court and without a hearing, or (ii) by the Court. The failure to satisfy or waive any of the Plan Implementation Conditions prior to the Plan Implementation Date may be asserted by the Company regardless of the circumstances giving rise to the failure of such Plan Implementation Conditions to be satisfied (including any action or inaction by the Company). The failure of the Company to exercise any of the foregoing rights shall not be deemed a waiver of any other rights, and each such right shall be deemed an ongoing right that may be asserted at any time. Notwithstanding the foregoing, the condition set forth at Section 8.1(c) hereof may only be waived in writing by the Company and the DIP Lender.

8.3. <u>Certificate of Implementation</u>

Upon the occurrence and/or fulfillment of all of the Plan Implementation Conditions (or waiver thereof, as the case may be) on or prior to the Plan Implementation Deadline, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Implementation").

8.4. Effect of Plan

Upon the issuance of the Certificate of Performance, the settlement of the Affected Claims in accordance with the Plan shall become final and binding on the Company, and all the Affected Creditors and their respective successors and assigns, and this Plan shall, upon the Aggregate Distribution being fully remitted to the Monitor and the Monitor's issuance and filing with this Court of the Certificate of Performance, result in the full and final settlement of all Affected Claims, including for greater certainty, any Restructuring Claim and any Claims resulting, directly or indirectly, from the consequences and effects relating to the acceptance of the Plan by the Affected Creditors, its sanction by the Court, or its implementation, and any indebtedness. obligations or undertakings which the Company may subsequently become subject to, directly or indirectly, as a result of an obligation, transaction or an event that occurred before the Determination Date, as well as any indebtedness, obligations or any undertakings that the Company may be subject to on any date whatsoever in connection with the Plan, the approval thereof by the Court or the implementation thereof. For greater certainty, this Plan shall not affect or impair any rights, remedies and recourses which the Company had, has or may have after the issuance of the Certificate of Performance in connection with transactions, facts or obligations existing prior to the Determination Date.

8.5. Nullity of Plan

In the event that all of the Plan Implementation Conditions have not occurred and/or been fulfilled (or waived, as the case may be) on or prior to the Plan Implementation Deadline, the Monitor shall issue and file with the Court a certificate to such effect (the "Certificate of Non-Implementation"). Upon the issuance of such Certificate of Non-Implementation, any settlement of the Affected Claims shall automatically become null, void and of no effect whatsoever and shall remain owing by the Company and neither the Company, the Affected Creditors, the Monitor nor any other Person affected by this Plan shall be bound, obliged or affected by any of the provisions of this Plan.

9. MISCELLANEOUS

9.1. Confirmation of Plan

Upon occurrence and/or fulfillment of all of the Plan Implementation Conditions (or waiver thereof, as the case may be) on or prior to the Plan Implementation Deadline, this Plan shall be binding upon the Company, the Affected Creditors and any and all other Persons affected by the provisions of this Plan as well as each of their respective successors and assigns.

9.2. Paramountcy

From and after the Plan Implementation Date, any conflict between this Plan and/or the covenants, warranties, representations, terms, conditions, provisions or obligations, express or implied, of any contract, mortgage, security agreement, and/or indenture, trust indenture, loan agreement, commitment letter, agreement for sale, the by-laws of the Company, lease or other agreement, undertaking or any other source of obligations, written or oral and any and all amendments or supplements thereto existing between one or more of the Affected Creditors and the Company as at the Plan Implementation Date will be deemed to be governed by the terms, conditions and provisions of this Plan and the Sanction Order, which shall take precedence and priority. For greater certainty, all Affected Creditors shall be deemed irrevocably for all purposes to consent to all transactions contemplated in and by this Plan.

9.3. Modification of Plan

The Company:

in consultation with the Monitor and the DIP Lender (if any DIP Claim remains outstanding), reserves the right to file any modification of, or amendment or supplement to, this Plan by way of supplementary plan or plans of reorganization, compromise or arrangement (or any one or more thereof) at or before the Creditors' Meeting, in which case any such supplementary plan or plans of reorganization, compromise or arrangement (or an one or more thereof), shall, for all purposes, be and be deemed to form part of and be incorporated into this Plan. The Company shall file any supplementary plans with the Court as soon as practicable. The Company shall give notice to Affected Creditors of the details of any such modification, amendment or supplement at the Creditors' Meeting prior to the vote being taken to approve this Plan. The Company may give notice of a proposed modification, amendment or supplement to this Plan at or before the Creditors' Meeting by notice which shall be sufficient if given to those Affected Creditors present at such meeting in person or by proxy; and/or

(b) after the Creditors' Meeting (and both prior to and subsequent to the obtaining of the Sanction Order), the Company, in consultation with the Monitor and, if prior to the Plan Implementation Date, the DIP Lender (if any DIP Claim then remains), at any time and from time to time vary, amend, modify or supplement this Plan, (except to reduce the amount of the Aggregate Distribution or extend any Payment Dates other than as expressly provided herein), without the need for obtaining an Order of the Court or providing notice to the Affected Creditors if the Monitor determines that such variation, amendment, modification or supplement would not be materially prejudicial to the interests of the Affected Creditors under this Plan or the Sanction Order and is necessary in order to give effect to the substance of this Plan or the Sanction Order. All of the foregoing shall not require any further vote by or approval by the Affected Creditors or any approval by the Court.

9.4. <u>Deeming Provisions</u>

In this Plan, any deeming provisions are not rebuttable and are conclusive and irrevocable.

9.5. Sections 38 and 95 to 101 BIA

Notwithstanding Section 36.1 of the *CCAA*, Section 38 and Sections 95 through 101 of the *BIA* shall not apply to this Plan. Accordingly, neither the Monitor, any Creditor nor any other Person shall be entitled to exercise any right, remedy or recourse, or to commence any action, motion, application or any other proceeding against any member of the Fisher Group, any Creditor or any other Person in relation to the Company, based on Section 38 and Sections 95 through 101 of the *BIA*.

9.6. Responsibilities of the Monitor

The Monitor is acting in its capacity as Monitor in the CCAA Proceedings with respect to the Company and not in its personal or corporate capacity and will not be responsible or liable for any responsibilities or obligations of the Company under this Plan or otherwise, including with respect to the making of distributions or the receipt of any distribution by any Creditor or any other Person pursuant to the Plan. The Monitor will have the powers and protections granted to it by this Plan, the CCAA, the Initial Order, the Claims Procedure Order, the Meeting Procedure Order, the Sanction Order and any other Orders.

9.7. Liability Limitations

The Monitor, the Company and their respective legal counsel and other professional advisors, shall not be liable to any Person for any act or omission in connection with, or arising out of, the CCAA Proceedings, the Plan, the pursuit of sanctioning of the Plan.

the consummation and implementation of the Plan or the administration of the Plan or the funds to be distributed under the Plan, except for their own wilful misconduct or gross negligence.

9.8. Notices

- (a) Any notices or communication to be made or given hereunder to the Company or the Monitor shall be in writing and shall refer to this Plan and may, subject as hereinafter provided, be made or given by personal delivery, by courier, by prepaid mail, by fax or by e-mail, addressed to the respective parties as follows:
 - (i) if to the Company:

LAURA'S SHOPPE (P.V.) INC.

2955 Jules-Brillant Laval, Quebec H7P 6B2

Attention:

Kalman Fisher and

Josh Fisher

E-Mails:

kfisher@laura.ca

josh.fisher@laura.ca

Fax:

(450) 973-6099

with copy to:

KUGLER KANDESTIN LLP

1 Place Ville Marie Suite 2101 Montreal, Quebec H3B 2C6

Attention:

Gerald F. Kandestin,

David Stolow and

Jeremy Cuttler

E-Mail:

gkandestin@kklex.com

dstolow@kklex.com

icuttler@kklex.com

Fax:

(514) 875-8424

(ii) if to the Monitor:

KPMG LLP

KPMG Tower 600 De Maisonneuve Blvd., West Suite 1500 Montreal, Quebec H3A 0A3

Attention:

Dev Coossa

E-Mail:

dcoossa@kpmg.ca

Fax:

(514) 840-2121

with copy to:

FISHMAN FLANZ MELAND PAQUIN, L.L.P.

1250 Rene-Levesque Blvd., West Suite 4100 Montreal, Quebec H3B 4W8

Attention:

Avram Fishman and

Mark Meland

E-Mails:

afishman@ffmp.ca

mmeland@ffmp.ca

Fax:

(514) 932-4170

or to such other address as any party may from time to time notify the others in accordance with this Section 9.8. In the event of any strike, lock-out or other event which interrupts postal service in any part of Canada, all notices and communications during such interruption may only be given or made by personal delivery, fax or e-mail and any notice or other communication given or made by prepaid mail within the 5 Business Day period immediately preceding the commencement of such interruption, unless actually received, shall be deemed not to have been given or made. All such notices and communications shall be deemed to have been received, in the case of notice by delivery, fax or e-mail prior to 5:00 p.m. (local time) on a Business Day, when received or if received after 5:00 p.m. (local time) on a Business Day or at any time on a non-Business Day, on the next following Business Day and, in the case of notice mailed as aforesaid, on the fourth Business Day following the date on which such notice or other communication is mailed. The unintentional failure by the Company or the Monitor to give any notice contemplated hereunder to any particular Creditor shall not invalidate this Plan or any action taken by any Person pursuant to this Plan.

(b) Any notices or communication to be made or given hereunder by the Monitor or the Company to a Creditor may be sent by fax, e-mail, ordinary mail, registered mail, courier or facsimile transmission to the e-mail address, address or fax number specified by such Creditor in its Proof of Claim, or in any subsequent written notice of change of address given to the Monitor. A Creditor shall be deemed to have received any document sent pursuant to this Plan 4 Business Days after the document is sent by ordinary or registered mail and on the Business Day immediately following the day on which the document is sent by courier, fax or e-mail.

9.9. Severability of Plan Provisions

If, prior to the Sanction Date, any term or provision of this Plan is held by the Court to be invalid, void or unenforceable, the Court, at the request of the Company which request shall be made in consultation with the Monitor and the DIP Lender (if any DIP Claim then remains), shall have the power to either:

- (a) sever such term or provision from the balance of this Plan and provide the Company with the option to proceed with the implementation of the balance of this Plan as of and with effect from the Plan Implementation Date; or
- (b) alter and interpret such term or provision to make it valid and enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpreted.

Notwithstanding any such severing, holding, alteration or interpretation, and provided the Company proceeds with the implementation of this Plan, the remainder of the terms and provisions of this Plan shall remain in full force and effect and shall in no way be affected, impaired or invalidated by such severing, holding, alteration or interpretation.

9.10. Revocation, Withdrawal or Non-Consummation

The Company, upon consultation with the Monitor and the DIP Lender, reserves the right to revoke or withdraw this Plan at any time prior to the Sanction Date and to file subsequent plans of arrangement and compromise. If the Company revokes or withdraws this Plan, or if the Sanction Order is not issued:

- (a) this Plan shall be null, void and inoperative in all respects;
- (b) any Claim, any settlement or compromise embodied in this Plan (including the fixing or limiting of any Claim to an amount certain), assumption or termination by this Plan, and any document or agreement executed pursuant to this Plan shall be deemed null, void and inoperative; and

- (c) nothing contained in this Plan, and no act taken in preparation for consummation of this Plan, shall:
 - constitute or be deemed to constitute a waiver or release of any Claims by or against the Company or any other Person;
 - (ii) prejudice in any manner the rights of the Company or any Person in any further proceedings involving the Company; or
 - (iii) constitute an admission of any sort by the Company or any other Person.

9.11. Further Assurance

Notwithstanding that the transactions and events set out in this Plan shall occur and be deemed to occur in the order set out herein without any additional act or formality, each of the Persons affected hereby shall make, do and execute, or cause to be made, done and executed at the cost of the requesting party, all such further acts, deeds, agreements, transfers, assurances, instruments or documents as may reasonably be required by the Company in order to better implement this Plan.

9.12. Governing Law

This Plan shall be governed by and construed in accordance with the law of the Province of Quebec and the federal laws of Canada applicable therein. Any questions as to the interpretation or application of this Plan and all proceedings taken in connection with this Plan and its provisions shall be subject to the exclusive jurisdiction of the Court.

9.13. Successors and Assigns

This Plan shall be binding upon and shall enure to the benefit of the heirs, administrators, executors, legal personal representatives, successors and permitted assigns of the Company, the Directors, the Affected Creditors or any other Persons affected by or benefiting from the provisions of this Plan.

9.14. French Language Version of Plan

In the event of any discrepancy between any of the provisions of the English language version of this Plan and any French language version thereof, the provisions of the English version of this Plan shall, under all circumstances, prevail and govern.

9.15. Choice of Language

The Company and each of the hereafter enumerated members of the Fisher Group acknowledges that it (or he) has required that this Plan and all related documents be

prepared, in English. La Compagnie et chacun des membres du "Fisher Group" ci-après énumerés reconnaît avoir exigé que ce Plan et tous les documents connexes soient rédigés en anglais.

(signature page to follow)

MONTREAL, Province of Quebec, this $\frac{18^{t}}{1000}$	day of November, 2015.		
	LAURA'S SHOPPE (P.V.) INC. Per:		
	Kalman Fisher, President		
EACH of the following members of the Fisher Group acknowledges haven taken cognizance of the contents of this Plan and consents to the provisions of Section 3.11 hereof.			
	KALMAN FISHER		
	3482731 CANADA INC. Per:		
	Kalman Fisher		
	9318-5494 QUEBEC INC. Per:		
	Kalman Fisher		

COMPANIES' CREDITORS ARRANGEMENT ACT, R. S.C. C. C-36 (SITTING AS A COURT DESIGNATED PURSUANT TO THE SUPERIOR COURT (COMMERCIAL DIVISION) PROVINCE OF QUÉBEC AS AMENDED)

IN THE MATTER OF THE PLAN OF COMPROMISE OR ARRANGEMENT OF:
MAGASIN LAURA (P.V.) INC. / LAURA'S SHOPPE (P.V.)

Debtor/Petitioner

-and-

CANADA LTD. **BOUTIQUE LAURA CANADA LTÉE/LAURA'S SHOPPE**

3482731 CANADA INC.

9318-5494 QUEBEC INC.

KALMAN FISHER

KPMG INC.

Stayed Parties

Monitor

MOTION FOR SANCTION OF THE PETITIONER'S RE-AMENDED PLAN OF COMPROMISE AND ARRANGEMENT AND FOR OTHER RELIEF (Sections 6, 9, 11 & 11.02 of the CCAA, R.S.C. 1985, C. C-36)

COPY

Me Gerald F. Kandestin

KuglerKandestin

1, Place Ville Marie, Suite 2101 Montréal (Québec) Canada H3B 2C6 T: 514 878-2861 gkandestin@kklex.com F: 514 875-8424

BG 0132

○ 5931-01/GFKUC